DIGITAL AVERAGE-CURRENT CONTROL FOR THE DUAL INTERLEAVED BOOST CONVERTER

A thesis submitted to The University of Manchester for the degree of

Doctor of Philosophy

in the Faculty of Engineering and Physical Sciences

2015

Ву

Alejandro Villarruel-Parra

School of Electrical and Electronic Engineering

Table of contents

TABLE OF CONTENTS	2
LIST OF FIGURES	8
LIST OF TABLES	19
LIST OF SYMBOLS	21
	20
	, Z <i>1</i>
DECLARATION	28
COPYRIGHT STATEMENT	29
ACKNOWLEDGEMENTS	31
CHAPTER 1 INTRODUCTION AND LITERATURE REVIEW	32
1.1. INTRODUCTION	32
1.1.1. Electric vehicles	32
1.1.2. Objectives	35
1.1.3. Structure of the literature review	35
1.2. Control for switched DC-DC converters	36
1.2.1. Analog control techniques	36
1.2.1.1. Voltage-mode control	36
1.2.1.2. Current-mode control	37
1.2.2. Digital control techniques	39
1.2.2.1. Architecture and elements of a digital controlled dc-dc converter	40
1.2.2.2. Synchronization of the ADC and the DPWM for current-control	43
1.2.3. Review of digital control techniques for dc-dc converters	43
1.2.3.1. Linear compensation or conventional compensation methods	43
1.2.3.2. Predictive or estimative methods	44
1.2.3.3. Multi-sampling methodologies	45
1.2.3.4. Mixed-signal methodologies	46
1.2.3.5. Non-linear methodologies	46
1.3. CONTROL TECHNIQUES FOR MULTI-PHASE DC-DC CONVERTERS	47
1.3.1. Control techniques for the dual interleaved boost converter	49

1.4.	SMALL-SIGNAL MODELLING OF DIGITALLY CONTROLLED DC-DC CONVERTERS AND COMPENSATOR	
DESIGN	۷	50
1.4	.1. Modelling of digital average-current-controlled dc-dc converters	52
1	1.4.1.1. DPWM Model	52
1	1.4.1.2. Quantization and limit cycle oscillations	54
1.5.	COMPENSATOR DESIGN FOR DIGITAL CONTROLLED DC-DC CONVERTERS	54
1.5	.1. Digital redesign	54
1.5	.2. Direct digital design	56
1.5	.3. Other design methodologies	57
1.6.	REVIEW OF MODELLING OF MULTI-PHASE DC-DC CONVERTERS	57
1.7.	SUMMARY OF LITERATURE REVIEW	60
1.8.	OUTLINE OF THIS THESIS	62
1.9.	References	64
СНАРТЕ	ER 2 CONVERTER OPERATION, MODELLING AND DIGITAL CONTROL	71
2.1.	INTRODUCTION	71
2.2.	OPERATION OF THE DUAL INTERLEAVED BOOST CONVERTER WITH INTER-PHASE TRANSFORMER	71
2.3.	SMALL-SIGNAL AC MODELLING OF THE CONVERTER	77
2.3	.1. Averaged switch modelling of the boost converter	77
2.3	.2. Averaged small-signal modelling of the dual interleaved boost converter	79
2.4.	SIMULATION MODELS OF THE CONVERTER	86
2.4	.1. MATLAB/Simulink averaged small-signal ac model	86
2.4	.2. SABER switched model	86
2.4	.3. SABER averaged model	88
2.5.	SIMULATION RESULTS	89
2.5	.1. Verification of the SABER switched and SABER averaged simulation models	90
2.5	.2. Validation of the small-signal ac model	95
2	2.5.2.1. Dynamic response	95
2	2.5.2.2. Response to small-sinusoidal perturbations	98
2.6.	DIGITAL CURRENT CONTROL WITH UNIFORMLY-SAMPLED PWM FOR THE DUAL INTERLEAVED BOO	DST
CONVE	ERTER WITH IPT 1	00
2.6	.1. Analysis of the leading-edge, trailing-edge and triangular carrier modulatio	ns
for	the DIBC with IPT 1	04
2.7.	SUMMARY 1	10
2.8.	References 1	12
СНАРТЕ	ER 3 DIGITAL CONTROL PLATFORM AND ALGORITHM IMPLEMENTATION 1	13
3.1.	INTRODUCTION	13
3.2.	Prototype of the dual-interleaved boost converter with inter-phase transformer 1	14
3.3.	DIGITAL CONTROL PLATFORM PROTOTYPE	16

Table of contents

J.J.T.	Digital Signal Controller (DSC)	118
3.3.2.	Current sensors and conditioning for the measured signals	119
3.3.3.	Signal conditioning for transistor driving signals and gate drivers	120
3.3.3.	Construction Details	121
3.4. Iм	PLEMENTATION OF THE DIGITAL CURRENT-CONTROL ALGORITHMS ON THE DSC	123
3.4.1.	Configuration and synchronization of the DSC modules	123
3.4.2.	ADC interrupt service routine, ADC_ISR	126
3.4.3.	Detailed timing of the control algorithms	128
3.4.4.	Average-current control with immediate DPWM update	131
3.5. SA	BER SIMULATION MODELS	133
3.6. Res	SULTS OF THE DIGITAL VALLEY-CURRENT AND PEAK-CURRENT CONTROL	133
3.6.1.	Steady-state operation of valley-current control strategy	134
3.6.2.	Steady-state operation of the peak-current control strategy	137
3.6.3.	Timing of the control algorithms	139
3.6.4.	Range of operation	142
3.6.5.	Dynamic performance	144
3.7. Re:	SULTS OF THE DIGITAL AVERAGE-CURRENT CONTROL STRATEGIES WITH NORMAL AND IN	MEDIATE
UPDATE		150
3.7.1.	Steady-state operation	151
3.7.2.	Timing of the control algorithms	154
272		
5.7.5.	Dynamic performance	157
3.8. S∪	Dynamic performance	157 162
3.8. S∪ 3.9. RE	Dynamic performance MMARY FERENCES	157 162 163
3.8. SU 3.9. RE	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT	157 162 163 ERS . 164
3.8. SU 3.9. RE CHAPTER 4 4.1. INT	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION	157 162 163 ERS . 164 164
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM	157 162 163 ERS . 164 164 164
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1.	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM Small-signal block diagram modelling	157 162 163 ERS . 164 164 164 165
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1. 4.3. Mo	Dynamic performance. MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT PRODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM Small-signal block diagram modelling DDELLING OF THE INTERACTION OF THE DIGITAL CURRENT CONTROL LOOPS	157 162 163 ERS . 164 164 164 165 168
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1. 4.3. Mo 4.3.1.	Dynamic performance. MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM Small-signal block diagram modelling DDELLING OF THE INTERACTION OF THE DIGITAL CURRENT CONTROL LOOPS The sampler-decomposition method	157 162 163 ERS . 164 164 165 168 168
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1. 4.3. Mo 4.3.1. 4.3.2.	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT FRODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM PRESENTATION OF THE INTERACTION OF THE DIGITAL CURRENT CONTROL LOOPS THE SAMPLER - DECOMPOSITION METHOD IMPLEMENTATION OF THE SAMPLER DECOMPOSITION METHOD IN THE DIGITAL CURRENT CONTROL LOOPS PRESENTATION OF THE SAMPLER DECOMPOSITION METHOD IN THE DIGITAL CURRENT CONTROL LOOPS PRESENTATION OF THE SAMPLER DECOMPOSITION METHOD	157 162 163 ERS . 164 164 164 165 168 168
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1. 4.3. Mo 4.3.1. 4.3.2. average	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM Small-signal block diagram modelling Small-signal block diagram modelling DDELLING OF THE INTERACTION OF THE DIGITAL CURRENT CONTROL LOOPS The sampler-decomposition method Implementation of the sampler decomposition method in the digital e-current control loop	157 162 163 ERS . 164 164 165 168 168 169
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1. 4.3. Mo 4.3.1. 4.3.2. average 4.3.3.	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM Small-signal block diagram modelling Small-signal block diagram modelling DELLING OF THE INTERACTION OF THE DIGITAL CURRENT CONTROL LOOPS The sampler-decomposition method Implementation of the sampler decomposition method in the digital e-current control loop Derivation of the closed-loop transfer functions <i>G</i> _{iairef} (z) and <i>G</i> _{ibiref} (z)	157 162 163 ERS . 164 164 164 165 168 168 169 170
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1. 4.3. Mo 4.3.1. 4.3.2. average 4.3.3. 4.3.4.	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM Small-signal block diagram modelling DELLING OF THE INTERACTION OF THE DIGITAL CURRENT CONTROL LOOPS The sampler-decomposition method Implementation of the sampler decomposition method in the digital e-current control loop Derivation of the closed-loop transfer functions <i>G</i> _{iairef} (<i>z</i>) and <i>G</i> _{ibiref} (<i>z</i>) Implementation of the models in MATLAB	157 162 163 ERS . 164 164 164 168 168 168 169 170 175
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1. 4.3. Mo 4.3.1. 4.3.2. average 4.3.3. 4.3.4. 4.4. VA	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM Small-signal block diagram modelling DELLING OF THE INTERACTION OF THE DIGITAL CURRENT CONTROL LOOPS The sampler-decomposition method Implementation of the sampler decomposition method in the digital e-current control loop Derivation of the closed-loop transfer functions <i>G</i> _{iairef} (<i>z</i>) and <i>G</i> _{ibiref} (<i>z</i>) Implementation of the models in MATLAB LIDATION OF THE CLOSED-LOOP MODELS	157 162 163 ERS . 164 164 165 168 168 168 169 170 175 175
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1. 4.3. Mo 4.3.1. 4.3.2. average 4.3.3. 4.3.4. 4.4. VA 4.4.1.	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM Small-signal block diagram modelling DELLING OF THE INTERACTION OF THE DIGITAL CURRENT CONTROL LOOPS The sampler-decomposition method Implementation of the sampler decomposition method in the digital e-current control loop Derivation of the closed-loop transfer functions <i>G</i> _{iairef} (<i>z</i>) and <i>G</i> _{ibiref} (<i>z</i>) Implementation of the models in MATLAB LIDATION OF THE CLOSED-LOOP MODELS Transient response.	157 162 163 ERS . 164 164 164 165 168 168 168 169 170 175 175 176
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1. 4.3. Mo 4.3.1. 4.3.2. average 4.3.3. 4.3.4. 4.4. VA 4.4.1. 4.4.2.	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM Small-signal block diagram modelling Small-signal block diagram modelling DELLING OF THE INTERACTION OF THE DIGITAL CURRENT CONTROL LOOPS The sampler-decomposition method Implementation of the sampler decomposition method in the digital e-current control loop Derivation of the closed-loop transfer functions <i>G</i> _{iairef} (<i>z</i>) and <i>G</i> _{ibiref} (<i>z</i>) Implementation of the models in MATLAB LIDATION OF THE CLOSED-LOOP MODELS Transient response to small sinusoidal perturbations	157 162 163 ERS . 164 164 165 168 168 168 169 170 175 175 176 179
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1. 4.3. Mo 4.3.1. 4.3.2. average 4.3.3. 4.3.4. 4.4. VA 4.4.1. 4.4.2. 4.4.3.	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM Small-signal block diagram modelling DDELLING OF THE INTERACTION OF THE DIGITAL CURRENT CONTROL LOOPS The sampler-decomposition method Implementation of the sampler decomposition method in the digital e-current control loop Derivation of the closed-loop transfer functions <i>G</i> _{iairef} (<i>z</i>) and <i>G</i> _{ibiref} (<i>z</i>) Implementation of the models in MATLAB LIDATION OF THE CLOSED-LOOP MODELS Frequency response to small sinusoidal perturbations Stability range with a digital PI compensator	157 162 163 ERS . 164 164 164 165 168 168 168 169 170 175 175 175 176 179 183
3.8. SU 3.9. REI CHAPTER 4 4.1. INT 4.2. REI 4.2.1. 4.3. Mo 4.3.1. 4.3.2. average 4.3.3. 4.3.4. 4.4. VA 4.4.1. 4.4.2. 4.4.3. 4.4.4.	Dynamic performance MMARY FERENCES MODELLING PHASE INTERACTIONS IN DUAL INTERLEAVED CONVERT RODUCTION PRESENTATION OF THE CONVERTER IN THE CONTROL DIAGRAM FORM Small-signal block diagram modelling DELLING OF THE INTERACTION OF THE DIGITAL CURRENT CONTROL LOOPS The sampler-decomposition method Implementation of the sampler decomposition method in the digital e-current control loop Derivation of the closed-loop transfer functions G _{iairef} (z) and G _{ibiref} (z) Implementation of the models in MATLAB LIDATION OF THE CLOSED-LOOP MODELS Transient response to small sinusoidal perturbations Stability range with a digital PI compensator Experimental validation of the stability range	157 162 163 ERS . 164 164 164 165 168 168 168 169 170 175 175 175 176 179 183 187

Table o	f contents
---------	------------

4.6.	References	192
СНАРТЕ	R 5 DIGITAL CONTROL LOOP: DYNAMICS AND DESIGN	193
5.1.	INTRODUCTION	193
5.2.	Sensitivity analysis via Root-locus	193
5.2.	1. Varying load resistance with a proportional compensator	193
5.2.	2. Varying proportional gain of the proportional compensator	195
5.2.	3. Varying proportional gain and integral gain of the PI compensator	196
5.2.	4. Varying load resistance with a PI compensator	199
5.2.	5. Varying PWM update delay with a PI compensator	199
5.2.	6. Varying input voltage with a PI compensator	200
5.3.	DESIGN OF THE PI COMPENSATOR FOR DIGITAL AVERAGE-CURRENT CONTROL WITH IMMEDI	ATE
UPDATE	E MODE	202
5.4.	PERFORMANCE EVALUATION OF THE DESIGNED PI COMPENSATOR	209
5.4.	1. Experimental evaluation to large step-increments	209
5.4.	2. Evaluation to small step-increments	212
5.5.	VOLTAGE CONTROL LOOP IMPLEMENTATION AND RESULTS	215
5.6.	Summary	219
СНАРТЕ	R 6 CONCLUSIONS AND FURTHER RESEARCH WORK	221
6.1.	INTRODUCTION	221
6.2.	SUMMARY OF THE THESIS	221
6.2.	1. Small-signal averaged modelling of the DIBC with IPT	221
6.2.	2. Analysis of the UPWM control techniques	222
6.2.	3. Implementation of the control strategies and experimental validation.	222
6.2.	4. Modelling of phase interactions in dual-interleaved converters	223
6.2.	5. Stability analysis of the DIBC with IPT for digital average-current control	ol via
roo	t-locus	224
6.2.	6. Design of a PI compensator for the current control loops	225
6.3.	CONTRIBUTIONS OF THIS RESEARCH WORK	225
6.3.	1. Small-signal averaged model of the dual interleaved boost converter v	vith
inte	er-phase transformer	225
6.3.	2. Comprehensive study of the uniformly sampled, valley-current, peak-c	current
and	average-current control for the DIBC with IPT	225
6.3.	3. Immediately updated DPWM operation for average-current control	226
6.3.	4. Modelling phase interactions in dual interleaved converters	226
6.3.	5. Stability study of the converter	226
6.4.	FURTHER RESEARCH OPPORTUNITIES	227
6.4.	1. Simplification of the small-signal averaged state-space model of the	
con	verter	227

	6.4.	2. Verte	Generalization of the sampler decomposition technique for <i>n</i> -interleaved	227
	6.4.	3. eak-	Application of the sampler decomposition technique to model the dynam	, nics 227
	64	2 4	Development of non-linear control methodologies	228
	6.4	т. 5	Digital implementation of hidirectional control for the dual interleaved	220
	con	J. Verte	ar	228
	64	6	Modelling and control of the dual interleaved converter for hidirectional	220
	0.4. 0ne	o. ratio	n	228
	6 4 [·]	7	Symmetrical (double) sampling on immediate undate control algorithm	220
	6.4.	7. 8.	Implementation of digital control for converter with wide band gap devic	es
				229
	6.4.9	9.	Implementation of other current-regulation methodologies	229
	6.4.	10.	Implementation of alternative methods for balancing the IPT current	229
6.	5.	Con	CLUSION	229
6.	6.	Refe	RENCES	231
APP	END	IX A	DERIVATION OF MATHEMATICAL EXPRESSIONS	232
A.	1.	Deri	VATION OF EQUATIONS FOR STEADY-STATE RIPPLES	232
	A.1.	1.	Input inductor current ripple for D < 0.5	232
	A.1.	2.	Input inductor current ripple for D < 0.5	232
	A.1.	3.	Differential current ripple for D < 0.5	233
	A.1.	4.	Differential current ripple for D > 0.5	233
	A.1.	5.	Large and small phase-current ripples	234
	A.1.	6.	Large and small phase-current ripple for D < 0.5	234
	A.1.	7.	Large and small phase-current ripple for D > 0.5	234
A.	2.	Deri	VATION OF THE CLOSED-LOOP TRANSFER FUNCTIONS $G_{IAIREF}(Z)$ and $G_{IBIREF}(Z)$ with NO	
IN	TERLE	AVIN	G	236
APP	END	IX B	MATLAB SCRIPTS	239
В.	1.	CALC	CULATION OF THE CLOSED-LOOP TRANSFER FUNCTIONS	239
В.	2.	STAE	BILITY RANGE SCRIPT	241
В.	3.	Roo	T-LOCI PLOTS	245
В.	4.	Ехро	DRT FILES FOR SABER COSMO SCOPE	249
APP	END	IX C	SABER SIMULATION MODELS	251
ſ	1	ς \λ/ιτ		251
C.	1. 2	SVII	TCHED MODEL OF THE AVERAGE-CURRENT CONTROL WITH NORMAL AND IMMEDIATE UPDA	ZJT TE
U.,	۷.	50011		יביי 252
		•••••		252

C.3.	AVERAGED MODEL OF THE AVERAGE-CURRENT CONTROL WITH NORMAL AND IMMEDIATE UPDAT	ГЕ
	· · · · · · · · · · · · · · · · · · ·	252
C.4.	SABER SCRIPTS	252
C.4.	1. Z-domain selector	252
C.4.	2. Z-domain saturation	253
C.4.	3. Z-domain event generator	253
APPEND	DIX D CONTROL IMPLEMENTATION DIAGRAMS	263
D.1.	UNIFORMLY-SAMPLED VALLEY-CURRENT, PEAK-CURRENT AND AVERAGE-CURRENT CONTROL	
IMPLEN	IENTATIONS	263
D.2.	UNIFORMLY-SAMPLED AVERAGE-CURRENT CONTROL WITH IMMEDIATE DPWM UPDATE	266
APPEND	DIX E SIMULATION AND EXPERIMENTAL RESULTS	269
APPEND	DIX F CONTROL PLATFORM SCHEMATICS	281

Final word count: 59,517

List of figures

Figure 1.1. Illustrative configuration of the electrical power-train on board of an electric vehicle
[10]
Figure 1.2. Dual interleaved boost converter with inter-phase transformer
Figure 1.3. Boost converter with analog voltage-mode control
Figure 1.4. Boost converter with analog current-mode control. (a) Peak-current mode control. (b)
Average-current mode control. (c) I ² averaged-current mode control [30]
Figure 1.5. Architecture of a digitally controlled dc-dc converter
Figure 1.6. Digital pulse-width modulator: Uniformly-sampled implementation [22]. (a)
Functional block diagram; (b) Count up mode operation; (c) Count down mode
operation; (d) Count up-down mode operation; Count down-up mode (e)
Figure 1.7. Block diagram of the model of a digital average-current controlled dc-dc converter
[37]
Figure 2.1. Dual interleaved boost converter with inter-phase transformer
Figure 2.2. Ideal steady-state waveforms of the dual interleaved boost converter with IPT. (a) D
< 0.5; (b) D = 0.5; (c) D > 0.5
Figure 2.3. Equivalent circuits of the dual interleaved boost converter with inter-phase
transformer for continuous conduction mode of operation.
Figure 2.4. (a) Single-phase boost converter switch network. (b) Switch network waveforms.
(c) Equivalent dc and ac small-signal averaged switch network model
Figure 2.5. Open loop circuit of the dual interleaved boost converter with IPT
Figure 2.6. dc and small-signal averaged model of the DIBC with IPT
Figure 2.7. MATLAB/Simulink small-signal ac model
Figure 2.8. SABER switched model of the dual interleaved boost converter with IPT
Figure 2.9. SABER averaged large-signal model of the dual interleaved boost converter with IPT.
Figure 2.10. Steady state simulation results of the open loop switched model and the average
model developed in SABER. $D = 0.25$; $V_{in} = 80$ V; $R_{Load} = 5.2 \Omega$

Figure 2.11. Steady state simulation results of the open loop switched model and the average
model developed in SABER. $D = 0.5$; $V_{in} = 80$ V; $R_{Load} = 5.2 \Omega$
Figure 2.12. Steady state simulation results of the open loop switched model and the average
model developed in SABER. $D = 0.75$; $V_{in} = 80$ V; $R_{Load} = 5.2 \Omega$
Figure 2.13. Transient response of the converter models for a synchronous 1% step increment in
the duty ratio of both phases. V_{in} = 80 V; R_{load} = 5.2 Ω . (a) D_a = D_b = 0.2; (b) D_a = D_b =
0.4; (c) $D_a = D_b = 0.6$; (d) $D_a = D_b = 0.8$
Figure 2.14. Transient response of the converter models for a synchronous 3% step increment in
the duty ratio of both phases. V_{in} = 80 V; R_{load} = 5.2 Ω . (a) D_a = D_b = 0.2; (b) D_a = D_b =
0.4; (c) $D_a = D_b = 0.6$; (d) $D_a = D_b = 0.8$
Figure 2.15. Transient response of the converter models for a synchronous 5% step increment in
the duty ratio of both phases. V_{in} = 80 V; R_{load} = 5.2 Ω . (a) D_a = D_b = 0.2; (b) D_a = D_b =
0.4; (c) $D_a = D_b = 0.6$; (d) $D_a = D_b = 0.8$
Figure 2.16. Relative error of the final average-value of $i_a(t)$ and $v_a(t)$ predicted by the small-signal
averaged model after a small step-increment in the duty-ratio. The relative error was
calculated with respect to the value predicted by the SABER switched model 98
Figure 2.17. Magnitude and phase response of the small-signal averaged model and the SABER
averaged model to small-sinusoidal perturbations in the duty-ratio. (a),(b)&(c) $D =$
0.25; (d),(e)&(f) $D = 0.75$. $V_{in} = 80 V$, $R_{load} = 5.2 \Omega$
Figure 2.18. Functional block diagram of the digital current control with UPWM for the dual
interleaved boost converter with IPT101
Figure 2.19. Waveforms of the digital current control with uniformly-sampled pulse-width
modulation from the dual interleaved boost converter with IPT. Leading-edge
modulation strategy. Valid for CCM and $D \leq 0.5$
Figure 2.20. Analysis of the digital current control modulation schemes for the dual interleaved
boost converter with inter-phase transformer105
Figure 2.21. Sampled value of the phase-a current waveform when the sampling instants are
located in the middle of the transistors on-intervals. $D < 0.5$
Figure 2.22. Sampled value of the phase-a current waveform when the sampling instants are
located in the middle of the transistors on-intervals. $D > 0.5$
Figure 3.1. DIBC topology and control platform114
Figure 3.2. Prototype of the dual-interleaved boost converter with inter-phase transformer115
Figure 3.3. Functional block diagram of the digital control platform116

Table of contents

Figure 3.4. Delfino TMS320F28335 Control Card118
Figure 3.5. Current sensor and signal conditioning for the ADC
Figure 3.6. Functional block diagram of the signal conditioning for the transistor driving signals
and the gate drivers120
Figure 3.7. Printed circuit board of the digital control platform. (a) Top layer; (b) Bottom layer.
Figure 3.8. Digital control platform122
Figure 3.9. Functional block diagram of the DSC modules124
Figure 3.10. (a) Synchronization of the DSC modules for leading-edge modulation (valley-current
control); (b) Synchronization of the DSC modules for triangular carrier modulation
with symmetric on-time (average-current control)
Figure 3.11. Valley-current, peak-current and average-current control algorithm. (a) Flow chart
of the algorithm executed during the ADC interrupt service routine. (b) Scaling and
compensation of the control-loops. (c) Compensation network: Backward-Euler
proportional–integral compensator in parallel form
Figure 3.12. Timing diagram of the turn-on instant of the Q_{1a} transistor and the DSC modules for
leading edge modulation or valley-current control
Figure 3.13. Average-current control with immediate DPWM update. (a) Control algorithm
executed during the ADC interrupt service routine. (b) Time diagram illustrating the
generation of spurious waveforms and how to avoid this issue by constraining the
minimum value of the duty-ratio132
Figure 3.14. Experimental and simulation waveforms of the dual interleaved boost converter with
IPT in steady state operation obtained employing the valley-current control
algorithm. (a) I _{ref} = 20 A, D = 0.43; (b) I _{ref} = 40 A, D = 0.62. System conditions: V _{in} = 80
V; R _{load} = 5.2 Ω
Figure 3.15. Experimental and simulation waveforms of the dual interleaved boost converter with
IPT in steady state operation obtained employing the peak-current control algorithm.
(a) I _{ref} = 20 A, D = 0.25; (b) I _{ref} = 60 A, D = 0.61. System conditions: V _{in} = 80 V; R _{load} =
5.2 Ω
Figure 3.16. Experimental measurement of the processing time of the valley-current control.
System conditions: V_{in} = 80 V; R_{load} = 5.2 Ω ; D = 0.613, I_{Lin} = 101 A, V_o = 217 V. (a) Two
complete cycles. (b) Expansion of the waveforms at the v_{1a} rising edge

- Figure 3.17. Experimental measurement of the processing time of the peak-current control.
 System conditions: V_{in} = 80 V; R_{load} = 5.2 Ω; D = 0.613, I_{Lin} = 101 A, V_o = 217 V. (a) Two complete cycles. (b) Expansion of the waveforms at the v_{1a} falling edge......141

- Figure 3.22. Experimental results of the step-load response of the peak-current current control. System conditions: I_{ref} = 60 A, V_{in} 80 V, R_{load1} = 2.6 Ω, R_{load2} = 5.2 Ω. (a) Overview of the test. (b) Magnification of the step-up transient. (c) Magnification of the step-down transient.

- Figure 3.25. Experimental measurement of the processing time of the average-current control with normal update. System conditions: $V_{in} = 80 V$; $R_{load} = 5.2 \Omega$; D = 0.613, $I_{Lin} = 101$

- Figure 4.3. Decomposition of the sampler S with sampling rate *T/n* into *n* synchronized subsamplers *S*₀ with sample rate *T*, [2]......168

- Figure 4.7. Frequency response of the closed-loop transfer function $G_{iairef}(s)$ for the SABER averaged model, the small-signal synchronous model and the small-signal nonsynchronous model. Normal update mode, $t_{d_proc} = (0.5)T = 16.66 \ \mu s$. $V_{in} = 80 \ V$, R_{load} = 5.2 Ω . (a) $I_{ref} = 13.6 \ A$, D = 0.25; (b) $I_{ref} = 30.7 \ A$, D = 0.5; (c) $I_{ref} = 123 \ A$, D = 0.75.181

- Figure 5.1. Root-locus of the poles of closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ of the small-signal non-synchronous model. Compensator: digital proportional controller with $K_p = 1T$. Ranging D from 0.2 to 0.8 and R_{load} from 1.4 Ω to 10.4 Ω . $V_{in} = 80$ V.194
- Figure 5.2. Root-locus of the poles of closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ of the small-signal non-synchronous model. Compensator: digital proportional controller. D = 0.2 to 0.8; $K_p = 0.1(T)$, 1(T), 5(T), 10(T), 20(T), 30(T); $R_{load} = 5.2 \Omega$; $V_{in} = 80 V$195

- Figure 5.5. Root-locus of the poles of closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ of the small-signal non-synchronous model. Compensator: digital PI controller. D = 0.2 to 0.8; $t_{d_proc} = 0.1(T)$, 0.3(T), 0.5(T); $K_p = 20(T)$; $K_i = 10$; $R_{load} = 5.2 \Omega$; $V_{in} = 80 V$200
- Figure 5.7. Stability-range predicted by the non-synchronous model for the design of the PI compensator. Immediate update mode, $t_{d_proc} = (0.126)T = 4.12 \ \mu s. \ D = 0.599, R_{load} = 5.2 \ \Omega, \ V_{in} = 100 \ V, \ I_{in} = 120 \ A.$ (Worst case scenario for stability of the design)......203
- Figure 5.8. Transient-response of the closed-loop transfer function $G_{iairef}(z)$ to a 1 A stepincrement in the reference-current. Small-signal non-synchronous model with immediate update mode. $t_{d_proc} = (0.126)T = 4.12 \,\mu s$. D = 0.599, $R_{load} = 5.2 \,\Omega$, $V_{in} = 100$ V, $I_{in} = 120 \,A$. (a)&(b) $K_i = 5$, (c)&(d) $K_i = 10$, (e)&(f) $K_i = 20$205

- Figure 5.11. Transient response of the converter prototype and the SABER switched model to a large step-increment in the reference-current. $V_{in} = 50 V$, $R_{load} = 5.2 \Omega$. Condition 1: $I_{ref1} = 30 A$; Condition 2: $I_{re2} = 60 A$. (a) $K_p = 10(T)$ and $K_i = 10$; (b) $K_p = 70(T)$ and $K_i = 10$. 210
- Figure 5.12. Transient response of the converter prototype and the SABER switched model to a large step-increment in the reference-current. $V_{in} = 100 V$, $R_{load} = 5.2 \Omega$. Condition 1: $I_{ref1} = 30 A$; Condition 2: $I_{re2} = 60 A$. (a) $K_p = 10(T)$ and $K_i = 10$; (b) $K_p = 70(T)$ and $K_i = 10$.

- Figure 5.15. Bode plots and root-locus of the open, digital voltage control-loop using a PI compensator. Operating poing: $V_{in} = 100 V$; $V_o = 175 V$; $R_{load} = 5.2 \Omega$; $t_{d_proc} = (1.2)T = 4.2 \ \mu s$.

Figure C.1. CARER ask smatter discuss of the switched model of the DIRC with IRT and the DSC with
Figure C.1. SABER schematic diagram of the switched model of the DIBC with IP1 and the DSC unit
employed to simulate the valley-current and the peak-current control strategies.254
Figure C.2. SABER schematic diagram of the DSC unit employed to simulate the valley-current and
the peak-current control strategies255
Figure C.3. SABER schematic diagram of the DSC unit employed to simulate the valley-current and
the peak-current control strategies. (a) DPWM A counter; (b) Phase-a control loop.
Figure C.4. SARER schematic diagram of the switched model of the DIRC with IDT and the DSC unit
Figure C.4. SABER schematic diagram of the switched model of the DibC with IPT and the DSC unit
employed to simulate the average-current control strategies with normal and
immediate update257
Figure C.5. SABER schematic diagram of the DSC unit employed to simulate the average-current
control strategies with normal and immediate update
Figure C.6. SABER schematic diagram of the DSC unit employed to simulate the average-current
control strategies with normal and immediate update. (a) DPWM A counter; (b)
Phase-a control loop259
Figure C.7. SABER schematic diagram of the average model of the DIBC with IPT and the DSC unit
employed to simulate the average-current control strategies with normal and
immediate update260
Figure C.8. SABER schematic diagram of the DSC unit employed to simulate the average-current
control strategies with normal and immediate update
Figure C.9. SABER schematic diagram of the current and voltage control loops added to the DSC
unit of Figure C.5
Figure D.1. Simulink diagram of the top-level tasks of the control algorithms
Figure D.2. Simulink diagram of the ADC interrupt service routine used for the normal DPWM
update control algorithms
Figure D.4. Simulink diagram of the ADC scaling process
Figure D.5. Simulink diagram of the sten-reference generator 265
Figure D.2. Simulink diagrams of the surrent control loops used for the normal DDWM undate
rigure D.S. Simulink diagrams of the current control-loops used for the normal Dewin update
control algorithms265
Figure D.6. Simulink diagram of the ADC interrupt service routine used for the immediate DPWM
update control algorithm266
Figure D.7. Simulink diagrams of the current control-loops used for the immediate DPWM update
control algorithm267

Figure D.8. Simulink diagram of the anti-glitch algorithm......268

- Figure E.3. Simulation results employed to validate the stability range predicted by the smallsignal synchronous model illustrated in Figure 4.10, Chapter 4. Initial test conditions $K_p = 30 K_i = 20$. V_{in} = 85, R_{load} = 5.2 Ω , D = 0.5480......273
- Figure E.4. Simulation results employed to validate the stability range predicted by the smallsignal synchronous model illustrated in Figure 4.10, Chapter 4. Initial test conditions $K_p = 30 K_i = 20$. V_{in} = 85, R_{load} = 5.2 Ω , D = 0.5480......274

Figure E.10. Experimental results employed to validate PI design given in Chapter 5, Section	ı 5.4.
$K_p = 70T K_i = 10. t_{d_proc} = (0.12)T, R_{load} = 5.2 \Omega, I_{ref1} = 30 A, I_{ref2} = 60 A.$ (a) $V_{in} = 50 V_{in}$	′. (b)
V _{in} = 100 V.	280
Figure F.1. Control platform schematics, scaling and conditioning stage	282
Figure F.2. Control platform schematics, power supplies	283
Figure F.3. Control platform schematics, gate drivers.	284

List of tables

Table 1.1.	Laplace-domain and frequency-domain models for uniformly-sampled pulse-width
	modulations as described in [37] by Van de Sype et al53
Table 2.1.	Summary of ripple current expressions for the DIBC with IPT. $D_a = D_b = D$
Table 2.2.	Transfer functions of the small-signal averaged model of the dual interleaved boost
	converter with IPT
Table 2.3.	Transfer functions of the small-signal averaged model of the dual interleaved boost
	converter with IPT
Table 2.4.	Simulation parameters for the validation of the small-signal model of the DIBC with
	IPT
Table 2.5.	Comparison of the theoretical and simulation waveforms for $D = 0.25$, $D = 0.5$ and
	<i>D</i> = 0.75
Table 2.6.	Equations of the line segments comprising $i_a(t)$ valid for D < 0.5 (Figure 2.19)108
Table 2.7.	Slopes of the line segments comprising $i_a(t)$ 108
Table 2.8.	Equations of the line segments comprising $i_a(t)$ valid for $D < 0.5$ (Figure 2.20)110
Table 3.1.	Parameters of the DIBC prototype
Table 3.2.	Principal Features of the Texas Instruments TMS320F28335 [2]118
Table 3.3.	Value of the timings shown in Figure 3.11130
Table 3.4.	Simulation parameters for valley-current and peak-current control
Table 3.5.	Comparison of the duty-ratio of the experimental and simulation results of the valley-
	current control algorithm (Figure 3.13)136
Table 3.6.	Phase-current measurements of the experimental and simulation results of the valley-
	current control algorithm. (Figure 3.14)136
Table 3.7.	Comparison of the duty-ratio of the experimental and simulation results of the peak-
	current control algorithm (Figure 3.15)139
Table 3.8.	Phase-current measurements of the experimental and simulation results of the peak-
	current control algorithm. (Figure 3.14)139

Table 3.9. Experimental timings of the valley-current and the peak-current control algorithms.
Table 3.10. Simulation parameters average-current control. 150
Table 3.11. Comparison of the duty-ratio of the experimental and simulation results of the
average-current control algorithm with immediate update (Figures 3.23 and 3.24).
Table 3.12. Phase-current measurements of the experimental and simulation results of the
average-current control algorithm with immediate update. (Figures 3.23 and 3.24).
Table 3.13. Experimental timings of the average-current control algorithm with normal and
immediate update155
Table 4.1. Constituent transfer functions of the digital average-current controlled DIBC with IPT
in the z-domain
Table 4.2. Closed-loop transfer functions of the digital average-current controlled DIBC with IPT
in the z-domain174
Table 4.3. Attributes of the converter and the DSC employed for the simulations. 176
Table 4.4. Parameters of the TDSA tool employed in SABER to obtain the frequency response of
the model to small-sinusoidal perturbations
Table 4.5. Comparison of the magnitude and frequency location of the resonant peak in the
frequency-response of the SABER averaged model, the small-signal synchronous
model and the small-signal non-synchronous models
Table 4.6. Comparison of the stability limits predicted by the non-synchronous model and the
SABER averaged model. (Figure 4.10)187
Table 5.1. Root-locus plots generated to analyse the behaviour of the closed-loop poles of the
small-signal non-synchronous model194
Table 5.2. Range of operation of the converter for the design of the digital PI controller202
Table 5.3. Extremes of operation for the design of the digital PI controller
Table 5.4. Attributes of the converter and the DSC employed for the simulations.
Table 5.5. Performance of the step-response of the closed-loop transfer function G _{iairef} (z). (Figure
5.8 and 5.9)
Table A.1. Constituent transfer functions of the digital average-current controlled DIBC236
with IPT in the z-domain assuming the synchronous operation of the samplers S_a and S_b

List of symbols

Symbol	Meaning
$\langle i_1(t) \rangle_{_{TSW}}, \langle i_2(t) \rangle_{_{TSW}}$	Local average of the current flowing through port 1/port 2 of the boost converter switch network over a switching period.
$\langle \mathbf{v}_1(t) \rangle_{_{TSW}}, \langle \mathbf{v}_2(t) \rangle_{_{TSW}}$	Local average of the voltage across port 1/port 2 of the boost converter switch network over a switching period.
Ααν	State-space matrix of the averaged small-signal model.
A _{AV}	Gain of the level-shifter for the ADC module.
B _{av}	Input matrix of the averaged small-signal model.
<i>C</i> (<i>z</i>)	Compensator transfer function.
Cav	Output matrix of the small-signal averaged model.
C _{in}	Converter input capacitance.
Co	Converter output capacitance.
D	Quiescent duty-ratio of the converter assuming $D_a = D_b = D$.
d(t)	Instantaneous duty-ratio of a simple boost converter.
ã	Small-signal component of a simple boost converter duty-ratio in the time-domain.
D'	Complement of the quiescent duty-ratio of the converter, (1 - D).
D_a, D_b	Quiescent duty-ratio of phase-a/phase-b.
$ ilde{d}_a$, $ ilde{d}_b$	Small-signal component of the phase-a/phase-b transistor duty-ratio in the time-domain.
$\tilde{d}_a(s)$, $\tilde{d}_b(s)$	Small-signal component of the phase-a/phase-b transistor duty-ratio in the Laplace-domain.
$\tilde{d}_a(z), \tilde{d}_b(z)$	Small-signal component of the phase-a/phase-b transistor duty-ratio in the z-domain.
$d_{a,n}, d_{b,n}$	Duty-ratio of phase-a/phase-b at interval n.
d _a [n], d _b [n], d _x [n]	Digital representation of phase-a/phase-b duty-ratio.
D _{av}	Feed forward matrix of the small-signal averaged model.
D _{min}	Minimum duty-ratio constraint for immediate-update DPWM operation.
f	Converter switching frequency/sampling frequency.
f*(s)	Laplace-transform of an impulse sampled function <i>f(t)</i> .
f _{clk}	Frequency of the master clock signal of the digital signal controller.
fclk_DPWM	Frequency of the clock signal of the DPWM modules.
G _{daia} (s)	Phase-a control-to-phase-a current transfer function in the Laplace- domain.

G _{daia} (z)	Phase-a control-to-phase-a current transfer function in the z-domain.
G _{daib} (s)	Phase-a control-to-phase-b current transfer function in the Laplace- domain.
G _{daib} (z)	Phase-a control-to-phase-b current transfer function in the z-domain.
G _{davo} (s)	Phase-a control-to-output voltage transfer function in the Laplace- domain.
G _{dbia} (s)	Phase-b control-to-phase-a current transfer function in the Laplace- domain.
G _{dbia} (z)	Phase-b control-to-phase-a current transfer function in the z-domain.
G _{dbib} (s)	Phase-b control-to-phase-b current transfer function in the Laplace- domain.
G _{dbib} (z)	Phase-b control-to-phase-b current transfer function in the z-domain.
G _{dbvo} (s)	Phase-b control-to-output voltage transfer function in the Laplace- domain.
G _{di} (z)	Direct control-to-phase current transfer function in the z-domain.
$G_{di}H_a(z)$, $G_{di}H_b(z)$	Constituent transfer functions of the converter control loop in the z- domain. See Table 4.1.
G _{dxi} (z)	Cross-coupling control-to-phase current transfer function in the z-domain.
Gho(s)	Zero-order hold transfer function in the Laplace domain.
gho(t)	Zero-order hold response to an impulse function in the time-domain.
Giairef(Z), Gibiref(Z)	Closed-loop transfer functions of the converter in the z-domain.
G _{izia} (s)	Disturbance current-to-phase-a current transfer function.
G _{izib} (s)	Disturbance current-to-phase-b current transfer function.
G _{izvo} (s)	Disturbance current-to-output voltage transfer function.
G _M (s)	DPWM transfer function in the Laplace-domain.
$G_{dxi\theta}(z), G_{dxi\phi}H_a(z)$	Constituent transfer functions of the converter control loop in the z- domain. See Table 4.1.
$G_{_{dxi\phi}}(z)$, $G_{_{dxi\theta}}H_{_b}(z)$	Constituent transfer functions of the converter control loop in the z- domain. See Table 4.1.
$H_a(s), H_b(s)$	Current-transducer transfer functions in the Laplace-domain.
$H_a(z), H_b(z)$	Current-transducer transfer functions in the z-domain.
I	Identity matrix.
i1(t), i2(t)	Instantaneous current flowing through port 1/port 2 of the boost converter switch network.
l ₁ , l ₂	DC component of the current flowing through port 1/port 2 of the boost converter switch network.
\tilde{i}_1 , \tilde{i}_2	Small-signal component of the current flowing through port 1 of the boost converter switch network.
i _a (t), i _b (t), i _x (t)	Instantaneous phase-a/phase-b current.
i _a *(t), i _b *(t), i _{ref} *(t)	Sampled phase-a, phase-b and reference currents.
Ia, Ib	Average value or dc component of the phase-a/phase-b current.

\tilde{i}_a , \tilde{i}_b	Small-signal component of the phase-a/phase-b current in the time- domain.
$\tilde{i}_a(s)$, $\tilde{i}_b(s)$	Small-signal component of the phase-a current in the Laplace-domain.
$\tilde{i}_a(z)$, $\tilde{i}_b(z)$	Small-signal component of the phase-a current in the z-domain.
i _a [n], i _b [n], i _x [n]	Digital representation of the Instantaneous phase-a/phase-b current.
İ _{a0}	Intersection of the $i_{a1}(t)$ segment with the y-axis.
i _{a1} (t), i _{a2} (t),, i _{a5} (t)	Line-segment functions comprising the phase-a current.
ĩ,	Small-signal component of the output capacitor current.
i _{diff} (t)	Instantaneous differential current, defined as half of the difference of the inter-phase transformer winding currents.
I _{Lin}	Average value or dc component of the input inductor current.
i _{Lin} (t)	Instantaneous input inductor current.
i _o (t)	Instantaneous converter overall output current. (Delivered to the load resistor).
lo	dc component of the converter overall output current. (Delivered to the load resistor).
ĩ _o	Small-signal component of the converter overall output current. (Delivered to the load resistor).
I _{oa} , I _{ob}	dc component of the output current flowing through the switch networks A and B of the converter.
\tilde{i}_{oa} , \tilde{i}_{ob}	Small-signal component of the output current flowing through the switch networks A and B of the converter.
l _{oab}	dc component of the converter output current. (Delivered to the output filter).
Ĩ _{oab}	Small-signal component of the converter output current. (Delivered to the output filter).
i _{ref} (t)	Instantaneous reference-current.
$\tilde{i}_{ref}(z)$	Small-signal component of the reference-current in the z-domain.
$\tilde{i}_{ref heta}(z)$	Small-signal component of the delayed reference-current in the z-domain.
i _{ref} [n]	Digital representation of the instantaneous reference-current.
ĩz	Small-signal component of the perturbation current in the time-domain.
$\tilde{i}_z(s)$	Small-signal component of the perturbation current in the Laplace- domain.
i	Imaginary unit. $j^2 = -1$.
k	Coupling coefficient.
Ki	Integral gain of a PI compensator.
K _n	Conversion ratio of the hall-effect current sensors.
Κρ	Proportional gain of a PI compensator.
Крым	DPWM gain.

La	IPT winding a self-inductance.
L _b	IPT winding b self-inductance.
L _c	Self-inductance of the inter-phase transformer windings when $L_a = L_b = L_c$.
L _{diff}	Differential inductance of the inter-phase transformer windings.
Lin	Inductance of the Input inductor.
Lm	Mutual inductance of the inter-phase transformer windings.
L _{Tot}	Sum of inductances: $L_{Tot} = 2L_{in}(L_c + L_m) + (L_c^2 - L_m^2)$
<i>m</i> ₁ , <i>m</i> ₂ ,, <i>m</i> ₅	Slopes of the line-segments comprising the phase-a current.
N _{ADC}	Resolution of the ADC in bits.
Po	Output power.
R_A, R_B, R_C, R_D	Values of the resistor network comprising the level-shifting stage for the ADC.
R _{in}	Input inductor parasitic resistance.
R _{load}	Converter load resistance.
R_m	Measurement resistor for current sensor.
R _{mod}	Mode selector resistor for the gate-drivers.
R _{wa} , R _{wb}	IPT Winding a/b resistance.
S	Generic sampler.
<i>S</i> ₀	Slow-rate sampler.
S_{a} , S_{b}	Phase-a, phase-b samplers.
Т	Converter switching period/sampling period.
T(z)	Open-loop transfer function.
t _{ADC}	Total conversion time of the ADC.
t _{ADC_INT}	Length of time of the ADC interrupt service routine.
t _d	Time that the ADC takes from the request of a conversion to the start of the sampling sequence.
t _{d(on)}	IGBT turn-on delay.
t _{d(sch)}	Time that takes for the results of the analogue-to-digital conversion to appear in the result-registers of the ADC module.
t _{d_proc}	Time that the digital controller takes from the start of an analogue-to- digital conversion to the calculation of a new duty ratio for the corresponding control-loop.
t _{dQ}	Time length from the moment where the DPWM channel is set high to the moment where its corresponding transistor fully conducts the phase-current.
tint_srvc	Time that the processor takes from the moment where an interrupt is acknowledged to the start of the interrupt-service routine.
t_{p_gd}	Gate-driver propagation delay.
t _{r(on)}	IGBT current rise-time.
tsн	Width of the sample-and-hold acquisition window of the ADC module.

T _{SH}	Sampling period of the ADC module sampling sequence.
ũ	Input-vector of the averaged small-signal model in the time-domain.
u _s (t)	Unit-step function.
v1(t), v2(t)	Instantaneous voltage measured at port 1/port 2 of the boost converter switch network.
V ₁ , V ₂	dc component of the voltage measured at port 1/port 2 of the boost converter switch network.
\tilde{v}_1, \tilde{v}_2	Small-signal component of the voltage measured at port 1 of the boost converter switch network.
V _{ADCmax}	Maximum analog input voltage of the ADC.
V _{cmax}	Maximum count of the DPWM module digital counters.
Vdiff(pkpk)	Differential peak-to-peak voltage.
v _{diff} (t)	Instantaneous differential voltage.
V _{ge1a} (t), V _{ge1b} (t), V _{ge2a} (t), V _{ge2b} (t)	Instantaneous IGBT gate-to-emitter voltages.
v _{ia} (t), v _{ib} (t), v _{ix} (t)	Conditioned voltage signals representing the phase-currents.
v _{ia} [n], v _{ib} [n], v _{ix} [n]	Digital representation of the conditioned voltage signals representing the phase-currents.
v _{in} (t)	Instantaneous input voltage of the converter.
V _{in}	Average value or dc component of the converter input voltage.
<i>V</i> _{in}	Small-signal component of the converter input-voltage.
V _{Lin(pkpk)}	Input inductor peak-to-peak voltage.
v _{Lin} (t)	Instantaneous voltage across the input inductor.
<i>v_o(t)</i>	Instantaneous output voltage of the converter.
Vo	Average value or dc component of the converter output voltage.
<i>v</i> _o	Small-signal component of the converter output voltage in the time- domain.
$\tilde{v}_o(s)$	Small-signal component of the converter output voltage in the Laplace- domain.
y*(t) _n	Original output signal of a decomposed sampler.
$y_{0}^{*}(t), y_{1}^{*}(t),, y_{p}^{*}(t)$	Partial output signals of a decomposed sampler.
x	State-vector of the averaged small-signal model in the time-domain.
ΔI_{diff}	Amplitude of the differential current ripple.
ΔI _{Lin}	Amplitude of the input inductor current ripple.
ΔI_{lrg}	Peak-to-peak value of the largest ripple in the phase-a/phase-b current.
ΔI_{smll}	Peak-to-peak value of the smallest ripple in the phase-a/phase-b current.
$\Delta PWM(s)$	DPWM delay transfer function.
ω	Angular frequency.

List of Abbreviations

Abbreviation

Meaning

A/D	Analogue-to-Digital
ас	Alternating current
ADC	Analogue-to-Digital Converter
ADC_ISR	Analogue-to-Digital Converter Interrupt Service Routine
ССМ	Continuous Current Mode
CPU	Central Processing Unit
D/A	Digital-to-Analogue
DAC	Digital-to-Analogue Converter
dc	Direct Current
DCM	Discontinuous Current Mode
DIBC	Dual-Interleaved Boost Converter
DPWM	Digital Pulse Width Modulator
DSC	Digital Signal Controller
EOC	End-Of-Conversion
EV	Electric Vehicle
FPGA	Field Programmable Gate Array
HEV	Hybrid Electric Vehicle
IGBT	Insulated Gate Bipolar Transistor
IPT	Inter-Phase Transformer
LCO	Limit Cycling Oscillation
MOSFET	Metal-Oxide Semiconductor Field-Effect Transistor
MSPS	Mega-Samples Per-Second
PI	Proportional Integral
PWM	Pulse Width Modulator
SOC	Start-Of-Conversion
TDSA	Time-Domain System Analyser
WWF	World Wildlife Fund
ZOH	Zero Order Hold
ZVS	Zero-Voltage Switching

Abstract

This Thesis addressed the challenge of ensuring balanced currents in the phases of a multi-kW, interleaved dc-dc converter by means of closed-loop digital control.

The Thesis examines uniformly-sampled, valley-current, peak-current and averagecurrent control for a dual interleaved boost converter with inter-phase transformer which might form part of the power train of an electric vehicle. Also, an enhancement of the average-current control is investigated in which the transistor duty-ratio is updated more rapidly, which allows an improvement of approximately ten times in the response speed of the system. Based on the theoretical analysis, the average-current control methodology was determined to be the most suitable technique for this type of converter as it ensures well-balanced phase currents over a wide range.

To provide a basis for control system analysis and design for interleaved converters, a modelling methodology is developed based on a combination of multi-rate data-sampled theory and a small-signal averaged converter model. The model is shown to represent accurately the interaction between the interleaved phases, revealing a reduced stability range compared with a non-interleaved converter.

The modelling and control methods are validated using switched and average value simulations obtained with the SABER software and by experimental results from a 25 kW, 30 kHz converter prototype. The control techniques were implemented on a Texas Instruments TMS320F28335 digital signal controller.

Declaration

No portion of the work referred to in this Thesis has been submitted in support of an application for another degree or qualification of this or any other university or other institute of learning.

Copyright statement

The author of this thesis (including appendices and/or schedules to this thesis) owns certain copyright or related rights in it (the "Copyright") and s/he has given The University of Manchester certain rights to use such Copyright, including for administrative purposes.

Copies of this thesis, either in full or in extracts and whether in hard or electronic copy, may be made only in accordance with the Copyright, Designs and Patents Act 1988 (as amended) and regulations issued under it or, where appropriate, in accordance with licensing agreements which the University has from time to time. This page must form part of any such copies made.

The ownership of certain Copyright, patents, designs, trade marks and other intellectual property (the "Intellectual Property") and any reproductions of copyright works in the thesis, for example graphs and tables ("Reproductions"), which may be described in this thesis, may not be owned by the author and may be owned by third parties. Such Intellectual Property and Reproductions cannot and must not be made available for use without the prior written permission of the owner(s) of the relevant Intellectual Property and/or Reproductions.

Further information on the conditions under which disclosure, publication and commercialisation of this thesis, the Copyright and any Intellectual Property and/or Reproductions described in it may take place is available in the University IP Policy (see http://documents.manchester.ac.uk/DocuInfo.aspx?DocID=487), in any relevant Thesis restriction declarations deposited in the University Library, The University Library's regulations (see http://www.manchester.ac.uk/library/aboutus/regulations) and in The University's policy on Presentation of Theses.

Dedicated to my parents, Vicente and Laura Velia, and my dearest sister, Laura, who have supported, inspired and encouraged me throughout my life.

"Caelum, non animum, mutant qui trans mare currunt" - Horace

Acknowledgements

First, my sincere gratitude to my supervisor, Professor Andrew J. Forsyth, for his guidance, support and insightful advice given throughout the development of this research project. Also, I would like to thank him for the time he dedicated to provide invaluable suggestions during the writing stage of this Thesis.

I am also grateful to the National Council of Science and Technology of Mexico (CONACyT), the National Polytechnic Institute of Mexico (IPN), and the Public Ministry of Education of Mexico (SEP), for their financial support to pursue my PhD degree.

I thank my parents and sister, Vicente, Laura Velia and Laura for all their love, help and support that carried me through difficult times.

Special thanks to Dr Gerardo Calderon-Lopez for his experience, useful discussions and friendship. Also I would like to thank my friends Jesus Carmona and Dr Damian Vilchis for the moments together.

I am very grateful with my friends Gillian Menichino, Carlos A. Duran, Jana Kefurtova and Alexander Cassell for their invaluable support and the great moments we lived together.

Last, I would like to thank Dr Ismael Araujo-Vargas for the encouragement to pursue a PhD degree.

Chapter 1 Introduction and literature review

1.1. Introduction

With terrestrial transport accounting for around one quarter of the UK's total CO_2 emissions [1], the transition to low carbon vehicle technologies is critical to meeting the targets set by the UK Climate Change Act. The Act commits the UK to reduce in 2050 greenhouse gas emissions by at least 80 % compared with 1990 levels [2], and to meet this objective, the CO_2 emissions have to be reduced in 2020 by at least 34 % relative to those generated in 1990 according to a report presented to the World Wildlife Fund UK in 2010 (WWF UK) [3].

1.1.1. Electric vehicles

Due to the growing concerns for the reduction of CO₂ emissions and for the depletion of fossil fuels [4], significant progress has been made during the past two decades in the development of electric vehicles, EVs. However, further improvements in range and reductions in cost are needed to make the technology more attractive and competitive in the market [5].

The range of an EV is limited by the capacity of the on-board power source and the efficiency of the systems that form the electrical power-train [6, 7]. The use of high-capacity batteries and/or fuel-cell storage systems has been a common solution proposed, often accompanied by a bank of super-capacitors or a flywheel in order to meet the high-power surges demanded by the traction drive and to absorb the energy from braking. An alternative approach which has had better success in the market in recent years, is the

Chapter 1

combination of an internal combustion engine with an electric propulsion system. Such vehicles are known as Hybrid Electric Vehicles, HEVs [5, 8].

A large number of electrical power-train architectures exist depending on the combinations of energy sources employed in the vehicle. A study carried out in 2006 [9] identifies at least 30 different architectures. For illustrative purposes Figure 1.1 depicts an overview of the power-train on an electric vehicle which can be powered by fuel-cells, batteries or both [10]. The voltage provided by the battery/fuel-cell and super-capacitors has to be stepped up and regulated by one or more bidirectional/unidirectional dc-dc converters. The regulated voltage is delivered into a high-voltage dc-bus. A dc-ac converter is supplied from the dc-bus to drive the traction motor of the EV.



Figure 1.1. Illustrative configuration of the electrical power-train on board of an electric vehicle [10].

Besides high-power capacity and bidirectional power flow capabilities, the dc-dc converters employed in the electrical power-train must be light-weight, highly-efficient, small-sized, highly-reliable, have low electromagnetic interference and have low current-ripple [10, 11]. To meet these demands switching frequencies are being increased, often through the adoption of new device technologies such as silicon carbide [12-14], and different circuit topologies are under investigation. These topologies can be hard-switched, soft-switched, multi-phase/interleaved, multi-port, isolated, or non-isolated depending on the requirements of the application [15].

The research presented in this thesis is based on a hard-switched, non-isolated topology named dual-interleaved boost converter with inter-phase transformer shown in Figure 1.2. This converter topology has been proposed by several researchers to interface the different voltage levels of fuel cells, super capacitors, battery banks and traction drives

within an electrical vehicle power train, mainly due to its high power-density capabilities [16-19], and was used in this work as a great deal of research work has been invested in this converter topology within the Power Conversion research group of the University of Manchester and several prototypes were readily available for their use.



Figure 1.2. Dual interleaved boost converter with inter-phase transformer.

Furthermore, to achieve the necessary voltage regulation and manage the bidirectional power flow of these increasingly complex, high frequency converters, high performance controllers are required. Due to the relatively high operating frequency typically above 20 kHz, dc-dc converters tend to be controlled by relatively simple analogue Whilst these methods offer high-bandwidth, real-time operation based methods. capabilities and theoretically infinite voltage resolution, they suffer from a number of limitations: they are sensitive to noise and temperature variations which are common in automotive applications, they usually comprise passive devices that are subject to ageing and have to be replaced/exchanged in order to perform modifications, such as parameter changes, and they require a prohibitive number of components when complex control algorithms have to be implemented thereby increasing cost and size. Furthermore, these characteristics can lead to poor reliability. However, the evolution of microprocessors and microcontrollers into faster and more versatile digital signal controllers and the advent of Field Programmable Gate Arrays, FPGAs, has created the possibility of dc-dc converter control through digital means.

Digital control provides a way to overcome many of the limitations of analogue control through the use of software that is more reliable and that provides the flexibility to

Introduction and literature review

Chapter 1

change parameters in real-time. It also facilitates the implementation of complex and advanced control algorithms without the need of extra components. Other advantages are insensitivity to component ageing, robustness to noise, the possibility to run self-diagnosis and tests, and the ability to interface directly with other digital systems embedded in the vehicle, thereby leading to a full digital implementation of the control system of an EV. However, the limited resolution of analog-to-digital converters and digital pulse-width modulation modules together with the processing delay caused by the digital signal controller, can have detrimental effects on the control bandwidth and the overall system performance [20]. For these reasons, there has been an increased interest in digital control in power converter applications generally and several studies have proposed new techniques and architectures to overcome these performance issues.

1.1.2. Objectives

The overall objective of the research reported in this thesis is to devise and demonstrate digital control techniques for dual-interleaved dc-dc converters that might form part of an EV power train, and to develop and validate a theoretical basis for the analysis and design of such systems. The key challenges include ensuring balanced currents in the two interleaved phases by means of cycle-by-cycle current control and representing accurately the interleaved current sharing in the system model.

1.1.3. Structure of the literature review

The literature survey is organized in five sections. In Section 1.2, a review of the most common control methodologies for simple dc-dc converters is presented as most of the research on digital control has been carried out for basic topologies and low-power applications. In Section 1.3, a review of digital control methods for multi-phase dc-dc converter topologies is undertaken since the converter studied in this thesis is a variation of a multi-phase dc-dc converter. Section 1.4 introduces the modelling techniques for digitally controlled dc-dc converters. In Section 1.5, the linear compensator design methodologies are reviewed. Finally, in Section 1.6, a survey of the most relevant modelling techniques for multi-phase dc-dc converters is presented focused on the dual-interleaved converter.

Chapter 1

1.2. Control for switched dc-dc converters

The main objective of dc-dc converter control is usually the regulation of the outputvoltage and this is typically achieved through a closed-loop control system that feeds back the output-voltage, and sometimes other circuit variables as well, to adjust the on-time-toperiod ratio or duty-ratio of the transistors. Two figures-of-merit that are often used to quantify the performance of dc-dc converter controllers are based on the systems ability to regulate the output-voltage in the presence of variations in the input voltage and/or load current: the line-regulation and the load-regulation. The former is defined as the change in the output voltage corresponding to a specific change in the input voltage, whilst the latter is defined as the change in the output voltage to a specific change in the load current [21, 22]. The controller implementation may be either analogue or digital.

1.2.1. Analog control techniques

Analogue control techniques for dc-dc converters are well-established and mature. Although numerous control approaches have been published in the literature, two main control strategies can be identified: voltage-mode control and current-programmed control [21-24].

1.2.1.1. Voltage-mode control

Voltage-mode control in its most simple form incorporates a single control-loop to regulate the output voltage of the converter. Figure 1.3 illustrates a schematic diagram where the voltage-mode control is implemented for a boost or step-up converter. The output-voltage of the converter, v_o , is fed back and compared against a target constant voltage-reference, v_{ref} . The error between the signals, v_{err} , is passed through an analog compensation network, H(s), which in its simplest form can be lead, lag or a lead-lag compensator. The compensator network generates a control signal, v_m , forming the input to the PWM modulator, which in its simplest form consists of a saw tooth waveform generator and a comparator. The compensation network usually comprises one or more operational amplifiers plus passive components. Whilst being simple and easy to implement, this architecture is not particularly effective at compensating for rapid load- and
line-variations as these will take some time to be reflected in the output-voltage, resulting in a slow response, large voltage deviations and/or the requirement for a large output filter capacitor [21].



Figure 1.3. Boost converter with analog voltage-mode control.

A variant of this architecture is sometimes used to improve immunity to linevariations [24]. In this variant, the input-voltage is employed as a feed-forward signal to adjust the slope of the modulator ramp. In consequence, any change in the input voltage is compensated for almost immediately, improving the overall dynamic response of the system.

1.2.1.2. Current-mode control

Many types of current-mode control are described in the literature which may be divided in two categories: fixed-frequency and variable-frequency, fixed frequency being the most common. Figure 1.4(a) illustrates the architecture of the peak-current mode control for a boost converter which is a form of fixed-frequency control. There are two control loops an outer voltage-loop and an inner current-loop. The inner control loop regulates the peak current or maximum current circulating through the input inductor, whilst the outer control loop provides a reference current for the inner control loop.

In contrast to voltage-mode control, the duty-ratio of the converter transistor is generated by comparing the sensed inductor current against a threshold value. This is commonly accomplished by employing an analog comparator. When the inductor current

reaches the threshold, the comparator resets the state of an SR-latch, which consequently turns the transistor off. A fixed-frequency clock-pulse is fed into the set terminal of the SR-latch which turns the transistor on at the start of each cycle. The constant activation/deactivation of the SR-latch produces a fixed-frequency driving signal. The outer-loop operates in a similar way to voltage-mode control. The sensed output voltage is fed-back and compared with a target reference. The error signal generated is then passed through a compensation network which will produce the reference current signal.



Figure 1.4. Boost converter with analog current-mode control. (a) Peak-current mode control. (b) Average-current mode control. (c) I² averaged-current mode control [30].

Employing this technique can be problematic in that instability is caused by the propagation of sub-harmonic oscillations when the converter operates at duty-ratios above 0.5. To address this issue, a compensating ramp is added to the reference current before it is compared against the inductor current. The slope of the compensating ramp is selected to ensure stability of the system over a determined range of operation [25].

The advantages of the current-mode control can be summarized as: immediate response to input voltage variations effectively providing input-voltage feed-forward, reduction of the order of the system transfer functions since the operation of the input inductor is similar to that of a voltage-controlled current source, inherent over-current protection, and inherent load-sharing when several converters are employed in parallel. The main disadvantage of this type of control is its susceptibility to switching noise generated by the transistor switching transients which might generate current over-shoots larger than peak-current and that can result in erratic operation and instability [21, 23, 24].

A number of variations on this control method have been proposed, for example using fixed on-time or fixed off-time in the current control loop as a free-running hysteresis technique [21]. Also an integrator is sometimes included in the current control loop to force the average inductor current to follow the current reference signal. This technique is known as average-current mode control [26], Figure 1.4(b), and is often employed in applications where the converter input current must be shaped or accurately controlled, for example in power factor correction circuits [27].

More recently (2014), a new control method termed l² average-current mode control has been proposed in [28] to improve the transient response of average current mode control and also improve light-load efficiency through using fixed on-time, variable switching frequency. Figure 1.4(c) shows a diagram of a boost converter with this control where an outer voltage loop is employed for voltage regulation. This technique combines both the fast-direct feedback of the peak-current mode control to ensure fast transient response with the slow integral feedback of the average-current mode control to achieve an accurate control of the inductor current. Moreover, by the use of fixed on-time modulation in conjunction with this technique, the propagation of sub-harmonic oscillations is avoided thereby eliminating the requirement for a compensating ramp.

1.2.2. Digital control techniques

A direct approach to form a digital controller for a dc-dc converter would be to reproduce digitally the well-known analog control techniques. However, such implementation yields an expensive solution as highly-specialized components would be

required to achieve the necessary sampling and processing speeds. For this reason, research effort has been devoted to finding cost-effective control solutions employing market available Digital Signal Controllers, DSC. The following section will review the architectures employed to control a dc-dc converter employing a DSC. Moreover, the operation and limitations of the techniques are considered.

1.2.2.1. Architecture and elements of a digital controlled dc-dc converter

Figure 1.5 illustrates the key-components that form a digital controlled dc-dc converter. As seen in this diagram, the power stage is interfaced with the DSC by scaling and conditioning stages in the feedback path and the gate-driver for the power-stage transistor(s). The DSC has three main elements: the Analog-to-Digital Converter, ADC, the Central Processing Unit, CPU, and the Digital Pulse-Width Modulator. Most of the DSCs available in the market have at least one ADC module and several DPWM modules [29, 30].



Digital Signal Controller

Figure 1.5. Architecture of a digitally controlled dc-dc converter.

The ADC converts the feedback signals into a digital representation. The typical sampling speeds of the ADC modules embedded in microcontrollers for motor-control and power converter applications are in the range of 4 MSPS to 12.5 MSPS [29, 30]. These figures correspond to conversion times of 250 ns to 80 ns respectively. The conversion speed of the ADC contributes to the processing delay of the controller and increases with the number of signals converted.

Following the analog-to-digital conversion process, the CPU executes the control calculation and determines the duty-ratio of the converter. The controller can be programmed to execute many different algorithms to ensure that the control target is met. However, the processing speed of the CPU is limited and it increases with the complexity of the algorithms implemented. Typical clock-frequencies of available DSCs range from *40 MHz* to *300 MHz* [*29, 30*]. The duty-ratio determined by the control algorithm, is converted into a transistor driving signal by means of the DPWM. The most common implementation of DPWM found in the DSCs available in the market is uniformly-sampled PWM, UPWM.

Figure 1.6(a) illustrates the functional block diagram of a uniformly-sampled PWM, whilst Figure 1.6(b), 1.6(c) and 1.6(d) show the key waveforms for different modes of operation. This implementation is directly derived from a naturally-sampled PWM architecture by substituting its main elements for digital components [22]. The operation is explained for the case where the binary-counter is programmed in count up mode and with the aid of the plots in Figure 1.6(b).

Under count up mode, the binary counter generates a digital leading-edge ramp waveform, v_c . The binary counter is incremented at the rising-edge of the DPWM clocksignal. The DPWM clock is commonly sourced from the master-clock signal of the digital signal controller which may be divided down by means of a prescaler circuit to achieve slower switching frequencies. v_c is constantly compared against a reference-register which contains the value of the programmed duty-ratio, v_m , sometimes called the modulating signal. This is accomplished by means of a binary comparator. When the value of v_c crosses v_m , an interrupt-request is triggered by the digital comparator. The interrupt-request is used for two purposes: to set the DPWM output-signal, v_{out} , to zero and to inform the processor of this event. When v_c reaches its maximum value, v_c max, it is reset to zero and v_{out} is set high. At the same time, the reference-register containing v_m is updated with the new programmed duty-ratio. Since v_m is updated only once per switching period, it is plotted as a constant value during the whole v_c cycle. In contrast, in an analogue naturallysampled PWM generator, the duty-ratio signal or modulating signal varies continuously, and as a result the digital, uniformly sampled, PWM exhibits an additional delay, which, for high duty-ratio pulses can approach a switching cycle for leading-edge and trailing-edge

modulation and half switching cycle for the triangular carrier modulation. This effect can limit the achievable control bandwidth when using digital control [31-33].



Figure 1.6. Digital pulse-width modulator: Uniformly-sampled implementation [22]. (a) Functional block diagram; (b) Count up mode operation; (c) Count down mode operation; (d) Count up-down mode operation; Count down-up mode (e).

In addition to the count up mode of operation, it is also possible to program the binary-counter in two alternative modes: count down mode and count up-down mode. In the former, the operation is similar to that of the count up mode, Figure 1.6(c). In the latter,

the only difference is that the update of the reference-register can be carried out when the counter reaches either its maximum or its minimum value, Figure 1.6(d).

1.2.2.2. Synchronization of the ADC and the DPWM for current-control

To avoid continuous sampling of the inductor current, a strategy adopted in digital current-controlled converters is to synchronize the operation of the ADC and the DPWM. Three forms of synchronization are possible allowing immediate access to the valley-current, peak-current or the average-current of the inductor. To sample the inductor valley-current, the DPWM is operated in count up mode and the ADC sampling is triggered when the DPWM counter is reset to zero, the inductor current is therefore sampled at the transistor turn on instant. Similarly, to acquire the inductor peak-current, the DPWM is operated in the sampling occurs when the DPWM counter is reset, the inductor current is then sampled at the transistor turn off instant. Finally, to acquire the average-current, the DPWM counter is reset, the inductor current is then sampled at the transistor turn off instant. Finally, to acquire the average-current, the DPWM counter is operated in count up-down mode and the sampling instants occur when the counter reaches its maximum value. This will result in the average value of the inductor current being obtained since the current is sampled in the middle of the transistor on-interval [31, 34, 35].

1.2.3. Review of digital control techniques for dc-dc converters

This section of the literature survey aims to review and classify the digital control methods found in the literature for the basic dc-dc converter topologies: buck, boost and buck-boost converters. In the following classification, the digital control strategies are grouped according to the methodology employed to regulate the inductor current, the output voltage or both. Five methodologies were identified: linear compensation methods or conventional methods, predictive/estimative methods, multi-sampling methods, mixed-signal methods and non-linear methods. This classification does not distinguish whether the implementation is carried out in a DSC or a FPGA.

1.2.3.1. Linear compensation or conventional compensation methods

These methods usually regulate the voltage or the current by means of a conventional linear digital compensator such as a PI or PID controller [36, 37]. In these

methodologies the feedback signals are the only information required to control the system. According to [38] in the late 1990s a great deal of work was focused on optimal design of linear compensators that performed as well as their analog counterparts. Enhanced versions of the linear controllers have been reported in the literature [39, 40] where extra poles and zeros are added to compensate the processing delay of the controller.

1.2.3.2. Predictive or estimative methods

Predictive current control methods in dc-dc converters have been proposed as a way to eliminate or reduce the phase-lag generated by the digital control-loop [34, 41-43]. The majority of these techniques employ extra information about the dc-dc converter usually in the form of a model which relies on circuit parameters or variables such as the value of the inductor or the input voltage. Errors in the model or parameter values will degrade the performance of the controller leading to poor robustness against parameter variations, for example due to ageing or temperature changes.

For instance, [34] presents a very interesting predictive current control technique where the switch duty-ratio of the next switching cycle is estimated employing the input voltage, output voltage and the inductor-current. The predictive technique is derived for three different control targets: inductor valley-current, average-current and peak-current. It was found that by choosing appropriately the modulation method for each control target, the predictive current law is the same for all the cases and the propagation of sub-harmonic oscillations is avoided. The correlation found was as follows: valley-current control under trailing-edge modulation; peak-current control under leading-edge modulation and average-current control under triangular carrier modulation. Furthermore, the control law is extended for use in buck, boost and buck-boost converters. The technique is validated experimentally employing current-mode control for a 100 W prototype of a power factor correction boost rectifier. The control strategy was implemented on a DSP and the outer voltage-loop was regulated using a conventional PI controller. One of the disadvantages of this control strategy is that the calculations for the prediction depend on the value of the

inductor employed in the converter, which is likely to be affected by temperature variations and manufacturing variations.

A subcategory of predictive controllers is the dead-beat controller. This technique aims to design a compensator that relocates all the poles of a closed-loop transfer function to the origin of the z-plane. The dead-beat response presents special transient characteristics such as: zero steady-state error, minimum rise-time, minimum settling-time and less than 2% overshoot/undershoot [44]. However, the design of deadbeat controllers is very difficult for dc-dc converters as an accurate model of the system is required [45].

Despite this, [45] derives a dead-beat current-control law for a single-phase buck converter with current-mode control. The approach stands out as it does not follow conventional discrete control design methodologies. Instead, the design is based on an analysis of the operation and the geometry of the converter waveforms. However there are two limitations to this method: the control law has to be modified to account for the computational delay, and the duty-ratio has to be kept constant for two switching-periods. The outer voltage control-loop of the controller uses a three-step-ahead predictive PI compensator. The proposed technique is validated experimentally employing a DSC and a 140 W buck converter prototype. The results show that the deadbeat implementation has very good performance, almost equal to that of analog peak-current-mode control.

1.2.3.3. Multi-sampling methodologies

These techniques have been proposed to improve the limited bandwidth of digital controllers [32, 33, 46, 47]. The principles are straightforward, while keeping the same switching frequency, the sampling frequency is increased by a factor of *N* and the control algorithm is executed *N* times during the switching period. A modelling study carried out in [46] identified two major drawbacks of this strategy: it injects high-frequency disturbances into the feedback-loop due to the voltage/current ripple and its interaction with the ADC and DPWM quantization may enhance limit-cycle oscillations. Also, due to the nature of the sampling requirements, these methods are best suited to FPGAs as several tasks can be achieved in parallel. An implementation in a DSC is possible but is likely to be limited by the maximum sampling frequency of the inbuilt ADC and the associated processing delay.

Moreover, the task is computationally intensive leaving the processor with little resources to execute other tasks.

1.2.3.4. Mixed-signal methodologies

These methods combine digital and analog elements to exploit the best characteristics of both technologies [48, 49]. In [48] a digitally calculated reference-current is converted to the analog domain by means of a one-bit Δ - Σ digital-to-analog converter, and then used in an analogue control loop. A dedicated integrated-circuit was built for the application. In [49], a continuous-time comparator inbuilt into the microcontroller is employed to compare the reference-current generated by a digital-to-analogue converter with the real-time inductor current. The slope compensation is combined digitally with the current-reference, which is generated by amplifying the voltage control loop error with a two-pole, two-zero compensator. The principal disadvantage of this approach with regard to multi-phase converters is that digital signal controllers do not usually have sufficient inbuilt comparators.

1.2.3.5. Non-linear methodologies

A large number of papers have been written examining the application of non-linear techniques to improve the performance of digitally controlled dc-dc converters. As this area is peripheral to the main topic of the research in the thesis, only a few interesting examples of these control methodologies are described [50, 51]. In these papers, a digital control algorithm is proposed where two different compensation approaches are employed for steady-state operation/small-load changes and for large-load changes. During steady-state operation the system is compensated with a typical current-mode architecture, whilst in the presence of large-load changes an algorithm based on the principle of capacitor charge balance is employed to obtain optimal dynamic response.

In [43] a time optimal control algorithm is presented, which attempts to drive the converter to steady-state operation in the minimum time possible by estimating the capacitor zero-current cross-over point. Using this information, the required duty-ratio to achieve optimal-time response is loaded from a look-up table that has been previously

calculated. These controllers behave in a linear manner when the converter is in steadystate and as a non-linear controller for transient conditions.

1.3. Control techniques for multi-phase dc-dc converters

Multi-phase or interleaved dc-dc converters have been widely adopted for applications such as power supplies for microprocessors and communications systems [52-55], power factor correction, [56-58], and power conversion systems for electric vehicles, [59-62], amongst others.

The parallel connected channels in an interleaved converter use the same topology, component-values and switching frequency, but the switching cycles are delayed with respect to each other to create an even distribution of the waveforms over a switching period. With two interleaved channels, the switching delay is half the switching period, with three it would be one third of the period and so on. This results in the ripple-current amplitude in the common input and output filter being reduced whilst its frequency is increased. Consequently the size of input and output filter capacitors can be reduced [21, 61-65].

Also, by the use of several phases the input power is distributed across several devices, reducing the current stress and thermal load. However, one of the challenges in the operation of interleaved converters is ensuring a balanced current distribution between the phases [58, 62, 63, 65]. This practical issue arises due to the non-identical characteristics of the parallel converters, such as passive and semiconductor component mismatches and parameter variations caused by temperature changes. Current mismatches between phases might affect overall system stability and limit the operation region [63]. Also, if one of the converter phases has an excessive load current, it will experience higher thermal stress, possibly reducing the system reliability [58]. More crucially, if a coupled inductor or inter-phase transformer is used to combine the interleaved channels, current mismatches can result in the saturation of the magnetic components causing a rapid rise in circuit current, potentially damaging the semiconductor devices.

For systems with a small number of phases, this issue can be mitigated by the use of an individual current control-loop for every converter comprising the system. The current control-loops share the same reference-current, thereby all the parallel converters will track the same target-reference [66]. However, when the number of phases is high, the number of current-loops that have to be implemented creates a significant increase in cost, weight and control complexity which can make the system impractical [61, 63, 65].

Simple and cost-effective techniques for ensuring balanced operation of voltagemode controlled interleaved converter for automotive applications were presented in [61] and [63]. These papers demonstrated that by operating the converter phases in discontinuous conduction mode the use of current-control loops for each phase may become unnecessary and therefore they can be eliminated. The control methodology employed was implemented in a FPGA and the practical evidence presented was obtained employing a sixteen-phase bidirectional converter in [61], and both sixteen- and thirty sixphase bidirectional converters in [63].

Other cost-effective techniques are introduced in references [62] and [65], which describe digital control methodologies to balance the phase currents of interleaved converters employing a single DC-link current sensor. This methodology, previously employed in three-phase PWM inverters, estimates the average value of the converter phase currents by strategically sampling the DC-link current in the middle of the transistor on-intervals and it is demonstrated that a similar performance to that of multi-loop current controller is achieved. In [62] the methodology is validated employing a 5 kW four-phase dc-dc converter for automotive battery-charging applications, whilst in [65] it is validated employing a 300 W dual-phase interleaved converter.

Other forms of control for interleaved converters include [55, 67, 68]. In [55] and [67] hysteretic voltage-mode control is employed in conjunction with a digital current balancing methodology for an *n*-phase interleaved converter that is used for a microprocessor voltage regulator. This methodology presents reduced undershoot and overshoot current caused by large magnitude transients, which is highly desirable in power supplies where tight regulation is required. Finally [68] proposes a non-linear sliding mode

control methodology to ensure current sharing in a dual-phase interleaved voltage regulator module for a microprocessor.

1.3.1. Control techniques for the dual interleaved boost converter

Relatively few papers have been published specifically relating to the control of interleaved boost converters. The studies presented in [16, 69] employ analog peakcurrent-mode control. The work in [16] describes a zero-voltage-switching, dual interleaved boost converter which is controlled using a commercially available dedicated integrated circuit, the UCC28220 of Texas Instruments. The work in [69] introduces a new analog control methodology, termed time-multiplexing current balance, which enhances the dynamic performance of the peak-current-mode control by reducing the effects of the right half-plane zero and achieves current balance at the same time. The control technique is implemented on a dedicated integrated circuit.

The work described in [37, 60] is concerned with the implementation of digital average-current control of the dual interleaved boost converter. Both techniques employ digital PI compensators and the control-loops are implemented in DSCs. The design of the PI compensators is carried out using an averaged small-signal model of the converter. In [37], the controller and converter are designed to operate in DCM. Conversely, in [60] the converter is designed to operate in CCM. Both publications show experimental results for a *20 kW* and a *30 kW* prototype of the converter respectively, detailing the dynamic performance of the system. However, they do not consider the effectiveness of the control methodology in balancing the phase currents.

In [70] another dual-loop digital average-current control methodology is presented, however, in contrast to the studies in [37, 60] an inner current control-loop is employed to ensure the equal distribution of the currents between the converter phases. The outer control-loop is a standard voltage regulation loop. However, the experimental results included to validate the control architecture correspond to a single point of operation and do not show the behaviour of the phase-currents during transitory events, such as load or reference changes.

[71, 72] both examine the use of predictive techniques for the control of the dual interleaved boost converter. In particular [72] shows the implementation of the predictive technique described in [34] employing valley-current control. The methodology is exhaustively verified using simulations; however, few experimental results which do not show the converter phase-currents are included, making it difficult to verify the fact that the phase-currents are balanced.

Non-linear control techniques for the dual interleaved boost converter are presented in [73-75]. In [73] a sliding-mode control methodology is implemented using a mixed-signal architecture. Reference [74] presents a current-mode controller based on hysteresis which is implemented on a FPGA.

In reference [76] an interesting multisampling control methodology implemented in a FPGA is presented. The technique stands out as the dynamic performance of the system is close to that achieved with peak current-mode control whilst also maintaining a good balance of the currents in the converter phases. In this technique, both the output voltage and the phase currents are sampled several times over a switching period. To eliminate the injection of high-frequency harmonics in the feedback loop due to the output voltage/phase current ripple discussed in [46], a moving-average filter is employed. In addition, to achieve the current regulation, the differential and the common mode currents of the converter are controlled using PI compensators.

1.4. Small-signal modelling of digitally controlled dc-dc converters and compensator design

Modelling the dynamics of all power electronic circuits including dc-dc converters is challenging due to the inherent high frequency switching action, consequently many different techniques have been proposed as a basis for control analysis and design. Amongst others, the most important modelling approaches employed in power electronics are the state-space averaging method and the sampled-data method.

The state-space averaging methodology was unified by Middlebrook and Cúk in 1977 [77]. This methodology allows derivation of an approximate, continuous, linear model of the converter by averaging the state-space equations of the different topological stages

of a circuit over a switching period and then linearizing the equations by considering small changes in the variables around a steady-state operating point. Small-signal models are often preferred for the design of converter control systems since their linear nature allows the full range of linear system techniques to be used. One of the major drawbacks of the state-space averaging approach is that it is not possible to model directly systems that employ a control variable which is different to the duty-ratio [25]. A notable extension to this modelling methodology was introduced by Ridley in [78] which enables the incorporation of the dynamics of peak-current-mode control.

So far, the most accurate technique to model dc-dc converters is the sampled-data modelling methodology [79] which is sometimes also referred to as the exact-modelling approach. This approach derives a model of the converter by the use of small-signal linearized difference equations, which describe the evolution of the state-variables at specific time-instants. However, in order to solve the difference equations, it is necessary to make use of terms with matrix exponentials, requiring a complex mathematical solution. This is perhaps, the major drawback of the methodology. Another significant drawback is that the procedure produces the converter model in closed-loop form. Therefore the mathematical procedure has to be carried out for each specific control approach.

The circuit averaging technique is a simple and intuitive approach which predates the state-space averaging method [22, 80]. There was renewed interest in the method after the development of the state-space averaging technique largely through the work of Vórperian in the early 90's [81, 82]. In his work, Vórperian proposes replacing the transistor and diode of the basic dc-dc converter topologies, by a general "averaged PWM-switch model". In this way he set out a more systematic approach for the use of circuit averaging. Another author who has also made significant contributions to models employing the circuit averaging method is Kazimierczuk [23, 83].

The detailed application of the circuit averaging method to a boost converter is presented in more detail in Chapter 2, Section 2.3.1, as it is the basis of the model developed in this work for the dual-interleaved boost converter with inter-phase transformer.

1.4.1. Modelling of digital average-current-controlled dc-dc converters

Figure 1.7 depicts the functional block diagram of a dc-dc converter with digital average-current control implemented in a DSC. The key elements are: the current transducers and their associated measurement delay, the analog-to-digital converter represented by an ideal sampler, the compensation process with its associated computational delay, the uniformly-sampled DPWM model, and the small-signal transfer function of the converter that represents the dynamics of the inductor-current response to perturbations in the duty-ratio. In this diagram it is assumed that the DPWM and the ADC are synchronized, allowing the average-current of the converter to be sampled in every cycle as described in Section 1.2.2.2. On this basis averaged converter models can be used directly to analyse the controller operation and design. Moreover, the quantization non-linearity of the ADC is often disregarded in the modelling procedure as it increases the complexity of the system, therefore the ADC is represented using an ideal sampler [31, 84, 85].





1.4.1.1. DPWM Model

Since the duty-ratio of the output waveform for a DPWM generator remains fixed across the switching cycle, the DPWM signal is commonly modelled as an extrapolating device such as the Zero-Order Hold, ZOH. The impulse response of a ZOH over a sampling period *T* can be represented mathematically as the sum of two unit-step functions [44], $u_s(t)$, as follows:

$$g_{h0}(t) = u_s(t) - u_s(t - T) \tag{1.1}$$

By taking the Laplace transform of Equation (1.1), the transfer function of the ZOH can be found:

$$G_{h0}(s) = \frac{1 - e^{-sT}}{s}$$
(1.2)

Equation (1.2) is usually found connected in cascade with the converter transfer functions to represent the effects of the modulator. This model has been widely employed by different authors to model the behaviour of a DPWM in power converters and is also adopted in this thesis. However, more detailed small-signal models of this element have been proposed by several authors. Some of the earliest derivations of the small-signal model of the UPWM with leading-edge operation were carried out in [86, 87]. More recently (2004), a Laplace-domain small-signal model of this element was developed in [35] for the three most common configurations: trailing-edge, leading-edge and triangular carrier modulation. Table 1.1 summarizes the expressions for this model. Two years later, this work was extended when the z-domain small-signal expressions of the DPWM were determined [84].

Table 1.1. Laplace-domain and frequency-domain models for uniformly-sampled pulse-widthmodulations as described in [37] by Van de Sype et al.

Modulation Strategy	DPWM(s)	DPWM(jω)
Leading-edge	e^{-sDT}	1∠(− <i>j</i> @DT)
Trailing-edge	$e^{-s(1-D)T}$	1∠(− <i>jα</i> (1− <i>D</i>)7)
Triangular carrier Symmetric on-time	$\frac{1}{2} \left(e^{-s\frac{1}{2}(1-D)\tau} + e^{-s\frac{1}{2}(1+D)\tau} \right)$	$\cos\left(\frac{\omega DT}{2}\right) \angle \left(-\frac{\omega T}{2}\right)$
Triangular carrier Symmetric off-time	$\frac{1}{2}\left(e^{-sDT}+e^{-s(2-D)T}\right)$	$\cos\left(\frac{\omega(1-D)T}{2}\right) \angle \left(-\frac{\omega T}{2}\right)$
Triangular carrier Double update-mode	$\frac{1}{2}\left(e^{-s(1-D)T}+e^{-sDT}\right)$	$\cos\left(\omega(D-\frac{1}{2})T\right) \ge \left(-\frac{\omega T}{2}\right)$
*D. O.	vieseent duty quele Ty Compliance	aariad

*D: Quiescent duty-cycle, T: Sampling period.

Furthermore, other authors, [85], have presented closed-form discrete models for the buck, boost, and buck-boost converters, where the effects of the DPWM are incorporated in the closed-loop model for leading- and trailing-edge modulation and zdomain expressions are given for each case.

Buso and Mattavelli in [31] introduced an approximation of the DPWM small-signal model proposed in [37] when the triangular carrier modulation with symmetric on-time is employed. The authors conclude that for this particular modulation case and when a smallsignal averaged model is employed to describe the converter dynamics, it is sufficient to approximate the transfer-function of the DPWM by that of a ZOH.

1.4.1.2. Quantization and limit cycle oscillations

Limit cycle oscillations, LCOs, caused by the quantization effects in digital controllers are due to the limited resolution of the ADC and DPWM [44, 88]. However for the sake of simplicity, these effects are not usually considered in the modelling of dc-dc converters. In [89-91] the presence of LCOs in dc-dc converters is studied and it is concluded that a high integral gain and that a coarse DPWM resolution contribute significantly to the presence of LCOs amongst other conditions. To improve the resolution of DPWM modules without the need to increase their clock frequency, techniques such as dithering and phase-locked loops have been proposed in [38, 90].

1.5. Compensator design for digital controlled dc-dc converters

Digital redesign and direct digital design [44, 88, 92] are the two main approaches that have been widely employed for the design of linear compensators in dc-dc converters. More recently in [93] an intuitive digital design approach was proposed which allows the use of analogue control tools to design digital compensators for dc-dc converters. These methodologies will be reviewed in the following sections.

1.5.1. Digital redesign

This is an indirect design technique where a digital equivalent of an analog compensator is used to regulate the system. In the design process the analog compensator

Introduction and literature review

Chapter 1

is first devised in the continuous-domain by means of a Laplace-domain model of the plant. The resulting analog compensator is then transformed to its equivalent in the digital domain, or the z-domain, by means of an appropriate discretization method. The principal advantage of this approach is that it provides the designer with the intuitiveness of analog control theory where several techniques such as root-locus, Bode diagrams and Nyquist plots amongst others, can be employed to choose a compensator to satisfy the desired system performance. A variety of discretization methods is available to approximate the analog controller. The most common are: the forward Euler method or forward rectangular rule, the backward Euler method or backward rectangular rule, the step-invariant or zero-order-hold method, the bilinear transformation or Tustin's method, and the pole-zero matching method [92].

However, this design approach has a number of limitations. First, it requires an accurate model of the complete system including the effects of the digital-to-analog and the analog-to-digital converters. This can be achieved by the use of zero-order-hold transfer functions and by the incorporation of transport delays. However the use of transport delays introduces a non-linearity. To overcome this issue, these delays are typically represented by the Padé approximation of the transport delay [94].

Second, none of the discretization methods preserves all the characteristics of the original analog controller in the digital domain. The backward Euler and forward Euler transformations are the simplest but they only provide an exact match in the time domain. Any information in the frequency domain is distorted due to aliasing effects [92]. The bilinear transformation preserves the magnitude and phase response of the system in the frequency-domain up to 1/10 of the sampling frequency, nonetheless, the time domain response is not accurately represented [92]. The step-invariant transformation maintains the response of the system to step or staircase inputs only. Finally, the pole-zero match transformation retains the location of the poles and zeros of the compensator, but its implementation is complicated and is affected by aliasing if the frequency-location of the zeros of the compensator is greater than half of the sampling frequency [92].

However, in spite of these difficulties, the digital redesign methodology has been widely adopted and studied for dc-dc converters. In [95, 96] studies were carried out to

determine the best discretization method to use in the design of a dc-dc converter controller. An analog PI compensator was discretized using the techniques described above and it was concluded that the backward Euler method offered the best performance. Some examples of the use of digital redesign for dc-dc converters can be found in [37, 39, 41, 45, 49, 95, 96].

1.5.2. Direct digital design

In the direct digital design methodology, the controller is directly devised in the zdomain by means of a discrete-time model of the plant. When the model of the plant is already in the z-domain the design procedure can be carried out directly. However, if the model of the plant is in the Laplace-domain, it is necessary to find an approximate z-domain model. Provided that the Laplace-model of the plant is connected in cascade with a digitalto-analog converter and that the digital-to-analog converter can be modelled as a zeroorder-hold, then the approximated discrete-time model of the plant and the DAC can be derived by the use of the zero-order-hold transform [44, 88, 92]. This is usually the case in power electronics and other application areas.

This design technique is advantageous as it allows the modelling of the different delays involved in the digital control process without the need for linear approximations. Moreover the digital compensator designed can be directly implemented in a microcontroller without any loss of accuracy. This design approach also enables the use of exclusive digital control tools, such as the design of dead-beat compensators. Perhaps, the most serious disadvantage of this approach is that it lacks the intuitiveness of analog control design procedures which are wide spread and well-known. For instance, the use of Bode plots, which are commonly used for dc-dc converters is not straightforward since the variable *z* is related to *j* ω by means of $e^{j\omega}$ [44]. In power converter controller design, although it has been demonstrated that the use of direct digital design leads to better performance than digital redesign [38, 95], direct digital design has not been widely adopted and only a few publications have been found employing this approach.

1.5.3. Other design methodologies

In 2007 [93], Al-Atrash proposed a new design approach where direct-digital design is carried out without the need of a discrete-time model of the converter. This is accomplished by plotting the combined frequency response of all the elements comprising the open-loop transfer function. The frequency response of the digital-elements, such as time-delays and zero-order-holds, is obtained using the direct relation $z^{-1} = e^{-sT} = e^{-j2\pi fT}$, which in contrast to the typical discretization methods such as backward Euler, or the bilinear transformation is an exact relationship. Therefore the frequency response is valid for any range of frequencies. Moreover, in this work, Al-Atrash defines a set of discrete zeroes, poles and complex conjugated zeroes and poles, which have a frequency response in the digital domain that is similar to that of their analog counterparts. With these two tools, the design of a digital compensator can be carried out in the analog domain without any loss of accuracy. Perhaps, the only disadvantage of this methodology is that a computational tool is necessary to obtain the frequency response of the mixed-signal system.

1.6. Review of modelling of multi-phase dc-dc converters

By far, the most widely used technique for analysing and designing the controllers for interleaved converters is the state-space averaging method [60, 70, 97-101]. This technique is mostly applied to model digitally controlled dc-dc converters with averagecurrent control, which is used to justify the use of an averaged model. Other methodologies reported, which do not necessarily use digital control include sampled-data modelling [102], signal flow graphs [103], multi-frequency small-signal modelling [54] and non-linear approaches [104, 105].

In the research presented in [97] and [98] a state-space averaged model is developed for the steady-state analysis of a dual interleaved boost converter in the time domain. In [97] only CCM is considered, whilst in [98] both CCM and DCM are considered. However, small-signal transfer functions are not derived in these papers.

The research in [70] and [99] describes the small-signal averaged state-space model of a dual interleaved boost converter with inter-phase transformer under CCM. In [70] the model equations are presented by considering the small-signal perturbations of the dutyratios for each of the converter phases independently. As a consequence, a cross-coupling transfer-function between the phases arises in the system. However, this new information is not exploited in the design of the controller that was proposed and the procedure to derive the model equations is not described. In [99] the full procedure to obtain the smallsignal model is illustrated for the unidirectional and the bidirectional operation of the converter. However, the small-signal perturbation of the duty-ratios of the converter phases are considered to be identical limiting the accuracy of the model.

In references [60] and [100] the small-signal averaged PWM switch techniques based on the approaches proposed by Kazimierczuk et al. in [83] and Vórperian et al. in [81] respectively, are employed to model multi-phase dc-dc boost converters. However, the models obtained assume identical duty-ratios in the converter phases even for small-signal perturbations.

The research in [52] and [69] presents small-signal averaged models of peak current mode controlled interleaved converters used as voltage regulators for microprocessors, which are derived employing the enhanced averaged model proposed by Ridely in [78]. In [52] a dual-phase interleaved buck converter is considered with and without coupled inductors. The model developed takes into account the small-signal perturbation of the duty-ratio of the converter phases independently and the performance of the converter is compared using the coupled and the non-coupled versions, revealing that the coupled inductors, or inter-phase transformer, improve the bandwidth of the system. In [69] a dual-phase interleaved buck converter is employed where a control technique termed time-multiplexing current balance control is proposed. Employing the model of the converter, which includes the small-signal perturbation of the phase duty-ratios separately, it is demonstrated that the control technique improves the system bandwidth by relocating the right-half plane zero of the control-to-input-current transfer function to higher frequencies. However, the models presented in both [52] and [69] do not account for the interleaved operation of the converter phases.

References [106] and [101] report the development of state-space averaged models for two different multiphase dc-dc converters. In [106], a generalized algebraic methodology is presented to obtain the small-signal state-space average model of a multiphase converter composed of 2N boost cells with coupled inductors. Two keyassumptions are made for the development of this model: the parameters of the boost cells are identical and the small-signal perturbations in the duty-ratios are the same for all the converters. The transfer-functions obtained from this model are second order no matter how many phases are employed in the converter. Moreover, the performance of the control-to-output, line-to-output and output impedance transfer functions are compared for different inductor couplings by simulation in the frequency-domain and in the timedomain experimentally. It is concluded that the use of directly-coupled inductors offers low overshoot and insensitivity to input-voltage variations; and that the use of inversely coupled inductors provides fast response and insensitivity to load variations. However, this modelling approach disregards the interleaved operation of the boost cells, and does not include any feedback control.

In [101] the modelling of a multiphase interleaved buck converter is presented. Similar to the work in [106], a generalized methodology is proposed to model *n* interleaved buck converter stages. However, in contrast with [106], this work takes into account the small-signal perturbation of each of the converter duty-ratios and several stages of coupled inductors are considered. The algebraic model is then applied to an eight-phase buck converter, and the resultant transfer functions obtained are employed to design a PI compensator using the digital-redesign approach. The performance of the design is verified experimentally in the frequency and the time domain. Also, ripple and efficiency analyses are presented. However, the modelling procedure of the current-feedback loop does not account for the effects of zero-order-holds and the interleaved interaction of the converter phases.

The sampled-data model of a dual interleaved boost converter with inter-phase transformer under peak current-mode control was described in [15, 102]. This model is capable of predicting the stability limits of the converter with a great deal of accuracy and is based on the technique introduced in [79]. The methodology proposed exploits the

symmetry of the converter waveforms to reduce the complexity of the algebraic procedure, nevertheless, the model obtained is still very complex and difficult to use for control design. This study also presents a detailed analysis of how the pole and zero locations of the control-to-output transfer functions of the converter are affected by steady-state operating conditions and the circuit parameters such as the ratio between input inductance and IPT self-inductance.

A notable study reported in [104] demonstrated by means of nonlinear analyses the inability of state-space averaging to distinguish between converters operating in phase from those operating using interleaving and the ineffectiveness of the technique to account for fast scale dynamics.

Currently, most of the reported research using state-space averaged models, apart from [52, 69, 70, 101], assumes that both the quiescent values of the duty-ratios and their small-signal perturbations are identical for each of the converter phases. Also, none of the modelling techniques presented using averaging techniques account for the interaction between the interleaved converters. This shortcoming is addressed in this thesis through the development of a method that enables an averaged model to account for the interleaved operation and interaction of a two-phase interleaved boost converter. This is carried out by means of multi-rate sampled-data systems theory and is used to establish new stability limits for the converter. To the best of the author's knowledge this technique has not previously been applied to dc-dc converters.

1.7. Summary of literature review

Digital control of the basic dc-dc converter topologies has been widely researched and reported in the literature, especially for low-power applications with operating frequencies of the order of hundreds of *kHz* and in some cases up to *1 MHz*. In these applications a key objective is achieving a fast-dynamic performance. As a consequence, improved linear compensators, predictive techniques and multi-sampling methodologies have been proposed to overcome the processing delay that can have a detrimental effect on the current-control-loop bandwidth. The use of DSCs, FPGAs and dedicated integratedcircuits has been reported for several applications favouring the use of DSCs as they already

contain all the necessary modules to control a power converter allowing rapid application development. However, most of the research work introducing new control architectures and techniques have been reported employing FPGAs.

With respect to multi-phase converters, a key priority is control strategies that ensure equal current-sharing between the converter cells whilst also providing good dynamic performance. These converters are widely applied in high-power applications as they allow a reduction in the size of passive components due to the effective increase of input and output ripple frequencies and reduced-stress on semiconductor devices. When a high number of converter cells are employed, cost-effective control solutions which reduce or eliminate the need of a single current-control loop per phase are desired. For converters with a small number of phases, such as the dual interleaved boost converter, the use of a single current-control loop per phase has been widely adopted as it ensures the equal distribution of the current between the converter cells. Analog control and digital average-current control implementations in DSCs and FPGAs have been reported in the literature.

With regard to the modelling and analysis of multi-phase converters, averaging techniques tend to be preferred due to their simplicity. Other techniques such as sampleddata modelling and non-linear approaches have also been employed to explore non-linear and switching related phenomena and to demonstrate the limitations of the averaging techniques.

Since much of the development of digital control techniques has been carried out for single-phase dc-dc converters, the compensator design for multi-phase converter tends to be approached in the same way. Also, there is a lack of a well-defined methodology to model the elements of a digital power electronic control-loop. Moreover, digital redesign has been widely adopted due to its strong links with analog control methodologies. Nonetheless, this technique has been demonstrated to be less accurate than direct digital design. Overall, digital modelling of power converters is a field still in development, and direct digital design has not been widely adopted since digital redesign is more similar to the familiar analogue design methodologies. In this thesis a direct digital design approach is used to model the dual interleaved boost converter in closed-loop and by the use of multi-rate data-sampled theory the circuit averaging techniques are enhanced to predict instabilities caused by the interleaved operation of multi-phase converters.

1.8. Outline of this thesis

In the first part of Chapter 2, the methodology employed to obtain a small-signal, averaged model of the dual-interleaved boost converter with inter-phase transformer is presented. The model is validated employing simulations from SABER. In the second part of Chapter 2, the operation of the converter employing digital valley-current, average-current and peak-current control is examined and the advantages and disadvantages of the techniques are identified.

In Chapter 3, the design and construction of a digital control platform for the dualinterleaved boost converter and the implementation of the digital average-current control are presented together with the development of a detailed closed-loop SABER model. The correct operation of the control platform and the SABER simulation model are verified using steady-state and transient experimental results from a 25 kW prototype.

In Chapter 4, a modelling methodology based on the sampler decomposition technique is introduced which allows to obtain the small-signal, closed-loop transfer functions of the converter accounting for the interleaved sampling and operation of the converter phases. It is demonstrated that the model is able to predict instabilities caused by this feature of the interleaved converter. The results of the proposed model are validated using time-domain and frequency-domain simulation results from the SABER models developed in Chapter 3.

In the first part of Chapter 5, a sensitivity analysis of the pole trajectories of the model developed in Chapter 4 is undertaken to identify how the main parameter variations of the converter affect the absolute stability of the system. In the second part of this Chapter the parameters of a PI compensator are selected to ensure maximum response speed and minimum overshoot to step-change transients based on the theories and models

developed in this thesis. The final design is validated using simulation and experimental results from the converter prototype.

Finally Chapter 6, presents the conclusions of this dissertation and its contributions to the field of research. The chapter concludes by suggesting potential areas where future research work can be carried out.

1.9. References

- [1] (17th March, 2015). *UK Emissions by Sector Road Transport*. Available: <u>http://www.theccc.org.uk/charts-data/ukemissions-by-sector/transport/</u>
- [2] (17th March, 2015). *The Climate Change Act and UK regulations*. Available: <u>http://www.theccc.org.uk/tackling-climate-change/the-legal-landscape/global-action-on-climate-change/</u>
- [3] Element-Energy-Limited, "EVs in the UK & Republic of Ireland: Greenhouse gas emission reductions & infrastructure needs. Final report for the WWF UK," August, 2010.
- [4] WWF, "Plugged in: The end of the oil age Summary report," March, 2008.
- [5] C. C. Chan, "The state of the art of electric and hybrid vehicles," *Proceedings of the IEEE*, vol. 90, pp. 247-275, 2002.
- [6] A. Emadi, S. S. Williamson, and A. Khaligh, "Power electronics intensive solutions for advanced electric, hybrid electric, and fuel cell vehicular power systems," *Power Electronics, IEEE Transactions on,* vol. 21, pp. 567-577, 2006.
- [7] C. C. Chan and K. T. Chau, "An overview of power electronics in electric vehicles," *Industrial Electronics, IEEE Transactions on,* vol. 44, pp. 3-13, 1997.
- [8] A. Khaligh and L. Zhihao, "Battery, Ultracapacitor, Fuel Cell, and Hybrid Energy Storage Systems for Electric, Hybrid Electric, Fuel Cell, and Plug-In Hybrid Electric Vehicles: State of the Art," *Vehicular Technology, IEEE Transactions on*, vol. 59, pp. 2806-2814, 2010.
- [9] S. M. Naylor, V. Pickert, and D. J. Atkinson, "Fuel Cell Drive Train Systems -- Driving Cycle Evaluation of Potential Topologies," in *Vehicle Power and Propulsion Conference. VPPC '06. IEEE*, 2006, pp. 1-6.
- [10] M. A. Sakka, J. Van Mierlo, and H. Gualous. (19th March, 2015). DC/DC Converters for Electric Vehicles, Electric Vehicles - Modelling and Simulations. Available: <u>http://www.intechopen.com/books/electric-vehicles-modelling-andsimulations/dc-dcconverters-for-electric-vehicles</u>
- [11] C. C. Chan and K. T. Chau, *Modern Electric Vehicle Technology*: Oxford University Press, 2001.
- [12] Z. Hui, L. M. Tolbert, and B. Ozpineci, "Impact of SiC Devices on Hybrid Electric and Plug-In Hybrid Electric Vehicles," *Industry Applications, IEEE Transactions on*, vol. 47, pp. 912-921, 2011.
- [13] B. Wrzecionko, J. Biela, and J. W. Kolar, "SiC power semiconductors in HEVs: Influence of junction temperature on power density, chip utilization and efficiency," in *Industrial Electronics. IECON '09. 35th Annual Conference of IEEE*, 2009, pp. 3834-3841.
- [14] G. Calderon-Lopez and A. J. Forsyth, "High power density DC-DC converter with SiC MOSFETs for electric vehicles," in *Power Electronics, Machines and Drives. PEMD* '14, 7th IET International Conference on, 2014, pp. 1-6.
- [15] G. Calderon-Lopez, "Design and control of the dual-interleaved boost converter with interphase transformer," Ph.D. dissertation, School of Electrical and Electronic Enginnering, The University of Manchetser, 2009.

Chapter	1
---------	---

[16]	G. Calderon-Lopez and A. J. Forsyth, "High-Power Dual-Interleaved ZVS Boost
	Converter with Interphase Transformer for Electric Vehicles," in Applied Power
	Electronics Conference and Exposition. APEC '09. Twenty-Fourth Annual IEEE, 2009,
	pp. 1078-1083.

- [17] M. Hirakawa, Y. Watanabe, M. Nagano, K. Andoh, S. Nakatomi, S. Hashino, *et al.*, "High power DC/DC converter using extreme close-coupled inductors aimed for electric vehicles," in *Power Electronics Conference. IPEC '10, International*, 2010, pp. 2941-2948.
- [18] A. Fratta, P. Guglielmi, F. Villata, and A. Vagati, "Efficiency and cost-effectiveness of AC drives for electric vehicles improved by a novel, boost DC-DC conversion structure," in *Power Electronics in Transportation*, 1998, pp. 11-19.
- [19] G. Calderon-Lopez, A. J. Forsyth, and D. R. Nuttall, "Design and Performance Evaluation of a 10-kW Interleaved Boost Converter for a Fuel Cell Electric Vehicle," in *Power Electronics and Motion Control Conference. IPEMC '06. CES/IEEE 5th International*, 2006, pp. 1-5.
- [20] D. Maksimovic, R. Zane, and R. Erickson, "Impact of digital control in power electronics," in *Power Semiconductor Devices and ICs. ISPSD '04. Proceedings of the 16th International Symposium on*, 2004, pp. 13-22.
- [21] S. Ang and A. Oliva, *Power-switching converters*. USA: CRC Press, 2005.
- [22] R. W. Erickson and D. Maksimovic, *Fundamentals of Power Electronics*. USA: Kluwer Academic Publishers, 2001.
- [23] M. K. Kazimierczuk, *Pulse-width modulated dc-dc power converters*. USA: Wiley, 2008.
- [24] R. Mammano, "Switching power supply topology, voltage mode vs. current mode," *Unitrode, Design Note,* 1999.
- [25] B. A. Hsu S.P., Rensink L., Middlebrook R.D., "Modelling and analysis of switching dcto-dc converters in constant-frequency current-programmed mode," in *Power Electronics Specialists Conference. PESC '79. IEEE 10th Annual*, 1979, pp. 284-301.
- [26] L. Dixon, "Average current mode control of switching power supplies," *Unitrode Application Note,* pp. 3-356-3-369, 1999.
- [27] S. Microelectronics. (2015, 20th March). *Fixed frequency average current mode power factor corrector L4981*. Available: <u>http://www.st.com/</u>
- [28] Y. Yingyi, F. C. Lee, P. Mattavelli, and L. Pei-Hsin, "I² Average Current Mode Control for Switching Converters," *Power Electronics, IEEE Transactions on*, vol. 29, pp. 2027-2036, 2014.
- [29] Texas-Instruments. (2015, 20th March). *Real-time control Microcontrollers (MCU)*. Available: <u>http://www.ti.com/lsds/ti/microcontrollers 16-bit 32-bit/c2000_performance/real-time_control/overview.page</u>
- [30] Microchip. (2015, 20th March). *dsPIC33FJ64GS610*. Available: <u>http://www.microchip.com/wwwproducts/Devices.aspx?dDocName=en547034</u>
- [31] S. Buso and P. Mattavavelli, *Digital Control in Power Electronics*. USA: Morgan & Claypool, 2006.
- [32] A. Syed, E. Ahmed, D. Maksimovic, and E. Alarcon, "Digital pulse width modulator architectures," in *Power Electronics Specialists Conference. PESC '04. IEEE 35th Annual*, 2004, pp. 4689-4695 Vol.6.

Cha	ntor	1
Cild	plei	Τ.

[33]	L. Corradini and P. Mattavelli, "Modeling of Multisampled Pulse Width Modulators
	for Digitally Controlled DC-DC Converters," Power Electronics, IEEE Transactions on,
	vol. 23, pp. 1839-1847, 2008.

- [34] C. Jingquan, A. Prodic, R. W. Erickson, and D. Maksimovic, "Predictive digital current programmed control," *Power Electronics, IEEE Transactions on*, vol. 18, pp. 411-419, 2003.
- [35] D. M. Van de Sype, K. De Gusseme, A. P. Van den Bossche, and J. A. Melkebeek, "Small-signal Laplace-domain analysis of uniformly-sampled pulse-width modulators," in *Power Electronics Specialists Conference. PESC 04. IEEE 35th Annual*, 2004, pp. 4292-4298 Vol.6.
- [36] S. Chattopadhyay and S. Das, "A digital current mode control technique for DC-DC converters," in *Applied Power Electronics Conference and Exposition. APEC '05. Twentieth Annual IEEE*, 2005, pp. 885-891 Vol. 2.
- [37] H. Xudong, W. Xiaoyan, T. Nergaard, L. Jih-Sheng, X. Xingyi, and L. Zhu, "Parasitic ringing and design issues of digitally controlled high power interleaved boost converters," *Power Electronics, IEEE Transactions on,* vol. 19, pp. 1341-1352, 2004.
- [38] L. Yan-Fei, E. Meyer, and L. Xiaodong, "Recent Developments in Digital Control Strategies for DC/DC Switching Power Converters," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 2567-2577, 2009.
- [39] G. Liping, J. Y. Hung, and R. M. Nelms, "PID controller modifications to improve steady-state performance of digital controllers for buck and boost converters," in *Applied Power Electronics Conference and Exposition. APEC '02. Seventeenth Annual IEEE*, 2002, pp. 381-388 vol.1.
- [40] D. Raviv and E. W. Djaja, "Technique for enhancing the performance of discretized controllers," *Control Systems, IEEE,* vol. 19, pp. 52-57, 1999.
- [41] S. Bibian and J. Hua, "Time delay compensation of digital control for DC switchmode power supplies using prediction techniques," *Power Electronics, IEEE Transactions on,* vol. 15, pp. 835-842, 2000.
- [42] Z. Guohua and X. Jianping, "Digital Average Current Controlled Switching DC-DC Converters With Single-Edge Modulation," *Power Electronics, IEEE Transactions on*, vol. 25, pp. 786-793, 2010.
- [43] Z. Guohua, X. Jianping, and J. Yanyan, "Elimination of Subharmonic Oscillation of Digital-Average-Current-Controlled Switching DC-DC Converters," *Industrial Electronics, IEEE Transactions on,* vol. 57, pp. 2904-2907, 2010.
- [44] B. C. Kuo, *Digital control systems*, 2nd ed. USA: Oxford University Press, Inc., 1992.
- [45] S. Bibian and J. Hua, "High performance predictive dead-beat digital controller for DC power supplies," in *Applied Power Electronics Conference and Exposition. APEC* '01. Sixteenth Annual IEEE, 2001, pp. 67-73 vol.1.
- [46] L. Corradini and P. Mattavelli, "Analysis of Multiple Sampling Technique for Digitally Controlled dc-dc Converters," in *Power Electronics Specialists Conference. PESC '06.* 37th IEEE, 2006, pp. 1-6.
- [47] C. Ye-Then and L. Yen-Shin, "Effect of sampling frequency of A/D converter on controller stability and bandwidth of digital-controlled power converter," in *Power Electronics. ICPE '07. 7th Internatonal Conference on*, 2007, pp. 625-629.

Chapter	1
---------	---

[48]	O. Trescases, Z. Lukic, N. Wai Tung, and A. Prodic, "A low-power mixed-signal
	current-mode DC-DC converter using a one-bit ΔΣ DAC," in Applied
	Power Electronics Conference and Exposition. APEC '06. Twenty-First Annual IEEE,
	2006, р. 5 рр.

- [49] M. Hallworth and S. A. Shirsavar, "Microcontroller-Based Peak Current Mode Control Using Digital Slope Compensation," *Power Electronics, IEEE Transactions on*, vol. 27, pp. 3340-3351, 2012.
- [50] F. Guang, E. Meyer, and L. Yan-Fei, "A New Digital Control Algorithm to Achieve Optimal Dynamic Performance in DC-to-DC Converters," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 1489-1498, 2007.
- [51] V. Yousefzadeh, A. Babazadeh, B. Ramachandran, E. Alarcon, L. Pao, and D. Maksimovic, "Proximate Time-Optimal Digital Control for Synchronous Buck DC-DC Converters," *Power Electronics, IEEE Transactions on*, vol. 23, pp. 2018-2026, 2008.
- [52] X. Ming, J. Zhou, K. Yao, and F. C. Lee, "Small Signal Modeling of a High Bandwidth Voltage Regulator Using Coupled Inductors," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 399-406, 2007.
- [53] Z. Xunwei, W. Pit-Leong, X. Peng, F. C. Lee, and A. Q. Huang, "Investigation of candidate VRM topologies for future microprocessors," *Power Electronics, IEEE Transactions on*, vol. 15, pp. 1172-1182, 2000.
- [54] Q. Yang, X. Ming, K. Yao, S. Juanjuan, and F. C. Lee, "Multifrequency Small-Signal Model for Buck and Multiphase Buck Converters," *Power Electronics, IEEE Transactions on*, vol. 21, pp. 1185-1192, 2006.
- [55] J. Abu-Qahouq, M. Hong, and I. Batarseh, "Multiphase voltage-mode hysteretic controlled DC-DC converter with novel current sharing," *Power Electronics, IEEE Transactions on*, vol. 19, pp. 1397-1407, 2004.
- [56] L. Po-Wa, Y. S. Lee, D. K. W. Cheng, and L. Xiu-Cheng, "Steady-state analysis of an interleaved boost converter with coupled inductors," *Industrial Electronics, IEEE Transactions on*, vol. 47, pp. 787-795, 2000.
- [57] B. A. Miwa, D. M. Otten, and M. F. Schlecht, "High efficiency power factor correction using interleaving techniques," in *Applied Power Electronics Conference and Exposition. APEC '92. Conference Proceedings of the IEEE Seventh Annual*, 1992, pp. 557-568.
- [58] N. Genc and I. Iskender, "DSP-based current sharing of average current controlled two-cell interleaved boost power factor correction converter," *Power Electronics, IET*, vol. 4, pp. 1015-1022, 2011.
- [59] O. Garcia, P. Zumel, A. de Castro, and J. A. Cobos, "Automotive DC-DC bidirectional converter made with many interleaved buck stages," *Power Electronics, IEEE Transactions on*, vol. 21, pp. 578-586, 2006.
- [60] O. Hegazy, J. Van Mierlo, and P. Lataire, "Analysis, Modeling, and Implementation of a Multidevice Interleaved DC/DC Converter for Fuel Cell Hybrid Electric Vehicles," *Power Electronics, IEEE Transactions on,* vol. 27, pp. 4445-4458, 2012.
- [61] N. Liqin, D. J. Patterson, and J. L. Hudgins, "High Power Current Sensorless Bidirectional 16-Phase Interleaved DC-DC Converter for Hybrid Vehicle Application," *Power Electronics, IEEE Transactions on,* vol. 27, pp. 1141-1151, 2012.

	Cha	pter	1
--	-----	------	---

ounghoon, A. Koran, H. Miwa, B. York, and L. Jih-Sheng, "An Active Current
nstruction and Balancing Strategy With DC-Link Current Sensing for a Multi-
e Coupled-Inductor Converter," Power Electronics, IEEE Transactions on, vol.
p. 1697-1705, 2012.
D S P

- [63] O. Garcia, A. de Castro, P. Zumelis, and J. A. Cobos, "Digital-Control-Based Solution to the Effect of Nonidealities of the Inductors in Multiphase Converters," *Power Electronics, IEEE Transactions on,* vol. 22, pp. 2155-2163, 2007.
- [64] R. Giral, L. Martinez-Salamero, and S. Singer, "Interleaved converters operation based on CMC," *Power Electronics, IEEE Transactions on*, vol. 14, pp. 643-652, 1999.
- [65] K. Hongrae, M. Falahi, T. M. Jahns, and M. Degner, "Inductor Current Measurement and Regulation Using a Single DC Link Current Sensor for Interleaved DC-DC Converters," *Power Electronics, IEEE Transactions on*, vol. 26, pp. 1503-1510, 2011.
- [66] B. Sun, "Digital current balancing for an interleaved boost PFC," *Analog applications journal*, 2012.
- [67] J. A. Abu-Qahouq, N. Pongratananukul, I. Batarseh, and T. Kasparis, "Multiphase voltage-mode hysteretic controlled VRM with DSP control and novel current sharing," in *Applied Power Electronics Conference and Exposition, APEC. Seventeenth Annual IEEE*, 2002, pp. 663-669 vol.2.
- [68] J. Cabiles-Magsino, R. C. Guevara, and M. Escoto, "Implementation of sliding mode control for current sharing in fixed frequency voltage regulator modules," in *TENCON. IEEE Region 10 Conference*, 2012, pp. 1-6.
- [69] L. Yean-Kuo, S. Yi-Ping, H. Yu-Ping, L. Yu-Huei, C. Ke-Horng, and H. Wei-Chou, "Time-Multiplexing Current Balance Interleaved Current-Mode Boost DC-DC Converter for Alleviating the Effects of Right-half-plane Zero," *Power Electronics, IEEE Transactions on*, vol. 27, pp. 4098-4112, 2012.
- [70] K. Kono, Y. Nakamura, and M. Yamamoto, "Optimal design of digital control system for interleaved boost chopper circuit with the coupled inductor," in *Power Electronics and Drive Systems. PEDS '11. IEEE Ninth International Conference on*, 2011, pp. 995-998.
- [71] P. Karamanakos, G. Papafotiou, and S. N. Manias, "Model predictive control of the interleaved DC-DC boost converter," in *System Theory, Control, and Computing. ICSTCC '11. 15th International Conference on*, 2011, pp. 1-6.
- [72] C. S. Babu and M. Veerachary, "Predictive controller for interleaved boost converter," in *Industrial Electronics. ISIE '05. Proceedings of the IEEE International Symposium on*, 2005, pp. 577-581 vol. 2.
- [73] R. Giral, L. Martinez-Salamero, R. Leyva, and J. Maixe, "Sliding-mode control of interleaved boost converters," *Circuits and Systems I: Fundamental Theory and Applications, IEEE Transactions on,* vol. 47, pp. 1330-1339, 2000.
- [74] J. S. Batchvarov, J. L. Duarte, and M. A. M. Hendrix, "Interleaved converters based on hysteresis current control," in *Power Electronics Specialists Conference. PESC 00. IEEE 31st Annual*, 2000, pp. 655-661 vol.2.
- [75] L. J. Alvarez, A. Nogueiras, A. Lago, J. Doval, J. Marcos, L. E. Eguizabal, et al., "Design, Analysis and Modeling of an Optimized Fuzzy Control Algorithm for Synchronous Multiphase DC-DC Converters in Automotive Applications," in *Power Electronics* Specialists Conference. PESC '06. 37th IEEE, 2006, pp. 1-6.

[76]	A. Fratta, G. Griffero, P. Guglielmi, S. Nieddu, and G. M. Pellegrino, "Application new FPGA-based top-performance digital control techniques to PWM per converters," in <i>Industrial Electronics Society. IECON '04. 30th Annual Conference</i> <i>IECE</i> 2004, pp. 810–815 Vol. 1
[77]	R. D. Middlebrook and S. ĆUk, "A general unified approach to modelling switch converter power stages," <i>International Journal of Electronics</i> , vol. 42, pp. 521-1977.
[78]	R. B. Ridley, "A new, continuous-time model for current-mode control [pc convertors]," <i>Power Electronics, IEEE Transactions on,</i> vol. 6, pp. 271-280, 1991
[79]	G. C. Verghese, M. E. Elbuluk, and J. G. Kassakian, "A General Approach to Samp Data Modeling for Power Electronic Circuits," <i>Power Electronics, IEEE Transac</i> <i>on</i> , vol. PE-1, pp. 76-89, 1986.
[80]	G. W. Wester and R. D. Middlebrook, "Low-Frequency Characterization of Swit dc-dc Converters," <i>Aerospace and Electronic Systems, IEEE Transactions on,</i> vol. 9, pp. 376-385, 1973.
[81]	V. Vorperian, "Simplified analysis of PWM converters using model of PWM sw Continuous conduction mode," <i>Aerospace and Electronic Systems, IEEE Transactory</i> <i>on</i> , vol. 26, pp. 490-496, 1990.
[82]	V. Vorperian, "Simplified analysis of PWM converters using model of PWM sw II. Discontinuous conduction mode," <i>Aerospace and Electronic Systems,</i> <i>Transactions on</i> , vol. 26, pp. 497-505, 1990.
[83]	D. Czarkowski and M. K. Kazimierczuk, "Energy-conservation approach to mode PWM DC-DC converters," <i>Aerospace and Electronic Systems, IEEE Transaction</i> , vol. 29, pp. 1059-1063, 1993
[84]	D. M. VandeSype, K. DeGusseme, F. M. L. L. DeBelie, A. P. VandenBossche, and Melkebeek, "Small-Signal z-Domain Analysis of Digitally Controlled Convert <i>Power Electronics, IEEE Transactions on</i> , vol. 21, pp. 470-478, 2006.
[85]	D. Maksimovic and R. Zane, "Small-Signal Discrete-Time Modeling of Dig Controlled PWM Converters," <i>Power Electronics, IEEE Transactions on,</i> vol. 22 2552-2556, 2007.
[86]	D. M. Mitchell, "Pulsewidth modulator phase shift," <i>Aerospace and Electic Systems, IEEE Transactions on,</i> vol. AES-16, pp. 272-278, 1980.
[87]	R. Tymerski, "Frequency analysis of time-interval-modulated switched netwo <i>Power Electronics, IEEE Transactions on,</i> vol. 6, pp. 287-295, 1991.
[88] [89]	J. T. Tou, <i>Digital and sampled-data control systems</i> . USA: McGraw-Hill, 1959. P. Hao, A. Prodic, E. Alarcon, and D. Maksimovic, "Modeling of Quantization Ef in Digitally Controlled DC-DC Converters," <i>Power Electronics, IEEE Transaction</i> , vol. 22, pp. 208-215, 2007.
[90]	A. V. Peterchev and S. R. Sanders, "Quantization resolution and limit cyclin digitally controlled PWM converters," <i>Power Electronics, IEEE Transactions on</i> , 18, pp. 301-308, 2003.
[91]	W. Stefanutti, P. Mattavelli, S. Saggini, and G. Garcea, "Energy-based approac predicting limit cycle oscillations in voltage-mode digitally-controlled c converters," in <i>Applied Power Electronics Conference and Exposition. APEC</i> <i>Twenty-First Annual IEEE</i> , 2006, p. 7 pp.

Chapte	er 1 Introduction and literature review
[92]	G. F. Franklin, J. D. Powell, and M. L. Workman, <i>Digital control of dynamic systems</i> . USA: Addison Wesley Longman 1998.
[93]	H. Al-Atrash and I. Batarseh, "Digital controller design for a practicing power electronics engineer," in <i>Applied Power Electronics Conference, APEC, Twenty Second Annual IEEE</i> , 2007, pp. 34-41.
[94]	B. C. Kuo and F. Golnaraghi, <i>Automatic control systems</i> , 2nd ed. USA: John Wiley & Sons Inc., 2003.
[95]	Y. Duan and K. Jin, "Digital controller design for switchmode power converters," in <i>Applied Power Electronics Conference and Exposition. APEC '99. Fourteenth Annual</i> , 1999, pp. 967-973 vol.2.
[96]	T. W. Martin and S. S. Ang, "Digital control for switching converters," in <i>Industrial Electronics. ISIE '95., Proceedings of the IEEE International Symposium on</i> , 1995, pp. 480-484 vol.2.
[97]	M. Veerachary, T. Senjyu, and K. Uezato, "Modeling and analysis of interleaved dual boost converter," in <i>Industrial Electronics. Proceedings. ISIE '01. IEEE International Symposium on</i> , 2001, pp. 718-722 vol.2.
[98]	X. Haiping, Q. Ermin, G. Xin, W. Xuhui, and K. Li, "Analysis and Design of High Power Interleaved Boost Converters for Fuel Cell Distributed Generation System," in <i>Power</i> <i>Electronics Specialists Conference. PESC '05. IEEE 36th</i> , 2005, pp. 140-145.
[99]	O. F. Ruiz and I. Cervantes, "Averaged modeling of transformer-coupled interleaved boost converters," in <i>Industrial Electronics Society. IECON '12. 38th Annual Conference</i> , 2012, pp. 256-261.
[100]	C. Li-Hsiu, H. Jonq-Chin, Y. Sheng-Nian, and Y. Ging-Chou, "Analysis and Design of Four-Leg Fuel Cell Boost Converter," in <i>IEEE Industrial Electronics. IECON '06. 32nd Annual Conference on</i> , 2006, pp. 4285-4290.
[101]	D. De, C. Klumpner, C. Patel, K. Ponggorn, M. Rashed, and G. Asher, "Modelling and control of a multi-stage interleaved DC-DC converter with coupled inductors for super-capacitor energy storage system," <i>Power Electronics, IET,</i> vol. 6, pp. 1360-1375, 2013.
[102]	A. J. Forsyth and G. Calderon-Lopez, "Sampled-Data Analysis of the Dual-Interleaved Boost Converter With Interphase Transformer," <i>Power Electronics, IEEE</i> <i>Transactions on</i> , vol. 27, pp. 1338-1346, 2012.
[103]	M. Veerachary, "Analysis of interleaved dual boost converter with integrated magnetics: signal flow graph approach," <i>Electric Power Applications, IEE Proceedings</i> , vol. 150, pp. 407-416, 2003.
[104]	S. K. Mazumder, "Stability analysis of parallel DC-DC converters," <i>Aerospace and Electronic Systems, IEEE Transactions on,</i> vol. 42, pp. 50-69, 2006.
[105]	S. C. Sivakumar, S. Sivakumar, and H. P. Tran, "Modelling and control of interleaved parallel converters with nonidentical parameters," in <i>Telecommunications Energy Conference, INTELEC '01. Twenty-Third International</i> , 2001, pp. 266-273.
[106]	H. B. Shin, E. S. Jang, J. G. Park, H. W. Lee, and T. A. Lipo, "Small-signal analysis of multiphase interleaved boost converter with coupled inductors," <i>Electric Power Applications, IEE Proceedings,</i> vol. 152, pp. 1161-1170, 2005.
	70
	70

Chapter 2 Converter operation, modelling and digital control

2.1. Introduction

This Chapter begins with a brief analysis of the dual-interleaved boost converter with inter-phase transformer under steady-state, continuous conduction mode. The main waveforms are examined and their relevant mathematical expressions are presented. Then, a methodology to obtain the averaged small-signal ac model of the converter is described and the state-space representation of this model is derived. Next, the averaged small-signal ac model is validated by the use of a switched and a large-signal averaged model implemented in SABER. Finally, the last section explains the principles of the digital phase-current regulation for the converter employing uniformly-sampled pulse-width modulation. Three different modulation strategies are described: leading-edge modulation, trailing-edge modulation and triangular carrier modulation with symmetric on-time.

2.2. Operation of the dual interleaved boost converter with inter-phase transformer

Figure 2.1 presents the circuit diagram of the dual interleaved boost converter with inter-phase transformer, DIBC with IPT. The circuit comprises an H-bridge formed by IGBTs connected with antiparallel diodes. The mid-point nodes of the switching legs of the H-bridge are denoted *A* and *B* in the circuit diagram. This will be used to designate the converter phases. In addition, the IGBTs and their respective diodes are designated according to their position in the switching legs, namely the upper transistors and diodes

 Q_{2a} , D_{2a} , Q_{2b} and D_{2b} ; and the bottom ones Q_{1a} , D_{1a} , Q_{1b} and D_{1b} . For the purposes of this work, the circuit is assumed to operate in boost or step-up mode only. In consequence, only the bottom transistors, Q_{1a} and Q_{1b} , are operated and their corresponding quiescent duty-ratios are referred to as D_a and D_b . The top transistors are deactivated at all times allowing their antiparallel diodes, D_{2a} and D_{2b} , to handle the current flowing towards the load. Normally, under steady-state conditions the duty-ratios of Q_{1a} and Q_{1b} would be equal, but with the turn-on instant of Q_{1b} delayed by half a cycle behind Q_{1a} thereby producing the interleaved operation of the circuit.



Figure 2.1. Dual interleaved boost converter with inter-phase transformer.

A pair of tightly, inverse-coupled inductors is connected to nodes *A* and *B* of the Hbridge, and the centre-tap of the coupled inductors forms the connection to the inputinductor *L*_{in}. Assuming that the input current divides equally between the windings of the IPT, the dc flux is zero due to the inverse coupling of the windings and only ac flux is present in the core. The IPT acts as an inductive voltage-divider and furthermore, due to the phaseshift between the boost cells, the frequency of the inductor's voltage is twice the switching frequency.

In the schematic of the circuit, the self-inductance of the coupled inductors is denoted L_a and L_b respectively, whilst their mutual inductance will be denoted L_m . The total inductance measured between the nodes A and B, will be referred to as the differential inductance of the IPT, $L_{diff} = L_a + L_b + 2L_m$. Finally, the positive and negative rails of the H-bridge are connected to the converter output, comprising the output capacitor, C_o , connected in parallel with the load resistance, R_{load} .
Under steady-state conditions, $D_a = D_b = D$, and assuming that the current flowing through the IPT windings is continuous, three operating modes can be identified: D < 0.5, D = 0.5 and D > 0.5. Figure 2.2 illustrates the principal voltage and current waveforms of the converter for these operating modes. The waveforms were sketched under the assumption that $L_a = L_b$ and that the windings are perfectly coupled. Furthermore it is assumed that the input-current divides equally between the two halves of the IPT.

The first graph of each group shows the transistor gate-to-emitter voltages. The solid waveform corresponds to the phase-a transistor, $v_{ge1a}(t)$, while the dashed waveform belongs to the transistor of phase-b, $v_{ge1b}(t)$. The second and third graphs show the input inductor voltage, $v_{Lin}(t)$, and the voltage between the nodes A and B, also known as the differential voltage, $v_{diff}(t)$. In the next two graphs, the input inductor current, $i_{Lin}(t)$, and the differential current, $i_{diff}(t)$, are sketched. Finally the last graph depicts the currents flowing through the IPT windings, $i_a(t)$ and $i_b(t)$, in solid and dashed waveforms respectively.

The converter can assume four different circuit configurations depending on the combination of active diodes/transistors. These configurations are identified with Roman numerals at the top of the waveforms and are illustrated in Figure 2.3. For D < 0.5, the converter switches between configurations I, II and III. For D > 0.5 configurations I, III, and IV occur. Finally, for the case of D = 0.5 only configurations I and II are present.

In configuration I transistor Q_{1a} is activated clamping node A to ground. Simultaneously diode D_{2b} is in conduction connecting node B to the converter output, consequently the differential voltage $v_{diff}(t)$ equals $-V_o$, where V_o is the average value of the output voltage. Additionally, a voltage divider is formed by the IPT windings, and for this reason the voltage level across the input inductor is $v_{Lin}(t) = V_{in} - V_o/2$.

In configuration II transistors Q_{1a} and Q_{1b} are deactivated whilst diodes D_{2a} and D_{2b} are in conduction. As a consequence, nodes A and B are connected to the same terminal and therefore $v_{diff}(t)$ is equal to zero. Furthermore, owing to the perfect coupling of the IPT the windings appear as a short-circuit to the input-current $i_{in}(t)$, therefore the input inductor voltage is $v_{Lin}(t) = V_{in} - V_{o}$, where V_{in} is the average value of the input voltage.

Chapter 2

Converter operation, modelling and digital control



74



Figure 2.3. Equivalent circuits of the dual interleaved boost converter with inter-phase transformer for continuous conduction mode of operation.

Configuration III arises when Q_{1b} is activated and D_{2a} is conducting. This configuration is a mirror image of configuration I where the differential voltage $v_{diff}(t)$ is now equal to $+V_o$ and the input inductor voltage is $v_{Lin}(t) = V_{in} - V_o/2$.

Finally Configuration IV arises when both transistors Q_{1a} and Q_{1b} are activated clamping the nodes A and B to ground and making $v_{diff}(t)$ equal to zero. The voltage seen across the input inductor in this case is $v_{Lin}(t) = V_{in}$.

Whenever the converter operates with $D \neq 0.5$ it switches between four configurations over a switching period. For this reason the input inductor voltage in Figure 2.2(a) and 2.2(c) is an asymmetrical square waveform alternating between two levels at twice the switching frequency. The differential voltage instead, is a three-step waveform at the switching frequency with a peak-to-peak value of twice V_o .

The voltage conversion ratio of the converter, is obtained by equating the positive and negative volt-seconds across L_{in} and is identical to that for the single-phase boost converter operating in the continuous conduction mode, [1,2]:

$$\frac{V_o}{V_{in}} = \frac{1}{1 - D}$$
(2.1)

From Equation (2.1) and assuming that the converter is lossless, a reciprocal expression for the current conversion ratio may be determined:

$$I_{Lin} = \frac{V_o}{R_{load} (1-D)} = \frac{V_{in}}{R_{load} (1-D)^2}$$
(2.2)

where R_{load} is the load resistor. The input inductor ripple current, ΔI_{Lin} , and the peak-topeak value of the differential current, ΔI_{diff} , can be inferred from their respective voltage waveforms. The equations to calculate these ripples are shown in Table 2.1. Of particular note is the fact that the input ripple current, ΔI_{Lin} , is zero for D = 0.5.

As the input current divides equally between the IPT windings, the phase currents are equal to half of the input current plus the differential current generated by the differential voltage across the IPT. Therefore, the resulting phase currents can be expressed as:

$$i_{a}(t) = \frac{i_{Lin}(t)}{2} - i_{diff}(t)$$
 (2.3)

$$i_{b}(t) = \frac{i_{Lin}(t)}{2} + i_{diff}(t)$$
 (2.4)

	D < 0.5	D = 0.5	D > 0.5
Input current ripple	$\Delta I_{Lin} = \frac{V_{in}DT}{2L_{in}} \left(\frac{1-2D}{1-D}\right)$	$\Delta I_{Lin} = 0$	$\Delta I_{Lin} = \frac{V_{in}T}{2L_{in}} (2D-1)$
Differential current ripple	$\Delta I_{diff} = \frac{V_{in}DT}{L_{diff}} \left(\frac{1}{1-D}\right)$	$\Delta I_{diff} = \frac{V_{in}T}{L_{diff}}$	$\Delta I_{diff} = \frac{V_{in}T}{L_{diff}}$
Large phase- current ripple	$\Delta I_{\rm Irg} = \frac{V_{\rm in}DT}{\left(1-D\right)} \left(\frac{1-2D}{4L_{\rm in}} + \frac{1}{L_{\rm diff}}\right)$	$\Delta I_{lrg} = \frac{V_{in}T}{L_{diff}}$	$\Delta I_{\rm irg} = V_{in}T\left(\frac{2D-1}{4L_{in}} + \frac{1}{L_{diff}}\right)$
Small phase- current ripple	$\Delta I_{smll} = \frac{V_{in}DT}{(1-D)} \left(\frac{1-2D}{4L_{in}} - \frac{1}{L_{diff}}\right)$	$\Delta I_{srdl}=0$	$\Delta I_{smil} = V_{in}T\left(\frac{2D-1}{4L_{in}} - \frac{1}{L_{diff}}\right)$

Table 2.1. Summary of ripple current expressions for the DIBC with IPT. $D_a = D_b = D$.

The waveforms corresponding to these currents are illustrated in the last graph of Figure 2.2. Two different ripple levels denoted large ripple-current, ΔI_{lrg} , and small ripple-current, ΔI_{smll} , are identified in these current waveforms.

Table 2.1 summarizes the expressions of the ripple-currents for $i_{Lin}(t)$, $i_{diff}(t)$, $i_a(t)$ and $i_b(t)$ for the three modes of operation of the DIBC with IPT. The derivation of these expressions is given in Appendix A.

2.3. Small-signal ac modelling of the converter

The small-signal model of the dual interleaved boost converter with IPT employed in this work is based on an extension of the well-known circuit averaging method. This technique referred to as averaged switch modelling by Erickson, [3], aims to replace the converter switching elements by an equivalent averaged switch model, thereby eliminating the time-dependent-switching of the circuit topology. The resulting equivalent circuit of the converter can then be perturbed and linearized to obtain a small-signal model. The following section will review this procedure for the single-phase boost converter.

2.3.1. Averaged switch modelling of the boost converter

Figure 2.4(a) depicts a single-phase boost converter switch-network comprising a transistor and a diode. The instantaneous voltages and currents at the ports of this network are denoted $v_1(t)$, $i_1(t)$, $v_2(t)$ and $i_2(t)$. For convenience ports 1 and 2 will be referred to as the input port and the output port respectively. Assuming lossless components and instantaneous switching, the waveforms of $v_1(t)$ and $i_2(t)$ are sketched in Figure 2.4(b). In this Figure a single period of the waveforms is illustrated. From Figure 2.4(b), $v_1(t)$ and $i_2(t)$ can be expressed as:

$$v_{1}(t) = \begin{cases} 0 & 0 < t < dT \\ v_{2}(t) & dT < t < T \end{cases}$$
(2.5)

$$i_{2}(t) = \begin{cases} 0 & 0 < t < dT \\ i_{1}(t) & dT < t < T \end{cases}$$
(2.6)



Figure 2.4. (a) Single-phase boost converter switch network. (b) Switch network waveforms. (c) Equivalent dc and ac small-signal averaged switch network model.

Under the assumption that the switching ripples of $v_1(t)$ and $i_2(t)$ are small or at least linear functions of time, Equations (2.5) and (2.6) can be averaged over a switching period, T, yielding:

$$\langle v_1(t) \rangle_{\tau} = (1 - d(t)) \langle v_2(t) \rangle_{\tau}$$
 (2.7)

$$\langle i_2(t) \rangle_{\tau} = (1 - d(t)) \langle i_1(t) \rangle_{\tau}$$
 (2.8)

Equations (2.7) and (2.8) may then be perturbed and linearized about a quiescent point of operation yielding:

$$V_1 + \tilde{v}_1 = (1 - D)(V_2 + \tilde{v}_2) - \tilde{d}V_2 - \tilde{d}\tilde{v}_2$$
(2.9)

$$I_{2} + \tilde{i}_{2} = (1 - D)(I_{1} + \tilde{i}_{1}) - \tilde{d}I_{1} - \tilde{d}\tilde{i}_{1}$$
(2.10)

where the voltage and currents at the input and output ports and the duty-ratio are expressed as the sum of the quiescent value plus a small ac variation: $V_1 + \tilde{v}_1$, $l_1 + \tilde{i}_1$, $V_2 + \tilde{v}_2$, $l_2 + \tilde{i}_2$ and $D + \tilde{d}$. The equivalent averaged switch model, Figure 2.4(c), can be obtained from Equations (2.9) and (2.10) by neglecting the nonlinear terms $\tilde{d}\tilde{v}_2$ and $\tilde{d}\tilde{i}_1$. This assumption is valid provided that the ac variations are much smaller than the quiescent values. The equivalent averaged circuit shown in Figure 2.4(c) is an alternative version of the original one devised by Vórperian [4] which results in a simpler boost circuit model. The equivalent switch comprises three elements:

1. A dependent voltage source, $\tilde{d}V_2$, where \tilde{d} is the small ac variation of the converter duty ratio and V_2 is the quiescent value of the voltage at the output port of the switch network.

- An ideal (dc) transformer with turns-ratio D':1, where D' = (1 D) and D is the quiescent value of the transistors duty ratio.
- 3. A dependent current source, $\tilde{d}I_1$, where the term I_1 is the dc or quiescent value of the current at the input port of the switch network.

The procedure followed to obtain the model of the DIBC converter using the switch model of Figure 2.4(c) will be explained next.

2.3.2. Averaged small-signal modelling of the dual interleaved boost converter

The DIBC with IPT is shown in Figure 2.5 where the transistors of each boost cell are controlled by the independent sources $d_a(t)$ and $d_b(t)$. Also two additional elements are included: the series resistance of the input inductor, R_{in} , and a small-signal current source, $i_2(t)$, connected in parallel with the load resistor. The latter element is also known as a perturbation current and can be employed to emulate load changes in the circuit model. Two switch networks are identified in this circuit which can be replaced by the averaged model of Figure 2.4(c).

Moreover, to model this converter accurately it is necessary to account for the presence of the inter-phase transformer. As demonstrated by Maksimović in [5], the well-known model of a transformer based on self and mutual inductances is well suited when the leakage inductance of the coupled inductors does not play a determinant role in the converter operation.



Figure 2.5. Open loop circuit of the dual interleaved boost converter with IPT.



Figure 2.6. dc and small-signal averaged model of the DIBC with IPT.

Upon replacement of the equivalent switch networks and substitution of the IPT by its equivalent model, the dc and averaged small-signal ac model of the converter shown in Figure 2.6 is obtained. In this new circuit the voltage at the input port of each averaged switch network is equal to the local average voltage at the nodes *A* and *B* of the original circuit respectively, whilst the input port currents are easily identified as the perturbed local average phase currents $I_a + \tilde{i}_a$ and $I_b + \tilde{i}_b$.

Owing to the parallel connection of the switch networks, the output port voltage of each network is the same as the converter output voltage. Similarly, the output currents at each network are denoted $I_{oa} + \tilde{i}_{oa}$ and $I_{ob} + \tilde{i}_{ob}$ which comprise together the total output current, $I_{oab} + \tilde{i}_{oab}$, delivered to C_o and R_{load} . Finally, the current flowing through C_o is denoted \tilde{i}_c and the current delivered to the load resistance is $I_o + \tilde{i}_o$.

By inspection of Figure 2.6 and employing Equation (2.10), the current at the output ports of the switch networks A and B can be written as:

$$\left(I_{oa}+\tilde{i}_{oa}\right)=\left(1-D_{a}\right)\left(I_{a}+\tilde{i}_{a}\right)-I_{a}\tilde{d}_{a}$$
(2.11)

$$\left(I_{ob} + \tilde{i}_{ob}\right) = \left(1 - D_{b}\right) \left(I_{b} + \tilde{i}_{b}\right) - I_{b}\tilde{d}_{b}$$

$$(2.12)$$

Adding Equations (2.11) and (2.12) to obtain the total converter output current, yields:

Converter operation, modelling and digital control

$$I_{o} + \tilde{i}_{o} + \tilde{i}_{c} + \tilde{i}_{z} = (1 - D_{o})(I_{o} + \tilde{i}_{o}) - I_{o}\tilde{d}_{o} + (1 - D_{b})(I_{b} + \tilde{i}_{b}) - I_{b}\tilde{d}_{b}$$
(2.13)

Equation (2.13) may be split by grouping the steady-state and the small-signal terms from both sides of the equation into two separate equations:

$$I_{o} = (1 - D_{a})I_{a} + (1 - D_{b})I_{b}$$
(2.14a)

$$\tilde{i}_{o} + \tilde{i}_{c} + \tilde{i}_{z} = (1 - D_{o})\tilde{i}_{o} - I_{o}\tilde{d}_{o} + (1 - D_{b})\tilde{i}_{b} - I_{b}\tilde{d}_{b}$$
 (2.14b)

Under steady-state conditions, the quiescent value of the duty-ratios D_a and D_b can be assumed identical, in other words $D_a = D_b = D$. Furthermore as I_{Lin} divides equally between the IPT windings, the quiescent value of the phase currents I_a and I_b is equal to $I_{Lin}/2$. Under these assumptions, Equations (2.14a) and (2.14b) can be rearranged as shown in Equations (2.15a) and (2.15b) respectively,

$$I_o = I_{Lin} (1 - D) \tag{2.15a}$$

$$C_{o} \frac{d \tilde{v}_{o}}{dt} = (1-D)\tilde{i}_{a} + (1-D)\tilde{i}_{b} - \frac{l_{Lin}}{2}\tilde{d}_{a} - \frac{l_{Lin}}{2}\tilde{d}_{b} - \frac{1}{R_{load}}\tilde{v}_{o} - \tilde{i}_{z}$$
(2.15b)

where $\tilde{i}_c = C_o \frac{d\tilde{v}_o}{dt}$ and $\tilde{i}_o = \frac{\tilde{v}_o}{R_{load}}$. Equation (2.15b) is the differential equation for the converter output voltage. To determine the differential equations for the phase currents, the voltage equations for the circuit loops around each phase may be written as:

$$\left(V_{in}+\tilde{V}_{in}\right) = \left(I_{Lin}+\tilde{I}_{Lin}\right)R_{in}+L_{Lin}\frac{d\left(I_{Lin}+\tilde{I}_{Lin}\right)}{dt}+L_{a}\frac{d\left(I_{a}+\tilde{I}_{a}\right)}{dt}-L_{m}\frac{d\left(I_{b}+\tilde{I}_{b}\right)}{dt}-\tilde{d}_{a}V_{o}+\left(V_{o}+\tilde{V}_{o}\right)\left(1-D_{a}\right)$$
 (2.16a)

$$\left(V_{in}+\tilde{v}_{in}\right) = \left(I_{Lin}+\tilde{i}_{Lin}\right)R_{in}+L_{Lin}\frac{d\left(I_{Lin}+\tilde{i}_{Lin}\right)}{dt}+L_{b}\frac{d\left(I_{b}+\tilde{i}_{b}\right)}{dt}-L_{m}\frac{d\left(I_{a}+\tilde{i}_{a}\right)}{dt}-\tilde{d}_{b}V_{o}+\left(V_{o}+\tilde{v}_{o}\right)\left(1-D_{b}\right)$$
(2.16b)

By substituting $I_{Lin} + \tilde{i}_{Lin} = (I_a + \tilde{i}_a) + (I_b + \tilde{i}_b)$ and separating the steady-state and the small-signal terms on each side of Equations (2.16a) and (2.16b), Equations (2.17a) to (2.18b) can be found:

$$V_{in} = I_{Lin}R_{in} + V_o(1 - D_o)$$
 (2.17a)

$$\tilde{\nu}_{in} = \left(L_{Lin} + L_a\right) \frac{d\tilde{l}_a}{dt} + \left(L_{Lin} - L_m\right) \frac{d\tilde{l}_b}{dt} + R_{in}\tilde{l}_a + R_{in}\tilde{l}_b - V_o\tilde{d}_a + (1 - D_a)\tilde{\nu}_o$$
(2.17b)

Converter operation, modelling and digital control

$$V_{in} = I_{Lin}R_{in} + V_o(1 - D_b)$$
 (2.18a)

$$\tilde{v}_{in} = \left(L_{in} + L_b\right) \frac{d\tilde{l}_b}{dt} + \left(L_{in} - L_m\right) \frac{d\tilde{l}_a}{dt} + R_{in}\tilde{l}_a + R_{in}\tilde{l}_b - V_o\tilde{d}_b + \left(1 - D_b\right)\tilde{v}_o$$
(2.18b)

Once more, assuming steady-state conditions, $D_a = D_b = D$, Equations (2.17a) and (2.18a) may be written as:

$$V_{in} = I_{Lin}R_{in} + V_o(1-D)$$
(2.19)

whilst Equations (2.17b) and (2.18b) can be expressed as:

$$\tilde{v}_{in} = \left(L_{Lin} + L_a\right) \frac{d\tilde{l}_a}{dt} + \left(L_{Lin} - L_m\right) \frac{d\tilde{l}_b}{dt} + R_{in}\tilde{l}_a + R_{in}\tilde{l}_b - V_o\tilde{d}_a + (1-D)\tilde{v}_o$$
(2.20a)

$$\tilde{v}_{in} = \left(L_{in} + L_b\right) \frac{d\tilde{i}_b}{dt} + \left(L_{in} - L_m\right) \frac{d\tilde{i}_a}{dt} + R_{in}\tilde{i}_a + R_{in}\tilde{i}_b - V_o\tilde{d}_b + (1-D)\tilde{v}_o$$
(2.20b)

From Equation (2.19) the conversion ratio of the converter when the input inductor resistance is included may be derived by substituting Equation (2.15a) into Equation (2.19):

$$\frac{V_o}{V_{in}} = \frac{(1-D)}{(1-D)^2 + \frac{R_{in}}{R_{load}}}$$
(2.21)

It is evident that if the ratio of the input inductor resistance to the load resistance is sufficiently small, in other words $R_{in} \ll R_{load}$, Equation (2.21) will reduce to Equation (2.1).

Moreover, rearranging Equations (2.20a) and (2.20b) into two first-order differential equations for \tilde{i}_a and \tilde{i}_b , yields:

$$\frac{d\tilde{i}_{a}}{dt} = \frac{V_{o}(\mathsf{L}_{in}+\mathsf{L}_{c})}{L_{\tau_{ot}}}\tilde{d}_{a} - \frac{V_{o}(\mathsf{L}_{in}+\mathsf{L}_{c})}{L_{\tau_{ot}}}\tilde{d}_{b} - \frac{(1-\mathsf{D})(\mathsf{L}_{m}+\mathsf{L}_{c})}{L_{\tau_{ot}}}\tilde{v}_{o} - \frac{R_{in}(\mathsf{L}_{m}+\mathsf{L}_{c})}{L_{\tau_{ot}}}\tilde{i}_{a} - \frac{R_{in}(\mathsf{L}_{m}+\mathsf{L}_{c})}{L_{\tau_{ot}}}\tilde{i}_{b} \quad (2.22a)$$

$$\frac{d\tilde{i}_{b}}{dt} = \frac{V_{o}(L_{in}+L_{c})}{L_{Tot}}\tilde{d}_{b} - \frac{V_{o}(L_{in}+L_{c})}{L_{Tot}}\tilde{d}_{a} - \frac{(1-D)(L_{m}+L_{c})}{L_{Tot}}\tilde{v}_{o} - \frac{R_{in}(L_{m}+L_{c})}{L_{Tot}}\tilde{i}_{a} - \frac{R_{in}(L_{m}+L_{c})}{L_{Tot}}\tilde{i}_{b} \quad (2.22b)$$

where $L_{Tot} = 2L_{in}(L_c + L_m) + (L_c^2 - L_m^2)$, $L_a = L_b = L_c$ and $\tilde{v}_{in} = 0$. Furthermore V_o can be expressed in terms of V_{in} using the voltage conversion ratio of the converter shown in Equation (2.1) under the assumption that $R_{in} \ll R_{load}$. Using Equations (2.15), (2.22a) and (2.22b), the small-signal model of the converter can be represented in state-space form as:

Converter operation, modelling and digital control

$$\dot{\tilde{\mathbf{x}}} = \mathbf{A}_{av}\tilde{\mathbf{x}} + \mathbf{B}_{av}\tilde{\mathbf{u}}$$
(2.23)

where the state vector, $\tilde{\mathbf{x}}$, comprises the small-signal components of the phase currents flowing through the IPT windings and the voltage across the output capacitor: $\tilde{\mathbf{x}} = \begin{bmatrix} \tilde{i}_a & \tilde{i}_b & \tilde{v}_a \end{bmatrix}^T$; the input vector, $\tilde{\mathbf{u}}$, contains the control inputs of each phase and the load disturbance current: $\tilde{\mathbf{u}} = \begin{bmatrix} \tilde{d}_a & \tilde{d}_b & \tilde{i}_z \end{bmatrix}^T$. Moreover, if the small ac perturbations of the input voltage are included in the development of the model, the vector $\tilde{\mathbf{u}}$ will contain a fourth variable \tilde{V}_{in} . The matrices A_{av} and B_{av} in Equation (2.23) may be expressed by Equations (2.24) and (2.25):

$$\mathbf{A}_{ov} = \begin{bmatrix} -\frac{R_{in}(L_{c} + L_{m})}{L_{tot}} & -\frac{R_{in}(L_{c} + L_{m})}{L_{tot}} & -\frac{(1 - D)(L_{c} + L_{m})}{L_{tot}} \\ -\frac{R_{in}(L_{c} + L_{m})}{L_{tot}} & -\frac{R_{in}(L_{c} + L_{m})}{L_{tot}} & -\frac{(1 - D)(L_{c} + L_{m})}{L_{tot}} \\ \frac{(1 - D)}{C_{o}} & \frac{(1 - D)}{C_{o}} & -\frac{1}{R_{load}C_{o}} \end{bmatrix}$$
(2.24)
$$\mathbf{B}_{ov} = \begin{bmatrix} \frac{V_{in}(L_{in} + L_{c})}{L_{tot}(1 - D)} & -\frac{V_{in}(L_{in} - L_{m})}{L_{tot}(1 - D)} & 0 \\ -\frac{V_{in}(L_{in} - L_{m})}{L_{tot}(1 - D)} & \frac{V_{in}(L_{in} + L_{c})}{L_{tot}(1 - D)} & 0 \\ -\frac{V_{in}}{2R_{load}C_{o}(1 - D)^{2}} & -\frac{V_{in}}{2R_{load}C_{o}(1 - D)^{2}} & \frac{1}{C_{o}} \end{bmatrix}$$
(2.25)

By the use of the Laplace-transformation the transfer functions of the state-space model were obtained. These are listed in Table 2.2 and their corresponding algebraic expressions are shown in Table 2.3. The procedure to find the transfer functions was carried out with the assistance of the software Mathematica, which allows the reduction of symbolic algebraic expressions. From the mathematical development of these expressions, it was found that the transfer functions for each of the converter phases are identical,

$$G_{daia}(s) = G_{dbib}(s) \tag{2.26}$$

$$G_{daib}(s) = G_{dbia}(s) \tag{2.27}$$

$$G_{davo}(s) = G_{dbvo}(s) \tag{2.28}$$

$$G_{izia}(s) = G_{izib}(s) \tag{2.29}$$

which is expected as the two phases of the converter are identical. Moreover, the transfer functions $G_{daib}(s)$ and $G_{dbia}(s)$, Equation (2.27), confirm the existence of a cross-coupling between the control input of one phase and the phase-current of its counterpart.

	ĩ̃ _a (s)	$\tilde{i}_{_b}(s)$	ĩv₀(s)
<i>d</i> (s)	$G_{daia}(\mathbf{s}) = \frac{\tilde{i}_{a}(\mathbf{s})}{\tilde{d}_{a}(\mathbf{s})}\Big _{\tilde{d}_{b}(\mathbf{s})=0; \ \tilde{i}_{z}(\mathbf{s})=0}$	$G_{daib}(s) = \frac{\tilde{i}_{b}(s)}{\tilde{d}_{a}(s)}\Big _{\tilde{d}_{b}(s)=0; \ \tilde{i}_{c}(s)=0}$	$G_{davo}(\mathbf{s}) = \frac{\tilde{\mathbf{v}}_{o}(\mathbf{s})}{\tilde{d}_{a}(\mathbf{s})}\Big _{\tilde{d}_{b}(\mathbf{s})=0; \tilde{\mathbf{i}}_{c}(\mathbf{s})=0}$
	Phase-a control-to- phase-a current	Phase-a control-to- phase-b current	Phase-a control-to- output voltage
<i></i> д́ _ь (s)	$G_{dbia}(\mathbf{s}) = \frac{\tilde{i}_{a}(\mathbf{s})}{\tilde{d}_{b}(\mathbf{s})}\Big _{\tilde{d}_{a}(\mathbf{s})=0; \tilde{i}_{b}(\mathbf{s})=0}$	$G_{dbib}(\mathbf{s}) = \frac{\tilde{i}_b(\mathbf{s})}{\tilde{d}_b(\mathbf{s})}\Big _{\tilde{d}_a(\mathbf{s})=0; \ \tilde{i}_b(\mathbf{s})=0}$	$G_{dbvo}(\mathbf{s}) = \frac{\tilde{v}_o(\mathbf{s})}{\tilde{d}_b(\mathbf{s})}\Big _{\tilde{d}_o(\mathbf{s})=0; \tilde{l}_b(\mathbf{s})=0}$
	Phase-b control-to- phase-a current	Phase-b control-to- phase-b current	Phase-b control-to- output voltage
ĩ̃,(s)	$\left G_{i_{zia}}(s) = \frac{\tilde{i}_{a}(s)}{\tilde{i}_{z}(s)} \right _{\tilde{d}_{a}(s)=0; \tilde{d}_{b}(s)=0}$	$\left \boldsymbol{G}_{izib}(\mathbf{s}) = \frac{\tilde{i}_{b}(\mathbf{s})}{\tilde{i}_{z}(\mathbf{s})} \right _{\tilde{d}_{a}(\mathbf{s})=0; \tilde{d}_{b}(\mathbf{s})=0}$	$G_{izvo}(s) = \frac{\tilde{v}_o(s)}{\tilde{i}_z(s)}\Big _{\tilde{d}_o(s)=0; \tilde{d}_b(s)=0}$
	Disturbance current-to- phase-a current	Disturbance current-to- phase-b current	Disturbance current-to- output voltage

Table 2.2. Transfer functions of the small-signal averaged model of the dual interleaved boostconverter with IPT.

In contrast to previous works [6,7], this averaged small-signal model has been developed for the boost converter and takes into consideration the small ac perturbation in each control input independently. This advantage is fully exploited in Chapter 4 where the model of the converter using digital average-current control is introduced, which includes the effects of the interleaved operation of the phases. The following sections will explain in more detail the implementation of the small-signal model in MATLAB and its validation using simulations carried out in SABER.



2.4. Simulation models of the converter

2.4.1. MATLAB/Simulink averaged small-signal ac model

The implementation of the small-signal ac model of the DIBC has two main sections: a script developed in MATLAB and a block diagram developed in Simulink. The script allows the user to set up the model parameters such as input voltage, quiescent duty-ratio, values of the converter passive elements and quiescent values of the state-variables.

This information was then used to calculate a numerical representation of the statespace matrices of the system. This script can be found in Appendix B.1. The data generated by the script was then employed in Simulink using the block diagram illustrated in Figure 2.7 to solve the system and plot its transient response. Finally, an additional script was generated in MATLAB to export the data obtained in Simulink and to compare it against the simulation results obtained with SABER.



Figure 2.7. MATLAB/Simulink small-signal ac model.

2.4.2. SABER switched model

This model was created to verify the correct operation of the converter and as the first stage of a more complex model which incorporates the digital current control. Figure 2.8 depicts the schematic diagram of the dual interleaved boost converter with IPT developed in SABER. In this diagram all the elements employed are standard SABER models provided with the simulation software. The circuit is divided into two sections.



Dual Interleaved Boost Converter Switched Model Version 3

Figure 2.8. SABER switched model of the dual interleaved boost converter with IPT.

The first section includes the power circuit components. An ideal dc voltage source is used to power the low-voltage side of the converter. Two ideal capacitors are employed at the low-voltage and high-voltage ends of the converter with no equivalent series resistance. The input inductor is modelled by an ideal inductor with inductance of $5.12 \,\mu$ H and an equivalent series resistance equal to $29 \,m\Omega$. The IPT windings were modelled using coupled inductors where, in SABER, the coupling coefficient between two target inductors can be set. The attributes of these elements are $75.14 \,\mu$ H for the self-inductance of the coupled inductors and 0.997 for the coupling coefficient. The values of these elements were obtained from a prototype of the converter, which is introduced in Section 3.2, Chapter 3. Finally, the devices that comprise the switching legs consist of a simple switch connected in anti-parallel with a piecewise linear diode. The switch model is controlled by a digital gate-signal that toggles the device on and off. The on-resistance, turn-on and turn-off rise times of the device can be selected by the user. In this particular application $0.1 \,\mu\Omega$, *10 ns* and *10 ns* were used for these properties respectively. These values were set to be very small to prevent switch related losses from affecting the correspondence between the simulation and the small-signal model and to reproduce ideal conditions. Additionally, a digital buffer is employed to drive the switch gate signal. By means of this buffer the propagation delay generated by the gate-drivers and the IGBT's turn-on and turn-off delays in the practical circuit are modelled. The delay set in this buffer equals *520 ns* and it corresponds to the practical value measured from the converter prototype in Section 3.6.3, Chapter 3.

The piecewise linear diode is modelled by a very high-resistance when the device is reverse-biased emulating an open circuit. Conversely, when the device is forward-biased the resistance between its terminals drops to the user specified value. The value employed for the simulations was $0.1 \mu \Omega$. No forward voltage drop was considered in this model.

The second section of the diagram contains the PWM generators used to provide the driving signals for the bottom transistors in the converter legs. Each generator comprises an analogue comparator, a 5 V dc voltage source and a voltage waveform generator. The waveform generators connected to the inverting input of the comparators were programmed to produce trailing-edged saw-tooth carrier waveforms with identical amplitude and frequency, but phase-shifted from each other by *180 degrees*. The noninverting inputs of the comparators are supplied with the modulating signal. This signal is produced by an additional dc voltage source which is common to the two phases.

2.4.3. SABER averaged model

The SABER averaged model, depicted in Figure 2.9, was created to provide further validation for the small-signal ac model of the converter developed in Section 2.2. This model, unlike the switched model, has no ripple in the current waveforms allowing the average values of the waveforms to be compared directly by superimposing the results.

The schematic shares practically the same structure as the one depicted in Figure 2.8, but unlike the switched model, the switch networks of each boost cell have been replaced with an averaged PWM switch network, which is available in the component library of SABER. The duty ratio of each averaged PWM switch network is controlled by two dc voltage sources located at the bottom of the schematic. Whilst being similar to the





Figure 2.9. SABER averaged large-signal model of the dual interleaved boost converter with IPT.

averaged switch modelling described in Section 2.3.1, no linearization or removal of steadystate components has been undertaken.

2.5. Simulation results

The simulation results obtained using the SABER switched model, SABER average model and the small-signal model of the converter are presented in this section. The parameters of the converter employed for these simulations are listed in Table 2.4 and they were obtained from a prototype of the converter which is described in Chapter 3. The

results are organized in two sections. The first section will demonstrate the operation of the SABER switched model and SABER averaged model in CCM. For this purpose three steady-state operating points are examined.

Parameter	Value		
Input inductance at 60 kHz, Lin	5.12 μΗ		
Input inductor stray resistance, R _{Lin}	29 mΩ		
IPT windings self-inductance at 30kHz, L _a , L _b	75.14 μH		
Coupling coefficient of the IPT windings, k	0.997		
Output capacitance, Co	45 μF		
Load Resistance, R _{load}	5.2 Ω		

Table 2.4. Simulation parameters for the validation of the small-signal model of the DIBC with

The second section will be used to validate the small-signal ac model implemented in MATLAB. To this end, the step response and the response to small-sinusoidal perturbations are compared against those obtained using the models developed in SABER.

2.5.1. Verification of the SABER switched and SABER averaged simulation models

The simulation results of the SABER switched model and the SABER averaged model introduced in Sections 2.4.2 and 2.4.3 are shown in Figures 2.10, 2.11 and 2.12. These results depict the steady-state waveforms of the converter for the operating points D = 0.25, D = 0.5 and D = 0.75 respectively, using an input voltage of 80 V and a load resistance of 5.2 Ω . The waveforms of both simulation models are displayed on top of each other and in the same order as the theoretical waveforms sketched in Figure 2.2, Section 2.2. In addition, the theoretical voltage and current levels as well as the ripple magnitudes were calculated for all the operating points shown using Equations (2.1) to (2.4) and those listed in Table 2.1. The results of these calculations together with the measurements on the simulation waveforms are summarized in Table 2.5 for the purpose of comparison.

The first graph in Figures 2.10, 2.11 and 2.12 displays the transistor driving signals $v_{ge1a}(t)$ and $v_{ge1b}(t)$. The duty-ratio of these waveforms was measured to ensure that the SABER switched model was at the desired operating point. As seen in Figures 2.10, 2.11 and 2.12, the rest of the converter waveforms are delayed with respect to the driving signals

due to the propagation delay introduced in the ideal switch to model the turn-on and turnoff delay of the IGBTs and gate-drivers.

Parameter	Theoretical		SABER switched model			
	D=0.25	D = 0.5	D = 0.75	D= 0.25	D = 0.5	D = 0.75
Input inductor voltage amplitude V _{Lin(pkpk)} [V]	53.3	0	160	53.4	0	160
Differential peak to peak voltage V _{diff(pkpk)} [V]	213.3	320	640	213.9	320.6	650
Average input current I _{Lin} [A]	27.3	61.5	246.1	26.9	61.6	250
Input current ripple <i>ΔI_{Lin}</i> [A]	43.4	0	130.2	43	0	130
Differential current ripple <i>∆I_{diff}</i> [A]	2.9	8.8	8.8	2.9	8.8	8.9
Phase current large-ripple ΔI _{alrg} , ΔI _{blrg} [A]	24.6	8.8	73.9	24	9	73
Average output voltage V _o [V]	106.6	160	320	110	160	320

Table 2.5. Comparison of the theoretical and simulation waveforms for D = 0.25, D = 0.5 and D = 0.75.

In the second graph $v_{Lin}(t)$ is displayed. The shape of the SABER switched model waveform exhibits the voltage levels expected from Figure 2.2. However, in contrast to the theoretical estimation, the lowest voltage level of $v_{Lin}(t)$ is not constant for D = 0.25 and D = 0.75 as the ripple of the output voltage is reflected in this waveform. For the particular case when D = 0.5, the $v_{Lin}(t)$ waveform is not completely smooth as expected. This was attributed to the switching delays in the device models resulting in slightly different switching instants in the two legs.

The third to the seventh graphs in Figures 2.10, 2.11 and 2.12 depict the differential voltage, $v_{diff}(t)$, the input inductor current $i_{Lin}(t)$, the differential current, $i_{diff}(t)$, and the phase-currents, $i_a(t)$ and $i_b(t)$, respectively. The shapes of the SABER switched model results for all these waveforms are in agreement with the theoretical ones, Figure 2.2. Moreover, the amplitude and average values measured from the simulation results match the

theoretical values calculated, Table 2.5. Here the SABER averaged model presents discrepancies in the average values of the waveforms $i_{Lin}(t)$, $i_a(t)$ and $i_b(t)$ in contrast to the switched model. These discrepancies are more evident at D = 0.75 and the relative error between the SABER averaged model and the SABER switched model and was found to be 1.2% for the phase currents and 0.6% for the input inductor current. These slight differences in the current were attributed to the differences in the circuit parameters, for example, the switch characteristics which are not taken into account in the averaged model.

Chapter 2



Figure 2.10. Steady state simulation results of the open loop switched model and the average model developed in SABER. D = 0.25; $V_{in} = 80$ V; $R_{Load} = 5.2 \Omega$.



Figure 2.11. Steady state simulation results of the open loop switched model and the average model developed in SABER. D = 0.5; $V_{in} = 80$ V; $R_{Load} = 5.2 \Omega$.

In conclusion, the SABER switched model operates correctly according to the theoretical analysis of the converter carried out in Section 2.2. Moreover, the SABER averaged model provides an accurate prediction of the average value of the waveforms.



Figure 2.12. Steady state simulation results of the open loop switched model and the average model developed in SABER. D = 0.75; $V_{in} = 80$ V; $R_{Load} = 5.2 \Omega$.

2.5.2. Validation of the small-signal ac model

In this section, the step response and the response to small sinusoidal perturbations of the small-signal model developed in Section 2.2 are compared against those obtained using the SABER averaged model and the SABER switched model.



2.5.2.1. Dynamic response

Figure 2.13. Transient response of the converter models for a synchronous 1% step increment in the duty ratio of both phases. $V_{in} = 80 V$; $R_{load} = 5.2 \Omega$. (a) $D_a = D_b = 0.2$; (b) $D_a = D_b = 0.4$; (c) $D_a = D_b = 0.6$; (d) $D_a = D_b = 0.8$.

The transient response of the converter phase-currents and output voltage to a 1%, 3% and 5% step increment in the duty-ratio are illustrated in Figures 2.13, 2.14 and 2.15 respectively. The step increment was performed simultaneously at the control inputs at t = 5 ms. In these Figures the simulation results of the SABER switched model, the SABER

Chapter 2

averaged model and the small-signal model are superimposed. The Figures are subdivided into four plots showing the response at different operating points. The operating points included are (a) D = 0.2, (b) D = 0.4, (c) D = 0.6 and (d) D = 0.8. The switching frequency was 30 kHz. The waveforms included on each plot are the phase-a current, $i_a(t)$, and the output voltage, $v_o(t)$. The phase-b current, $i_b(t)$, is not included since its behaviour is exactly the same as $i_a(t)$ for this particular case where both duty ratios are excited synchronously and with the same amplitude.



Figure 2.14. Transient response of the converter models for a synchronous 3% step increment in the duty ratio of both phases. $V_{in} = 80 V$; $R_{load} = 5.2 \Omega$. (a) $D_a = D_b = 0.2$; (b) $D_a = D_b = 0.4$; (c) $D_a = D_b = 0.6$; (d) $D_a = D_b = 0.8$.

As seen in the graphs, the results show that the waveforms of the SABER averaged model follow closely the oscillation envelope of the switched model waveforms for any condition or step increment. At the same time, the small signal model shows an acceptable Chapter 2

accuracy with the 1% step for all the operating points examined, Figure 2.13. However, the small-signal model fails to predict accurately the current and voltages for the 3% and 5% step increments with *D* above *0.5*, Figures 2.14(c), 2.14(d), 2.15(c) and 2.15(d). This is expected, as larger disturbances take the model away from the steady-state operating point, where the assumptions of the linearization process become less accurate.



Figure 2.15. Transient response of the converter models for a synchronous 5% step increment in the duty ratio of both phases. $V_{in} = 80 V$; $R_{load} = 5.2 \Omega$. (a) $D_a = D_b = 0.2$; (b) $D_a = D_b = 0.4$; (c) $D_a = D_b = 0.6$; (d) $D_a = D_b = 0.8$.

In order to quantify the accuracy of the small-signal model, the final average-value of $i_a(t)$ and $v_o(t)$ after the step-increment was measured and the relative error between the small-signal model and the average-values of the SABER switched model were calculated. The relative errors calculated for *D* ranging from 0.1 to 0.9 are summarized in the graphs shown in Figure 2.16. As seen from these graphs, the small-signal model has a maximum error of 1% for $v_o(t)$ and 2.5% for $i_a(t)$ when the system undergoes a 1% step-increment in

the duty-ratio. Furthermore, the maximum relative error for 3% and 5% step-increments in the duty-ratio is found to go well beyond 5% for values of D above 0.7, but remain less than 10% for duty-ratios up to 0.8.



Figure 2.16. Relative error of the final average-value of $i_a(t)$ and $v_o(t)$ predicted by the small-signal averaged model after a small step-increment in the duty-ratio. The relative error was calculated with respect to the value predicted by the SABER switched model.

2.5.2.2. Response to small-sinusoidal perturbations

This section presents the results of small sinusoidal perturbations in the averaged small-signal model implemented in MATLAB and the SABER averaged model. The results of this analysis are superimposed in the plots of Figure 2.17 verifying the validity of the small-

signal ac model. The following paragraphs will explain the methodology employed to plot these results and how they are organized.



Figure 2.17. Magnitude and phase response of the small-signal averaged model and the SABER averaged model to small-sinusoidal perturbations in the duty-ratio. (a),(b)&(c) D = 0.25; (d),(e)&(f) D = 0.75. $V_{in} = 80 V$, $R_{load} = 5.2 \Omega$.

The numerical state-space representation of the small-signal model of the converter was found using the MATLAB script developed in Section 2.4.1. This MATLAB script was

extended to compute the transfer functions in the Laplace-domain. For this study only the transfer functions relating the control inputs, $\tilde{d}_a(s)$ and $\tilde{d}_b(s)$, to the states of the system, $\tilde{i}_a(s)$, $\tilde{i}_b(s)$ and $\tilde{v}_o(s)$ were computed. These six transfer functions are denoted $G_{daia}(s)$, $G_{dbib}(s)$, $G_{daib}(s)$, $G_{daib}(s)$, $G_{daib}(s)$, $G_{daio}(s)$, $G_{davo}(s)$ and $G_{dbvo}(s)$ and it was demonstrated in Section 2.3 that they reduce to only three transfer functions as $G_{daia}(s) = G_{dbib}(s)$, $G_{daib}(s) = G_{dbia}(s)$, $G_{davo}(s) = G_{dbvo}(s)$. Finally, the magnitude and phase response of these transfer functions to small sinusoidal perturbations were plotted using MATLAB.

The response to small sinusoidal perturbations of the SABER averaged model was obtained employing the small-signal ac analysis tool included with the software. Prior to the small-signal ac analysis, a DC analysis on the circuit was run to obtain the quiescent operating point of all the variables. The data collected was then used to calculate the smallsignal response due to the variations at the control input of phase-a. The rest of the system inputs were set to zero.

Figure 2.17(a), 2.17(b) and 2.17(c) depict the frequency responses of transfer functions $G_{daia}(s)$, $G_{daib}(s)$ and $G_{davo}(s)$ respectively, for the operating point D = 0.25. The data collected from both models is superimposed in the graphs shown where the results of the SABER averaged model and the small-signal model are plotted with a solid and a dashed line respectively. Likewise, Figures 2.17(d), 2.17(e) and 2.17(f) illustrate the same plots for the operating point D = 0.75. The results plotted show that the frequency responses of the SABER averaged model and the small-signal transfer functions have virtually no difference, confirming the validity of the state-space representation of the small-signal model.

2.6. Digital current control with uniformly-sampled PWM for the dual interleaved boost converter with IPT

In this section the implementation of the digital current control with uniformly sampled PWM [8-10], will be analysed for its use in the dual interleaved boost converter with IPT. The principles of this strategy are reviewed in detail in Chapter 1, Section 1.3.2.1.



Figure 2.18. Functional block diagram of the digital current control with UPWM for the dual interleaved boost converter with IPT.

Figure 2.18 shows a schematic diagram of the converter with uniformly-sampled PWM. The diagram is divided in two sections. The top part illustrates the power-circuit whilst the bottom section depicts the current feedback loop implemented using a digital controller. The diagram is simplified by omitting the signal conditioning stages that would be needed in a practical system.

The digital controller regulates the current flowing through each phase of the converter using two individual current feedback loops operated in an interleaved manner. As the individual control loops share the same reference input, this topology ensures an equal distribution of the input current between the IPT windings. The PWM waveforms generated by the controller pass through gate-drivers to the converter transistors.

Chapter 2

The digital controller comprises two Digital Pulse-Width Modulators, DPWMs, and an Analogue-to-Digital Converter, ADC. The DPWM modules operate in a similar way as their analogue counter-part: a digital counter, which is analogous to a carrier waveform, is constantly compared against a reference-register by means of a digital comparator. The reference-register stores the value of the desired duty-ratio and is analogous to the modulating waveform of the analogue PWM.

When the digital counter crosses the value contained in the reference-register, the output of the comparator is toggled high or low depending on its configuration. Furthermore, the comparator can be configured to toggle its output state whenever its digital counter reaches either maximum or minimum count. The digital counters of each DPWM are phase-shifted by *180 degrees*, thereby generating phase-shifted PWM waveforms. The ADC sampling is linked to the operation of the DPWM counters and can be requested whenever they are either at their maximum or at their minimum count.

The operation of the digital control loops will be explained with the aid of the diagram of Figure 2.18 and the waveforms depicted in Figure 2.19. The waveforms of Figure 2.19 were sketched for the particular case where the DPWM counters operate in up-count mode and are reset to zero every time they reach their maximum value, V_{cmax} , which is known as leading-edge modulation. These waveforms are valid for any D < 0.5 when the converter is in CCM only, and illustrate non-periodic waveforms to show the repercussion that slight variations on the duty-ratio have on the currents of each phase.

The first plot in the waveforms shows the instantaneous phase currents $i_a(t)$ and $i_b(t)$ in solid and dashed lines respectively. The second plot shows, the reference-register waveform, $v_{ma}(t)$, and the counter waveform, $v_{ca}(t)$, of the phase-a DPWM module. The third plot shows the reference-register waveform, $v_{mb}(t)$, and the counter waveform, $v_{cb}(t)$. The fourth plot shows the sampling instants of the phase-a and the phase-b current waveforms. Finally, the last graph shows the PWM waveforms at the output of the DPWM modules, $v_{1a}(t)$ and $v_{1b}(t)$, in solid and dashed lines respectively. It is important to point out that in these illustrations even though $v_{ca}(t)$ and $v_{cb}(t)$ are generated by a digital counter, they are sketched as continuous instead of stepped waveforms.



Figure 2.19. Waveforms of the digital current control with uniformly-sampled pulse-width modulation from the dual interleaved boost converter with IPT. Leading-edge modulation strategy. Valid for CCM and $D \le 0.5$.

At the beginning of Figure 2.19 the phase-a DPWM counter is set to zero. At this instant three events take place: $v_{1a}(t)$ is set on, $v_{ma}(t)$ is updated and $i_a[n-1]$ and $i_{ref}[n-1]$ are acquired by sampling $i_a(t)$. The waveforms show that the sample $i_a[n-1]$ corresponds to the minimum-current or the valley-current. Meanwhile the phase-b DPWM counter, $v_{cb}(t)$, is increasing, as it is phase-shifted 180 degrees with respect to the phase-a DPWM counter.

Immediately after $i_a(t)$ is sampled, the CPU of the controller compares $i_a[n-1]$ against $i_{ref}[n-1]$ to calculate the error, $i_{ea}[n-1]$, which is then fed into a digital controller, C(z), to recalculate the new duty-ratio, $d_a[n]$. This process is illustrated in Figure 2.18 in block diagram form and the length of time that the CPU dedicates to it will be referred to as the processing delay, t_{d_proc} . t_{d_proc} is illustrated in Figure 2.19 just after the sampling instants $i_a[n-1]$. The DPWM reference-register will not be updated with $d_a[n]$ until the beginning of the next cycle, therefore $v_{ma}(t)$ will be constant for the rest of the cycle.

At the same time as the CPU handles the sampled current of phase-a, $v_{ca}(t)$ will continue increasing until it equals $v_{ma}(t)$. At this moment $v_{1a}(t)$ will be turned off until the beginning of the next switching cycle where the DPWM counter of phase-a is reset. This process is completely independent from the CPU of the digital controller as it is carried out in the DPWM module, leaving the CPU free of tasks. The phase-b process is identical but is delayed behind phase-a by half a switching cycle T/2.

The waveforms show that the leading-edge modulation strategy results in the regulation of the valley-current. Secondly, as the processing of the sampled currents of both phases is shared by the same CPU, t_{d_proc} cannot be longer than half of a switching period, in other words $t_{d_proc} \leq T/2$. Finally, since the calculated duty-ratio is held and not updated until the beginning of the next switching-cycle, a delay of a sampling period will always be present in the control response.

To overcome some of the limitations of the leading-edge modulation strategy, the following section reviews two other alternative approaches.

2.6.1. Analysis of the leading-edge, trailing-edge and triangular carrier modulations for the DIBC with IPT

Figure 2.20 depicts the principal waveforms for the three different current sampling schemes of the dual interleaved boost converter with IPT. The waveforms were sketched assuming instantaneous operation of the components, are valid for the converter operating in CCM and illustrate non-periodic waveforms to show the repercussion that slight variations on the duty-ratio have on the currents of each phase. The graphs are arranged in two groups. Figures 2.20(a), 2.20(c) and 2.20(e) are valid for $D \le 0.5$. They illustrate the main waveforms of leading-edge modulation (as in Figure 2.19), trailing-edge modulation, and triangular-carrier modulation with symmetric on-time respectively. Likewise, Figures 2.20(b), 2.20(d) and 2.20(f) depict the waveforms of these modulation schemes for D > 0.5.

Each figure is arranged in a similar way to Figure 2.19. The first graph shows the currents $i_a(t)$ and $i_b(t)$ before and after a perturbation in the phase-a duty ratio, $d_a[n]$. $d_b[n]$ is assumed to be constant during the whole illustration. In the second graph $v_{ma}(t)$ and $v_{ca}(t)$ are illustrated. $v_{mb}(t)$ and $v_{cb}(t)$ are similar to those of phase-a, but delayed by half a cycle.



The third graph depicts the sampled values of the phase-a current waveform $i_a[n]$ and the length of the processing time t_{d_proc} . Finally, the bottom plot shows $v_{ge1a}(t)$ and $v_{ge1b}(t)$.

As seen in Figures 2.20(a) and 2.20(b) for the leading-edge modulation, the DPWM counters are configured in count up mode, $v_{1a}(t)$ and $v_{1b}(t)$ are switched on when their respective DPWM counters are reset to zero, and the sampled current will be the valley-current for both D < 0.5 and D > 0.5. For the trailing-edge modulation, Figures 2.20(c) and 2.20(d), the DPWM counters are configured in count down mode, $v_{1a}(t)$ and $v_{1b}(t)$ are switched off when their respective DPWM counters are reset to Zero, and the sampled current to zero, and the sampled current will be the valley-current will be the valley-current for both D < 0.5 are configured in count down mode, $v_{1a}(t)$ and $v_{1b}(t)$ are switched off when their respective DPWM counters are reset to zero, and the sampled current will be the peak-current for both D < 0.5 and D > 0.5.

Finally Figures 2.20(e) and 2.20(f), illustrate the waveforms for the case where the DPWM counters are configured in up-down count mode. As seen in these figures, $v_{1a}(t)$ is switched on whenever $v_{ca}(t)$ is counting down and it crosses $v_{ma}(t)$. Conversely, $v_{1a}(t)$ is switched off every time $v_{ca}(t)$ is counting up and it crosses $v_{ma}(t)$. Unlike the leading-edge and the trailing-edge modulations where the sampling of $i_a(t)$ occurs at the same time as the update of $v_{ma}(t)$, in this modulation strategy these events occur at different times. $i_a(t)$ is sampled every time its DPWM counter reaches zero and $v_{ma}(t)$ is updated when its counter is at maximum count. This particular configuration has two advantages. First, the on-time of $v_{1a}(t)$ is always symmetrical with respect to the sampling instant $i_a(t)$, thereby the name triangular carrier modulation with symmetric on-time. Second, unlike the previous modulation strategies where the update of $v_{ma}(t)$ has a delay of one sampling period, T, in this modulation the update delay is reduced to T/2.

A close look at Figure 2.20(e) and 2.20(f) reveals that the sampling instants are strategically positioned in the middle of the transistors on-state intervals. This strongly suggests that if the symmetry of the phase current waveforms is preserved over a switching period, the sampled value will be close to the local average of the waveforms. Figure 2.21 will be employed to demonstrate this. In this figure, the phase-a current waveform, $i_a(t)$, and the transistor driving signals $v_{1a}(t)$ and $v_{1b}(t)$ are sketched for D < 0.5 under the assumption that the converter is in steady-state operation, and the on-state intervals of $v_{1a}(t)$ and $v_{1b}(t)$ are denoted $d_{b,n}T$ and $d_{a,n}T$ respectively, where the subindex *n* indicates the

Chapter 2

corresponding switching interval and i_{a0} is the steady state average-value of the phase current at the beginning of the switching cycle and is defined in Table 2.7.

By inspection of Figure 2.21, $i_a(t)$ may be divided into five line segments, $i_{a1}(t)$ to $i_{a5}(t)$, which are defined as shown in Table 2.6. The slopes of $i_{a1}(t)$ to $i_{a5}(t)$, namely m_1 to m_5 , are found in Table 2.7 and were calculated for each line segment using the derivative of Equation (2.3):

$$\frac{di_{a}(t)}{dt} = \frac{1}{2} \frac{di_{in}(t)}{dt} - \frac{di_{diff}(t)}{dt}$$
(2.30)

According to Figure 2.21, the sampling instant of $i_a(t)$ occurs at T/2, which is located in the interval corresponding to $i_{a3}(t)$. Therefore, in order to find the value of the sampled current, it is necessary to find $i_{a3}(T/2)$. Under the assumption that the converter is in steadystate conditions and that $d_{a,n} = d_{b,n} = d_{b,n+1} = D$, $i_{a3}(t)$ can be defined for any D < 0.5 employing the equations of Table 2.6 and Table 2.7 as shown in Equation (2.31).

$$i_{a3}(t) = m_3 \left(t - \frac{1-D}{2}T \right) + m_2 \left(\frac{1-D}{2}T - \frac{D}{2}T \right) + m_1 \left(\frac{D}{2}T + \frac{D}{2}T \right) + i_{a0}$$
(2.31)



Figure 2.21. Sampled value of the phase-a current waveform when the sampling instants are located in the middle of the transistors on-intervals. *D* < 0.5.

Function	Equation	Interval
i _{a1} (t)	$m_1\left(t+\frac{d_{b,n}T}{2}\right)+i_{a0}$	$\left(0,+\frac{d_{b,n}T}{2}\right)$
i _{a2} (t)	$m_2\left(t-\frac{d_{b,n}T}{2}\right)+i_{a1}\left(\frac{d_{b,n}T}{2}\right)$	$\left(+\frac{d_{b,n}T}{2},+\left(\frac{1}{2}-\frac{d_{a,n}}{2}\right)T\right)$
i _{a3} (t)	$m_{3}\left(t-\left(\frac{1}{2}-\frac{d_{a,n}}{2}\right)T\right)+i_{a2}\left(\left(\frac{1}{2}-\frac{d_{a,n}}{2}\right)T\right)$	$\left(\left(\frac{1}{2}-\frac{d_{a,n}}{2}\right)T,\left(\frac{1}{2}+\frac{d_{a,n}}{2}\right)T\right)$
i ₀₄ (t)	$m_4\left(t - \left(\frac{1}{2} + \frac{d_{a,n}}{2}\right)T\right) + i_{a3}\left(\left(\frac{1}{2} + \frac{d_{a,n}}{2}\right)T\right)$	$\left(\left(\frac{1}{2}+\frac{d_{a,n}}{2}\right)T,\left(1-\frac{d_{b,n+1}}{2}\right)T\right)$
i _{a5} (t)	$m_{5}\left(t-\left(1-\frac{d_{b,n+1}}{2}\right)T\right)+i_{a4}\left(\left(1-\frac{d_{b,n+1}}{2}\right)T\right)$	$\left(\left(1-\frac{d_{b,n+1}}{2}\right)T,T\right)$

Table 2.6. Equations of the line segments comprising $i_a(t)$ valid for D < 0.5 (Figure 2.19).

Table 2.7. 9	Slopes of the line	segments com	prising <i>i_a(t)</i> .

Slope	Equation D < 0.5	Equation D > 0.5	
i _{a0} *	$I_{a} - \frac{\Delta I_{smll}}{2} = \frac{I_{in}}{2} - \frac{\frac{V_{in} - V_{o}}{2}}{4L_{in}} DT + \frac{V_{o}}{2L_{diff}} DT$	$I_{a} - \frac{\Delta I_{lrg}}{2} = \frac{I_{in}}{2} - \frac{V_{in} - \frac{V_{o}}{2}}{4L_{in}} DT - \frac{V_{o}}{2L_{diff}} DT$	
<i>m</i> 1	$\frac{V_{in} - V_o/2}{2L_{in}} - \frac{V_o}{L_{diff}}$	$\frac{\frac{V_{in} - \frac{V_o}{2}}{2L_{in}} + \frac{V_o}{L_{diff}}}{\frac{V_o}{L_{diff}}}$	
<i>m</i> ₂	$\frac{V_{in} - V_o}{2L_{in}}$	$\frac{V_{in}}{2L_{in}}$	
m3	$\frac{V_{in} - V_o/2}{2L_{in}} + \frac{V_o}{L_{diff}}$	$\frac{V_{in} - V_o/2}{2L_{in}} - \frac{V_o}{L_{diff}}$	
m4	m ₂	<i>m</i> ₂	
m_5	m_1	m_1	
* ΔI_{smll} and ΔI_{Irg} as defined in Table 2.1			

Equation (2.31) can be simplified, yielding:

$$i_{a3}(t) = m_3 \left(t - \frac{1 - D}{2} T \right) + m_2 \left(\frac{T}{2} - DT \right) + m_1 DT + i_{a0}$$
(2.32)

Substituting i_{a0} from Table 2.6 and evaluating Equation (2.32) at T/2 yields:
Converter operation, modelling and digital control

$$i_{a3}(\frac{T}{2}) = \frac{1}{2} \left(\frac{V_{in} - V_{o}}{2L_{in}} + \frac{V_{o}}{L_{diff}} \right) DT + \frac{1}{2} \left(\frac{V_{in} - V_{o}}{2L_{in}} \right) T - \left(\frac{V_{in} - V_{o}}{2L_{in}} \right) DT + \frac{V_{in} - V_{o}}{2L_{in}} - \frac{V_{o}}{L_{diff}} DT + I_{a} - \left(\frac{V_{in} - V_{o}}{4L_{in}} - \frac{V_{o}}{2L_{diff}} \right) DT + \frac{V_{o}}{2L_{diff}} DT$$
(2.33)

Simplifying Equation (2.33), results in:

$$i_{\sigma 3}\left(\frac{T}{2}\right) = \frac{1}{2}\left(\frac{V_{in} - V_{o}}{2L_{in}}\right)T + \left(\frac{V_{o}}{4L_{in}}\right)DT + I_{o}$$
(2.34)

Equation (2.34) can be further simplified using the conversion ratio of the converter, Equation (2.1), resulting in:

`

$$i_{a3}\left(\frac{T}{2}\right) = -\frac{V_{in}DT}{4L_{in}(1-D)} + \frac{V_{in}DT}{4L_{in}(1-D)} + I_a = I_a$$
(2.35)

Equation (2.35) demonstrates that sampling in the middle of the transistors' onstate intervals when the triangular carrier modulation is employed, will acquire the average-



Figure 2.22. Sampled value of the phase-a current waveform when the sampling instants are located in the middle of the transistors on-intervals. D > 0.5.

value of the phase-current. A similar procedure can be undertaken for D > 0.5 using Figure 2.22 and the corresponding equations from Table 2.7 and 2.8, leading to the same conclusion.

Function	Equation	Interval
i _{a1} (t)	$m_1\left(t+\frac{d'_{a,n}T}{2}\right)+i_{a0}$	$\left(0,+\frac{d'_{a,n}T}{2}\right)$
i _{a2} (t)	$m_2\left(t-\frac{d'_{a,n}T}{2}\right)+i_{a1}\left(\frac{d'_{a,n}T}{2}\right)$	$\left(+\frac{d'_{a,n}T}{2},+\left(\frac{1}{2}-\frac{d'_{b,n}}{2}\right)T\right)$
i _{a3} (t)	$m_{3}\left(t-\left(\frac{1}{2}-\frac{d'_{b,n}}{2}\right)T\right)+i_{a2}\left(\left(\frac{1}{2}-\frac{d'_{b,n}}{2}\right)T\right)$	$\left(\left(\frac{1}{2}-\frac{d'_{b,n}}{2}\right)T,\left(\frac{1}{2}+\frac{d'_{b,n}}{2}\right)T\right)$
i _{a4} (t)	$m_4\left(t-\left(\frac{1}{2}+\frac{d'_{b,n}}{2}\right)T\right)+i_{a3}\left(\left(\frac{1}{2}+\frac{d'_{b,n}}{2}\right)T\right)$	$\left(\left(\frac{1}{2}+\frac{d'_{b,n}}{2}\right)T,\left(1-\frac{d'_{a,n+1}}{2}\right)T\right)$
i _{a5} (t)	$m_{5}\left(t-\left(1-\frac{d'_{a,n+1}}{2}\right)T\right)+i_{a4}\left(\left(1-\frac{d'_{a,n+1}}{2}\right)T\right)$	$\left(\left(1-\frac{d'_{a,n+1}}{2}\right)T,T\right)$

Table 2.8. Equations of the line segments comprising $i_a(t)$ valid for D < 0.5 (Figure 2.20).

2.7. Summary

Starting with a review of the steady-state waveforms for the dual interleaved boost converter with IPT, an averaged small-signal model was derived by substitution of the converter switch networks by the averaged PWM switch model. In contrast with previous models, the small-signal model accounts for the small-ac perturbation of each control input independently. This characteristic is exploited in Chapter 4 where the model is further developed to predict the stability issues that arise from the interleaved interaction of the converter phases.

The accuracy of the small signal model has been established through detailed comparison with both a switched and large signal averaged simulations using SABER software. Excellent agreement was found for duty-ratio disturbances of up to 5 %, except at very high duty-ratio values (D = 0.8), where noticeable discrepancies occurred, and this was attributed to the assumptions of the linearization analysis becoming invalid. However,

with duty-ratio disturbances of up to 5 % the error in the small signal model remained less than approximately 10 %.

By analysing the main digital current-control strategies that employ uniformlysampled pulse-width modulation, the sampling strategies of the trailing and the leading edge modulations were seen to result in the valley and the peak current of the converter being sampled. Finally, the triangular carrier modulation with symmetric on-time was shown to result in the sampled value of the phase currents being equal to the average value of the phase current waveforms when the system is in steady-state conditions. This approach is therefore well-suited to the control of the dual interleaved converter, providing a straight-forward method of ensuring current-balance between two phases.

2.8. References

- [1] G. Calderon-Lopez, A. J. Forsyth, and D. R. Nuttall, "Design and Performance Evaluation of a 10-kW Interleaved Boost Converter for a Fuel Cell Electric Vehicle," in *Power Electronics and Motion Control Conference, 2006. IPEMC 2006. CES/IEEE 5th International,* 2006, pp. 1-5.
- [2] G. Calderon-Lopez and A. J. Forsyth, "High-Power Dual-Interleaved ZVS Boost Converter with Interphase Transformer for Electric Vehicles," in *Applied Power Electronics Conference and Exposition, 2009. APEC 2009. Twenty-Fourth Annual IEEE*, 2009, pp. 1078-1083.
- [3] R. W. Erickson and D. Maksimovic, Fundamentals of Power Electronics. USA: Kluwer Academic Publishers, 2004, Chap. 7, p.p. 238-239.
- [4] V. Vorperian, "Simplified analysis of PWM converters using model of PWM switch. Continuous conduction mode," *Aerospace and Electronic Systems, IEEE Transactions on,* vol. 26, pp. 490-496, 1990.
- [5] D. Maksimovic, R. W. Erickson, and C. Griesbach, "Modeling of cross-regulation in converters containing coupled inductors," *Power Electronics, IEEE Transactions on*, vol. 15, pp. 607-615, 2000.
- [6] H. Xudong, T. Nergaard, L. Jih-Sheng, X. Xingyi, and L. Zhu, "A DSP based controller for high-power interleaved boost converters," in Applied Power Electronics Conference and Exposition, 2003. APEC '03. Eighteenth Annual IEEE, 2003, pp. 327-333 vol.1.
- [7] H. Xudong, W. Xiaoyan, T. Nergaard, L. Jih-Sheng, X. Xingyi, and L. Zhu, "Parasitic ringing and design issues of digitally controlled high power interleaved boost converters," *Power Electronics, IEEE Transactions on*, vol. 19, pp. 1341-1352, 2004.
- [8] S.S. Hsu, A. Brown, L. Rensink, and R. D. Middlebrook, "Modeling and analysis of switching dc-to-dc converters in constant-frequency current programmed mode," in Proceedings of IEEE Power Electronics Specialists Conference, 1979, PESC'79, pp. 284-301, 1979.
- [9] S. Chattopadhyay and S. Das, "A digital current mode control technique for DC-DC converters," *in Applied Power Electronics Conference and Exposition, 2005. APEC 2005. Twentieth Annual IEEE*, 2005, pp. 885-891 Vol. 2.
- [10] S. Buso and P. Mattavavelli, Digital Control in Power Electronics. USA: Morgan & Claypool, 2006.

Chapter 3 Digital control platform and algorithm implementation

3.1. Introduction

In the previous Chapter, three digital control methodologies based on the uniformlysampled pulse-width modulation were discussed for the dual-interleaved boost converter: valley-current, peak-current and average-current control. To investigate the practical advantages and disadvantages of these control strategies, a control platform containing a digital signal processor and all the necessary components to interface it with a prototype dual-interleaved converter was developed. In this Chapter the construction of this control platform, the implementation of the control algorithms and the practical and simulation results obtained are presented and discussed.

The first and second sections discuss the design, component selection and construction details of the digital control platform. The third section covers the implementation of the valley-current, peak-current and average-current control algorithms in the Digital Signal Controller, DSC. Also, in this section a variation of the average-current control strategy is proposed and implemented. The fourth section presents a short discussion of the SABER simulation models developed to investigate the behaviour of the control strategies. Finally, the last two sections present a discussion of the experimental and simulation results obtained with the control strategies and demonstrate the validity of the simulation models.

3.2. Prototype of the dual-interleaved boost converter with inter-phase transformer

Figure 3.1 shows a converter circuit diagram with the control platform, which indicates the input/output signals that will drive the converter under current control mode. The DIBC prototype is a 25 kW high-density bi-directional version based on the requirements of a small fuel-cell based vehicle power train. This converter has been developed within the Power Conversion Group by Dr Gerardo Calderon-Lopez as part of a separate research project. The prototype has an input voltage range of $V_{in} = 75 V to 150 V$; an output voltage range, $V_o = 220 V to 275 V$; an individual-phase switching frequency, $f_s = 30 \text{ kHz}$; and it has a maximum power capability, $P_o = 25 \text{ kW}$.



Figure 3.1. DIBC topology and control platform.

The input inductor was designed to give an overall input ripple current of approximately 114 A peak-to-peak under worst-case conditions (Vin = 75 V, Vo = 275 V), whilst the IPT was designed to have a maximum peak-to-peak differential current of approximately 8 A under worst case conditions ($D \le 0.5$). The output capacitance was calculated to obtain a maximum output voltage ripple of 2 % at maximum voltage conditions. In order to reduce the parasitic inductance of the switching paths and prevent

large over-voltage transients, the components and devices are interconnected using planar bus bars. For the thermal management, the half-bridge semiconductor modules and magnetic component cases are mounted onto a liquid-cooled heat sink. Table 3.1 presents a summary of the prototype details whilst a picture of the converter is shown in Figure 3.2.

Operation	n Ratings	Device Specifications		
Maximum power, Po	25 kW	Input capacitance, C_{in} 22.4 μ F		
Input voltage range, V _{in}	75 V – 150 V	Output capacitance, C _o	45 μF	
Output voltage range, Vo	220 V – 275 V	Input inductance, L _{in}	5.12 μH	
Operating frequency <i>f</i>	30 kHz	Inter-Phase Transformer		
Maximum input current, I _{Lin} *	330 A	Winding self- inductance, L _a , L _b	75.14 μH	
Maximum output current, <i>I_o*</i>	114 A	Mutual inductance, <i>L_m</i>	74.9 µH	
		Differential inductance, L _{diff}	299.7 μH	
		Coupling coefficient, k	0.997	
		Active Devices		
		IGBT Module	CM300DU-21NFH	

Table 3.1. Parameters of the DIBC prototype.

* Calculated assuming operation at maximum power.



Figure 3.2. Prototype of the dual-interleaved boost converter with inter-phase transformer.

3.3. Digital control platform prototype

The control platform was created as an interface between the DSC and the converter prototype. This platform consists of several stages which are represented in the block diagram form in Figure 3.3. The functionality of these stages will be described next. The block located at the top right of the diagram represents the interleaved converter. The transistors of the converter are driven by four gate-to-emitter signals which are represented as inputs to the functional block: v_{ge1a} , v_{ge1b} , v_{ge2a} and v_{ge2b} ; these signals range from -5 V to +15 V. The converter output voltage, v_o , load current, i_{load} , and phase currents, i_a and i_b respectively, are represented as output signals. The range of i_a and i_b was determined under the assumption that the input current, i_{Lin} , divides equally between the phases. Employing the maximum value of i_{Lin} from Table 3.1, the maximum value of i_a and i_b was determined to be approximately 175 A.



Figure 3.3. Functional block diagram of the digital control platform.

Two hall-effect current sensors, as depicted by the block diagram, were employed to measure the phase-currents. These transducers generate two current signals i_{a_sens} and

 i_{b_sens} proportional to i_a and i_b respectively. The generated signals have a conversion ratio, K_N , of 1:2000 and range between 0 mA to 100 mA. These current signals representing the phase currents are fed into a conditioning stage where they are converted into voltage signals ready to be read by the ADC. These signals are denoted v_{ia} and v_{ib} in the diagram. Additionally a third signal designated v_{iref} is generated in this stage. This signal allows the user to set the target or reference phase-current for the converter by means of a potentiometer.

The control card, which is the next block in the diagram, is divided into three main parts: the ADC module which samples the signals v_{ia} , v_{ib} and v_{iref} ; the PWM module which generates the driving signals v_{1a} , v_{1b} , v_{2a} and v_{2b} ; and the General Purpose Input/Outputs, GPIOs, which are used to generate the control signals *mode*_a and *mode*_b and to read the fault signals *fault*_{oc-a} and *fault*_{oc-b}. All the signals generated from the DSC range from 0 to 3 V. The driving signals v_{1a} , v_{1b} , v_{2a} and v_{2b} are passed through a buffer/conditioning stage, as depicted in the diagram, where their voltage level is adjusted to be compatible with the gate drivers. Finally the gate drivers, generate the isolated gate-emitter driving signals v_{ge1a} , v_{ge1b} , v_{ge2a} and v_{ge2b} for the converter transistors. Additionally these devices are capable of two modes of operation: half-bridge mode or direct mode. In half-bridge mode, v_{ge1a} follows v_{1a} and v_{ge1b} follows the complement of v_{1a} ; v_{1b} is used to enable/disable the gatedriver outputs. In direct mode, v_{ge1a} and v_{g1b} follow the driving signals v_{1a} and v_{1b} . The mode of operation is selected by the *mode*_a and *mode*_b signals generated in the GPIOs of the DSC.

Two over-current fault signals, $fault_{oc-a}$ and $fault_{oc-b}$, are set by the gate drivers in the event of a transistor over-current. These signals are latched and stop the PWM generation in the processor until the user clears the fault manually using a push-button provided in the control platform, $clear_fault_{oc}$. The board is powered by a single external power supply of +15 V and incorporates additional dc-dc converters to generate +12 V, -12 V, +5 V and +3.3 V. Having established the functionality of all the stages in the control platform, the following sections will address the device selection and design for each.

3.3.1. Digital Signal Controller (DSC)

The control platform developed in this project was designed to interface a Delfino TMS320F28335 digital signal controller from Texas Instruments with the converter prototype [1]. This floating-point, digital signal controller was chosen as it was one of the fastest processors available at that time in the market. The processor is provided in an industry-standard DIMM-form-factor control card, shown in Figure 3.4, which is a complete board-level module incorporating all the necessary components to power up and code the processor using a standard 100-pin DIMM interface.



Figure 3.4. Delfino TMS320F28335 Control Card.

CPU	C28x	
Million Multiply Accumulate Cycles per Second, MMACS	150 MAC	
FPU	Yes	
Master Clock Frequency	150 MHz	
RAM	68 Kb	
Flash	512 kB	
DMA	1 x 6-Channels DMA	
PWM	9 Modules, 18 Outputs	
Analog to Digital Converter, ADC	1 x 16-Channels 12-Bit Pipelined	
ADC Conversion Time	80 ns	
Inter-Integrated Circuit protocol, I2C	1	
Universal Asynchronous Receiver Transmitter, UART (SCI)	3	
Serial Port Interface, SPI	1	
CAN,	2	
Timers	3 32-Bit CPU,1 WD	
General Port Input Output, GPIO	88	

 Table 3.2. Principal Features of the Texas Instruments TMS320F28335 [2].

The processor incorporates a 12-bit analogue-to-digital converter, ADC, with a minimum conversion time of *80 ns* per sample and is capable of sampling up to 16 multiplexed channels. It also contains nine digital pulse width modulation modules with a

minimum resolution of 6.66 ns. The maximum processor clock frequency is f_{clk} = 150 MHz. Table 3.2 provides a summary of the processor's main characteristics [2].

3.3.2. Current sensors and conditioning for the measured signals

Figure 3.5 shows a detailed diagram of the current sensors and associated conditioning stage. The current transducers employed to measure the phase currents in the converter were the *LA 200-P* manufactured by LEM which have a conversion ratio, K_N , of 1:2000 [3]. A measurement resistor, R_m , of 10 Ω is connected to the output of the current transducer to generate 5 *mV per Ampere* of current measured. The voltage generated across R_m is then scaled using an operational amplifier configured as a level shifter. The gain of the amplifier is set by the resistors R_a , R_b , R_c and R_d as shown in Equation (3.1):

$$A_{AV} = \frac{R_B}{R_A} \tag{3.1}$$

where $R_A = R_a = R_c$ and $R_B = R_b = R_d$. The level-shifting function provided by the voltage V_{ref} connected to R_d enables the ADC, which only accepts positive inputs, to read both positive and negative phase currents. The gain of the level shifter, A_{AV} , was set to 1.5 and the zero-current reference on the level-shifter was established as 1.5 V. This allowed the measurement of phase-currents ranging from -200 A to +200 A using a voltage range from 0 V to 3 V at the ADC input. An OPA2343UA amplifier was used in the level shifter [4]. This is a rail-to-rail amplifier specially designed to drive A/D converters.



Figure 3.5. Current sensor and signal conditioning for the ADC.

Finally, the voltage delivered to the ADC per Ampere of current measured can be calculated using Equation (3.2):

$$\mathbf{v}_{ix}(t) = \left(\frac{i_x(t)}{K_N}R_m\right)\frac{R_B}{R_A} + 1.5 = 1.5\left(\frac{i_x(t)}{200} + 1\right)$$
(3.2)

The full schematic diagram of the conditioning stage of the control platform can be found in Figure F.1, Appendix F.

3.3.3. Signal conditioning for transistor driving signals and gate drivers

The following paragraphs will describe how the gate drivers were used in combination with the DSC. For this purpose Figure 3.6 shows a simplified functional block diagram of this stage. A detailed diagram showing all the components employed in the design can be found in Figure F.3, Appendix F.

The gate drivers employed were the 2SC0435T manufactured by CONCEPT [5, 6]. These circuits provide a dual channel IGBT/MOSFET driver capability which is fully isolated and has short-circuit protection through v_{ce} monitoring.



Figure 3.6. Functional block diagram of the signal conditioning for the transistor driving signals and the gate drivers.

According to Figure 3.6, the transistor driving-signals produced by the PWM module on the controller, v_{1a} and v_{2a} respectively, are taken to an array of AND logic gates. This array provides a way to interrupt the propagation of v_{1a} and v_{2a} in the event of an overcurrent failure and notifies the processor of the fault by means of the signal Tz_a . An UCC37324 MOSFET-driver was used to interface the TTL logic signals with the gate drivers which requires 15 V input. The over-current signals, $fault_{oc-a}$ and $fault_{oc-b}$, are pulled down when a fault occurs and held in this state by a latch circuit. The fault must be cleared manually using an external push button to re-enable operation of the system.

Additionally, an external N-Channel MOSFET driven directly by the DSC is connected in parallel to a mode selector resistor, R_{mod} . This arrangement enables the DSC to control the mode of operation of the gate drivers. When this MOSFET is turned off, the gate drivers operate in bridge mode, when is turned on they operate in direct mode. Finally the block shown at the right hand side of the CONCEPT driver denoted as interface circuitry includes the IGBT gate drive resistors.

3.3.3. Construction Details

A double-sided PCB with ground planes was created to accommodate the control platform circuitry. The complete schematic diagrams of this platform are found in Appendix F. The layout is shown in Figure 3.7 and was produced using Altium Designer. The control card is located on the top layer whilst the gate drivers are located on the bottom layer. Figure 3.8 shows a picture of the control platform.

Digital control platform and algorithm implementation



Figure 3.7. Printed circuit board of the digital control platform. (a) Top layer; (b) Bottom layer.



Figure 3.8. Digital control platform.

3.4. Implementation of the digital current-control algorithms on the DSC

The DSC was programmed using MATLAB Simulink and MATLAB Embedded Coder to create a block diagram which was compiled into C code and then into assembly files using Code Composer Studio. The assembly files were downloaded into the DSC flash memory by means of a docking station for the control card [7]. The Simulink diagrams generated can be found in Appendix D.

In the following sections the implementation of the valley-current, peak-current and average-current control strategies described in Section 2.6, Chapter 2, are discussed. Also, a variation of the average-current control strategy is investigated, which has not been reported before. In this variation the DPWM modules are updated immediately after the calculation of the duty-ratio, thereby enabling a reduction in the processing delay of the system. The algorithms for these control strategies share the same foundation. The only difference between them is the configuration of the DPWM modules and the location of the current-sampling instants which can be easily modified in the Simulink diagram.

3.4.1. Configuration and synchronization of the DSC modules

In this section, the configuration of the DSC is explained using the diagram of Figure 3.9 and the waveforms of Figure 3.10(a) and 3.10(b). In Figure 3.9, the functionality of the modules employed in the DSC and their inter-connections are represented using a block diagram. This diagram consists of four blocks: the analog-to-digital converter module, *ADC*, the ADC Interrupt Service Routine, *ADC_ISR*, and the digital pulse-width modulators, *DPWM A* and *DPWM B*.

The *ADC* was programmed to sample three signals in the following sequence: the phase-a current, v_{ia} , first, the phase-b current, v_{ib} , second, and the reference-current, v_{iref} , third. The sampled-data is stored in separate registers. The contents of these registers are labelled in Figure 3.9 as $v_{ia}[n]$, $v_{ib}[n]$ and $v_{iref}[n]$ respectively and can be calculated using Equation (3.3):

$$v_{ix}[n] = 2^{N_{ADC}} \frac{v_{ix}(t)}{V_{ADC\,\text{max}}} = (2^{12}) \frac{v_{ix}(t)}{3}$$
(3.3)

where N_{ADC} is the effective resolution of the ADC in bits, V_{ADCmax} is the maximum reference voltage of the ADC and v_{ix} is the sampled voltage in the corresponding channel. At the end of the sampling sequence, the ADC sets a flag to signal to the CPU the end of conversion. This flag will be referred to as EOC.



Figure 3.9. Functional block diagram of the DSC modules.

The *ADC_ISR* was programmed to be serviced by the CPU when the EOC flag is detected. This routine executes the control algorithm for the converter phases. A detailed discussion of the algorithms implemented in this service-routine is undertaken in the next section. To summarize, the service routine employs the sampled-data acquired by the *ADC* to calculate the duty-ratio of the converter transistors, $d_a[n]$ and $d_b[n]$.

As determined in Chapter 2, the binary-counters of the DPWM modules were programmed in count up mode to achieve valley-current control, in count down mode to achieve peak-current control, and count up-down mode to achieve average-current control. The strategic sampling for these control schemes was attained by programming the DPWM modules to initiate an ADC conversion sequence whenever their binary-counters reached zero.

Also, the reference-registers of the DPWM modules were programmed to be updated at the maximum count of the binary-counters. Finally, to achieve the generation of the interleaved PWM waveforms for the converter transistors, the binary-counters of these modules were synchronized and phase-shifted by 180 degrees.

The maximum count of the binary counters was determined using Equation (3.4) for the count up and the count down mode, and using Equation (3.5) for the count up-down mode:

$$v_{cmax} = \frac{f_{clk_DPWM}}{f_{sw}} = \frac{150MHz}{30kHz} = 5000$$
 (3.4)

$$v_{cmax} = \frac{1}{2} \left(\frac{f_{clk_DPWM}}{f_{sw}} \right) = 2500$$
(3.5)

where f_{clk_DPWM} is the frequency of the DPWM clock, and f_{sw} is the switching frequency of the converter transistors. In the block diagram of Figure 3.9, the signal that initiates the ADC conversion is denoted as the start-of-conversion flag, SOC, and it interconnects the DPWM modules with the ADC.

The waveforms depicted in Figure 3.10(a) and 3.10(b) will be employed to discuss the synchronization of the blocks of Figure 3.9 when the DPWM counters are configured in count up mode and count up-down mode. The count down mode waveforms are not illustrated as they can be easily deduced from the ones in Figure 3.10(a). In the first and second plots of these figures the *DPWM A* and *DPWM B* counters are sketched respectively. The third and fifth plots illustrate the timing diagrams of the Start-of-Conversion flag, SOC, and the End-of-Conversion flag, EOC. Finally, the fourth and sixth graphs illustrate the timing diagrams corresponding to the ADC and CPU tasks respectively.

Initially in Figure 3.10(a), the *DPWM A* counter is reset to zero. At this instant the SOC flag is set by the *DPWM A* initiating an ADC conversion-sequence. The duration of the ADC conversion is depicted in the ADC timeline. Once the ADC conversion sequence is finished, the EOC flag is set by the ADC which requests the execution of the *ADC_ISR*. The *ADC_ISR* is then serviced by the CPU, which calculates the new duty-ratio $d_a[n]$. The time duration of *ADC_ISR* is illustrated in the CPU timeline. After $d_a[n]$ has been loaded into the

DPWM A registers, the CPU remains idle until the *DPWM B* counter is reset to zero. At this point, a new ADC conversion-sequence is requested and the routine is executed for phaseb. The synchronization of the DSC modules for the count up-down mode of operation, Figure 3.10(b), follows the same principle as described for the leading-edge modulation.



Figure 3.10. (a) Synchronization of the DSC modules for leading-edge modulation (valley-current control); (b) Synchronization of the DSC modules for triangular carrier modulation with symmetric on-time (average-current control).

3.4.2. ADC interrupt service routine, ADC_ISR

The scaling and control-loop compensation algorithm is executed during the ADC interrupt service routine. A flowchart of this algorithm is depicted in Figure 3.11(a), and it will be referred to as the normal-update control algorithm. The first task executed by the algorithm is to read the sampled-data acquired by the ADC module: $v_{ia}[n]$, $v_{ib}[n]$ and $v_{iref}[n]$. Followed by this, it determines which DPWM module triggered the ADC conversion and decides which phase has to be updated in the current cycle. Once the phase-in-turn has been identified, the routine carries out the scaling and compensation calculation which determines the new duty-ratio, $d_a[n]$ or $d_b[n]$. Finally, the last task of the control algorithm

is to load the new duty-ratio into the registers of its corresponding DPWM module and return to the main program.

The scaling and compensation process carried out by the control algorithm is illustrated in the block diagram of Figure 3.11(b), and it has identical structure for both phases. In this diagram, the data read from the ADC registers, $v_{iref}[n]$ and $v_{ix}[n]$ is first transformed back into Amperes. This is accomplished using Equation (3.6), which is found by substituting Equation (3.2) into Equation (3.3) and solving for i_x :

$$i_{x}[n] = \frac{K_{N}R_{A}}{R_{m}R_{B}} \left(\frac{V_{ADCmax}}{2^{ADCbits}} v_{ix}[n] - 1.5 \right) = (97.656e^{-3}) v_{ix}[n] - 200$$
(3.6)

The phase-current and the reference-current, $i_x[n]$ and $i_{ref}[n]$, are then compared in order to obtain the error-current, $i_{ex}[x]$, which is passed on to a digital compensation network that calculates the new duty-ratio, $d_x[n]$. Finally, $d_x[n]$ is multiplied by the modulator gain, K_{PWM} , and is loaded into the DPWM module.



Figure 3.11. Valley-current, peak-current and average-current control algorithm. (a) Flow chart of the algorithm executed during the ADC interrupt service routine. (b) Scaling and compensation of the control-loops. (c) Compensation network: Backward-Euler proportional–integral compensator in parallel form.

In this thesis, a proportional-integral compensator implemented using the backward Euler integration method is adopted as the compensation network. The transfer function of this digital compensator is as shown in Equation (3.7), and its block diagram representation in parallel form is shown in Figure 3.10(c).

$$\frac{Y(z)}{R(z)} = K_{p} + K_{i}T \frac{1}{1 - z^{-1}} = \frac{\left(K_{p} - K_{i}T\right)z - K_{p}}{z - 1}$$
(3.7)

3.4.3. Detailed timing of the control algorithms

In Section 3.4.2 the synchronization of the DSC modules was discussed assuming the instantaneous switching of the converter transistors and neglecting the propagation delay caused by the conditioning stages of the control platform and gate-drivers.

In this section, the sampling and the control algorithm timing is analysed together with the transistor switching waveforms to determine the effects of the non-zero switching times and the propagation delays in the performance of the control algorithms. For this purpose, Figure 3.12, shows a magnification of the turn-on instant of the Q_{1a} transistor when the leading-edge modulation (valley-current control) is employed. A similar analysis can be carried out for the trailing-edge modulation (peak-current control) algorithm during the turn-off instant of Q_{1a} . The first plot of Figure 3.12 shows the transistor driving signal, v_{1a} , which is generated at the output of the DPWM modules. The second plot illustrates the gate-to-emitter voltage, v_{ge1a} . The third plot shows the collector current i_{c1a} . The fourth plot illustrates the phase-a current, i_a . The fifth plot illustrates the timing of the ADC sampling-sequence. The sixth plot shows the timing of the conversion and encoding process of the ADC module. Finally, the last plot illustrates the timing of the CPU tasks that are involved in the execution of the control algorithms.

Initially in Figure 3.12, v_{1a} is set to a high value by its corresponding *DPWM* module. The time delay from the moment where v_{1a} is set high to the moment where the phasecurrent is fully conducted by the transistor Q_{1a} is denoted t_{dQ} . In the figure, t_{on} is defined as the sum of the gate-driver propagation delay, $t_{p_{gd}}$, the transistor turn-on delay, $t_{d(on)}$, and the transistor current rise-time, $t_{r(on)}$:



Figure 3.12. Timing diagram of the turn-on instant of the Q_{1a} transistor and the DSC modules for leading edge modulation or valley-current control.

$$t_{dQ} = t_{p_{gd}} + t_{d(on)} + t_{r(on)}$$
(3.8)

At the same time as v_{1a} is set high, an ADC conversion-sequence is requested. The time delay from the moment that this request takes place to the start of the sampling-sequence is denoted t_d . During the sampling-sequence, each signal is sampled and coded with a uniform sampling period T_{SH} and an acquisition window of t_{SH} . The latency for the results of each conversion to appear in the registers of the ADC is denoted $t_{d(sch)}$. The total time taken by the ADC to complete the conversion-sequence is defined as the ADC conversion time, t_{ADC} , and may be calculated from Figure 3.12 as:

Digital control platform and algorithm implementation

$$t_{ADC} = t_d + 2.5T_{SH} + t_{d(sch)}$$
(3.9)

Once the last result of the conversion is loaded into the result registers, the ADC module requests an interruption to the CPU which takes the necessary steps to run the ADC service routine. During this time, denoted $t_{INT SRVC}$, the processor receives and determines the source of the interrupt request, approves the interrupt, executes context saving and prepares for the interrupt service routine [8]. The ADC interrupt service routine has a duration of $t_{ADC INT}$ seconds. During this time the new duty ratio of the corresponding phase is determined and loaded in the DPWM register. The duration of $t_{ADC INT}$ depends directly on the complexity of the controller compensation calculation. The length of time from the request of the ADC conversion sequence to the moment where the new duty-ratio is loaded in the DPWM registers is defined as the processing time delay, t_{d_proc} . After the ADC interrupt is finished, the CPU carries out the necessary steps to exit the service routine and to return to the main program. The length of time spent on these tasks is equal to t_{INT} SRVC [8]. A list of the values of the times identified in Figure 3.12 can be found in Table 3.3.

Parameter	Value		
$t_{p_{gd}}*$	85 ns		
$t_{d(on)}/t_{d(off)}$ **	350 ns/700 ns		
$t_{r(on)}/t_{f(off)}$ **	150 ns		
t _{dQ}	585 ns		
<i>td</i> ***	100 ns		
Т _{SH} ***	80 ns		
t _{sH} ***	40 ns		
t _{d(sch)} ***	160 ns		
t _{ADC}	460 ns		
t _{INT_SRVC} ****	Not specified		
•	Length depends on the complexity of the		
LADC_INT	operations		
$t_{d_proc} = t_{ADC} + t_{INT} s_{RVC} + t_{ADC} i_{NT}$	460 ns + t_{INT} srvc + t_{ADC} int		

Table 3.3. Value of the timings shown in Figure 3.11.

 $t_{d_proc} = t_{ADC} + t_{INT_SRVC} + t_{ADC_INT}$

* As indicated in the gate-driver datasheet [5]. ** As indicated in the CM300DU-12NFH IGBT datasheet [9]. *** As indicated in the TMS320F28335 manual [2]. **** [2,8]

From Figure 3.12, it is evident that the sampling of the phase currents will not occur at the location of the valley-current due to the delayed turn on of the transistor Q_{1a} causing a constant error which will not be detected by the control-loops. The magnitude of the error is variable and depends on the slope of the current waveform.

3.4.4. Average-current control with immediate DPWM update

In this variation of the average-current control, the configuration of the DPWM modules was modified to provide immediate access to the contents of the registers that contain the value of the duty-ratio in the DPWM modules. Employing this modification a new duty-ratio can be loaded immediately into the reference-register, eliminating the delay of the control loop caused by the DPWM module. However, the use of this mode of operation may generate spurious PWM waveforms under certain conditions.

For instance, consider the particular case illustrated in Figure 3.13(b) where the value of the duty-ratio calculated by the control algorithm in the n^{th} -cycle is smaller than the actual value of the DPWM counter. Under these circumstances, the counter will not cross over the value in the reference-register during the first half of the cycle. As a consequence v_{1a} will not be set-low causing an over-current fault. A simple way to prevent this issue is illustrated in the waveforms of Figure 3.13(c). Here, the minimum value of the duty-ratio is constrained to values that are higher than the value of the counter at the update instant. From Figure 3.13(c) it is evident that the value for the minimum constraint is directly related to the processing time of the control algorithm and the switching period of the converter, as shown by Equation (3.8):

$$D_{\min} = 2 \frac{t_{d_{proc}}}{T_{sw}} = \frac{4.2\,\mu s}{33.33\,\mu s} = 0.252 \approx 25\%$$
(3.8)

However, it is clear that this solution imposes a limit in the minimum range of operation that the converter can achieve, making it less attractive for applications where the converter is required to operate over the full-range of the duty-ratio, when the ratio of the processing time to the switching period is not small enough to meet the application requirements, or when a large number of converter phases is used.

Also, after testing this new configuration in conjunction with the control algorithm introduced in Section 3.4.3, an unexpected over-current fault was detected for duty-ratios

around 79 %. A close-look at the fault-instants revealed that the transistor control-signals were being latched-on for a couple of switching cycles. Further analysis showed that these fault events occurred when a new duty-ratio was loaded into the DPWM modules at the same time instant that the other phase was changing state.



Figure 3.13. Average-current control with immediate DPWM update. (a) Control algorithm executed during the ADC interrupt service routine. (b) Time diagram illustrating the generation of spurious waveforms and how to avoid this issue by constraining the minimum value of the duty-ratio.

To overcome this issue, an "anti-glitch" stage was added to the control algorithm of Figure 3.11(a). The flowchart of this algorithm is illustrated in Figure 3.13(a). Compared with the normal-update version of the average-current control algorithm, this version includes a stage that prevents the DPWM reference register of one phase being updated at the same time as the opposite phase is being turned off. After performing this modification

the immediately updated PWM mode algorithm was able to operate correctly as demonstrated experimentally in Section 3.7.

3.5. SABER simulation models

Three simulation models were developed in SABER to investigate the behaviour of the DIBC with IPT in conjunction with the valley-, peak- and average-current control methodologies. These simulation models are based on the SABER switched model of the converter introduced in Chapter 2, Sections 2.4.2. To achieve an accurate simulation, the functionality of the modules employed in the DSC was rigorously replicated employing the functional blocks provided by SABER and also by programming special blocks specific for this application. The advantage of these models is that they accurately predict the behaviour of the control algorithms including the quantization effects of both the ADC and the DPWM modules, but at the cost of long simulation times, for instance a *100 ms* simulation will take about *40 min* to be completed. The SABER schematics of these models are depicted in Figures C.1 to C.6, Appendix C.

A fourth simulation model was specifically developed for the average-current control methodology. This simulation model makes use of the SABER average model of the converter developed in Chapter 2, Section 2.4.3. The control loops implemented in this model do not account for the quantization effects of the ADC and DPWM modules, thereby allowing a dramatic improvement in the simulation times at the cost of reduced accuracy, for example, a *100 ms* simulation will take about *3 min* to be completed. The SABER schematics of this model are depicted in Figures C.7 and C.8, Appendix C.

3.6. Results of the digital valley-current and peak-current control

The performance and operation of the control algorithms and control platform were verified using the converter prototype described in Sections 3.2. For all the experiments performed, the converter was powered using a 15 kW dc power supply with maximum output voltage of 100 V and maximum output current of 150 A. A 48 kW resistive load with selectable values of 2.6 Ω , 3.8 Ω , 5.2 Ω and 10.4 Ω was employed as the load. The gains of

the proportional integral current controller were selected by trial and error and were $K_p = 30T$ and $K_i = 1$ for both control strategies.

The simulation results presented were obtained using the SABER switched models developed for valley-current and peak-current control. The simulation parameters are listed in Table 3.4.

rower converter			
Input inductance at 60 kHz, L _{in}	5.12 μΗ		
Input inductor winding resistance at 60 kHz, R _{in}	0.029 Ω		
IPT windings self-inductance at 30 kHz, L _a , L _b	75.14 μH		
Coupling coefficient of the IPT windings, k	0.997		
Output capacitance, Co	45 μH		
Load resistor, R _{load}	5.2 Ω		
Switching delay, <i>t_{dQ}*</i>	520 ns		

Table 3.4.	Simulation parameters for v	alley-current and peak-current control.
	Power C	onverter

Digital Signal Controller

Master clock frequency, <i>f</i> _{clk}	150 MHz
DPWM counters configuration	Count up mode/Count down mode accordingly
DPWM counter maximum value, v _{cmax}	5000
Sampling delay phase-a	140 ns
Sampling delay phase-b	220 ns
Processing delay, t _{d_proc}	3.52 μs
DPWM update delay	One-cycle

* As measured in Section 3.6.3. Includes the gate driver propagation delay, IGBT turn-on/turn-off delay, and the IGBT rise/fall time: $t_{dQ} = t_{p \ qd} + t_{d(on)} + t_{r(on)}$.

3.6.1. Steady-state operation of valley-current control strategy

Figures 3.14(a) and 3.14(b) show a comparison of the experimental and simulation results obtained using the valley-current control algorithm at the points of operation D = 0.43 and D = 0.62 respectively. The point of operation D = 0.5 was unattainable with this control strategy due to the non-linear behaviour of the phase-current ripple amplitude [10]. This issue is discussed with more detail in Section 3.6.4.

The waveforms plotted in these figures are the phase-a and phase-b switching signals, v_{1a} and v_{1b} respectively, the converter phase-a and phase-b currents, i_a and i_b respectively, the input inductor current, i_{Lin} , calculated as $i_{Lin} = i_a + i_b$ and the differential



current, i_{diff} , calculated as $i_{diff} = \frac{1}{2}$ ($i_a - i_b$). Also, the average-values of the current waveforms presented are included below the legends of each plot. The results shown were obtained using a load resistance R_{load} of 5.2 Ω , an input voltage V_{in} of 80 V and a reference-current of 20 A and 40 A, Figures 3.14(a) and 3.14(b) respectively.

The experimental and the simulation results correspond closely and are in agreement with the theoretical waveforms presented in Figure 2.2, Chapter 2. A discrepancy in the duty-ratio of the experimental and the simulation waveforms can be observed. To quantify this difference, the duty-ratio of the driving signals were measured and summarized in Table 3.5. From this table the difference between the experimental and the simulation results was found to be of 2 % for D = 0.43 and 3 % for D = 0.62. This discrepancy is attributed to delays in the control loop that were not taken into account in the simulation model, such as measurement delays of the current transducers for example.

Table 3.5. Comparison of the duty-ratio of the experimental and simulation results of thevalley-current control algorithm (Figure 3.13).

Point of Operation	Reference valley-current	Experimental duty-ratio	Simulation duty- ratio	Experimental and simulation difference
D < 0.5	20 A	43 %	45 %	2 %
D > 0.5	40 A	62 %	65 %	3 %

Table 3.6 shows a comparison of the valley-current measured from both the experimental and simulation results of Figure 3.14(a) and 3.14(b). The relative error of these measurements with respect to the reference-current was calculated and is also included in this table. It is observed that the valley-current from both the experimental and the simulation results is generally smaller than the desired reference-current.

Table 3.6. Phase-current measurements of the experimental and simulation results of thevalley-current control algorithm. (Figure 3.14).

Doint of	Poforonco	Experimental		Simulation	
Operation	valley-current	Phase-a I _{a_v} / %Err	Phase-b I _{b_v} / %Err	Phase-a I _{a_v} / %Err	Phase-b I _{b_v} / %Err
D < 0.5	20 A	14.2 A / 28 %	16.7 A / 16 %	16.6 A / 17 %	16.6 A / 17 %
D > 0.5	40 A	33.8 A / 15 %	36.3 A / 9 %	37.8 A / 5.5 %	37.8 A / 5.5 %

As explained in Section 3.5.2, this is attributed mainly to the turn-on delay of the transistors which produces a mismatch between the sampling instants and the location of the valley-current. Moreover, the errors from both the simulation and the experimental results are much smaller at high-current profiles since the difference between the valley-current and the desired reference-current is very similar over the whole range of operation.

A close look at the experimental data of Table 3.6 reveals an imbalance of 2.5 A in the phase currents that is not evident in the simulation results. This is thought to be caused by the difference in the parameters of the switching devices which are likely to cause different delays in the turn-on instants. Also, other factors such as mismatch between the current sensor parameters may contribute as well. The current imbalance is reflected as a dc offset of approximately -1 A in the differential current.

Finally, from the data of Table 3.6, it is possible to observe that the error of the experimental results is generally larger than the error obtained with the simulation. This may be attributed to the tolerance of the components forming the conditioning stages and a variation of their parameters due to environmental conditions, mainly temperature.

3.6.2. Steady-state operation of the peak-current control strategy

In a similar way as in the previous section, Figure 3.15(a) and 3.15(b) show a comparison of the experimental and simulation results obtained with the peak-current control algorithm during steady-state operation. These figures show the waveforms obtained for the points of operation D = 0.25 and D = 0.61 respectively, with $R_{load} = 5.2 \Omega$ and $V_{in} = 80 V$. The reference-current was programmed at 25 A in Figure 3.15(a), and 70 A in Figure 3.15(b). The simulation parameters are listed in Table 3.4. The point of operation D = 0.5 was also unattainable with this control algorithm.

The simulation and experimental waveforms are in agreement with the theoretical waveforms in Chapter 2. In a similar way as in the previous section, Table 3.7 shows a comparison of the duty-ratios measured from the experimental and the simulation results, whilst Table 3.8 summarizes the peak-current measurements for the experimental and simulation waveforms and the relative error with respect to the reference-current.





Point of Operation	Reference peak-current	Experimental duty-ratio	Simulation duty- ratio	Experimental and simulation difference
D < 0.5	20 A	25 %	18 %	6 %
D > 0.5	60 A	61 %	60 %	1 %

Table 3.7. Comparison of the duty-ratio of the experimental and simulation results of the peak-current control algorithm (Figure 3.15).

Table 3.8.	Phase-current measurements of the experimental and simulation results of the peak-
	current control algorithm. (Figure 3.14).

Doint of	Poforonco	Experimental		Simulation	
Operation	peak-current	Phase-a I _{a_p} / %Err	Phase-b I _{b_p} / %Err	Phase-a I _{a_p} / %Err	Phase-b I _{b_p} / %Err
D < 0.5	20 A	25.9 A / 29 %	26.8 A / 34 %	22.1 A / 10 %	21.5 A / 7.5 %
D > 0.5	60 A	69.0 A / 15 %	69.4 A / 15 %	62.6 A / 4.3 %	63.4 A / 5.6 %

From Table 3.7 a difference in the experimental and simulation results of 6% for D = 0.25 and 1% for D = 0.61 is observed. From the data of Table 3.8, the peak-current obtained is generally larger than the reference-current, whilst the relative error shows a tendency to be smaller for the simulations. These results are consistent with the results obtained for the valley-current control algorithm shown in Table 3.6. Once again, the difference between the peak-current and the desired reference-current can be explained by the mismatch between the location of the sampling instants and the peak-current, which also have a repercussion in the duty-ratio discrepancy observed in Table 3.7. The imbalance between the phase currents was found to be approximately 1 A for D = 0.25 and for D = 0.61.

3.6.3. Timing of the control algorithms

To examine the processing time, t_{d_proc} , of the control algorithms and to estimate the location of the sampling instants on the phase-current waveforms, a digital output port of the DSC was programed to be set high at the start of the ADC interrupt service routine and to be cleared low when the new calculated duty-ratio was loaded into the DPWM registers. The square waveform generated by this port provides a way to measure the length of the interrupt service routine and will be denoted v_{timing} .



Figure 3.16. Experimental measurement of the processing time of the valley-current control. System conditions: $V_{in} = 80 V$; $R_{load} = 5.2 \Omega$; D = 0.613, $I_{Lin} = 101 A$, $V_o = 217 V$. (a) Two complete cycles. (b) Expansion of the waveforms at the v_{1a} rising edge.

Figures 3.16 and 3.17 present the experimental waveforms measured for the valleycurrent and the peak-current control respectively. The results shown were acquired under steady-state operation by gradually adjusting the reference-current to achieve an average input current of 100 A, which corresponds to a duty-ratio of approximately 63.1 %. The system conditions employed were $V_{in} = 80$ V and $R_{load} = 5.2 \Omega$. Each of these figures show two screen captures where four waveforms are depicted in the following order: the transistor driving signals, v_{1a} and v_{1b} respectively, v_{timing} and the phase-a current, i_a . The waveforms were arranged in a similar way to the theoretical waveforms shown in Figure 3.12, which were employed to analyse the timing of the control algorithms in Section 3.4.3.



Figure 3.17. Experimental measurement of the processing time of the peak-current control. System conditions: $V_{in} = 80 V$; $R_{load} = 5.2 \Omega$; D = 0.613, $I_{Lin} = 101 A$, $V_o = 217 V$. (a) Two complete cycles. (b) Expansion of the waveforms at the v_{1a} falling edge.

For the valley-current control algorithm, Figure 3.16(b) shows the waveforms at the turn-on instant of v_{1a} . Conversely for the peak-current control algorithm, Figure 3.17(b) shows an expansion of the waveforms at the turn-off instant of v_{1a} . The main events of the control process are denoted at the bottom of these figures with respect to the rising-edge or the falling-edge of v_{1a} accordingly. The sampling instants of the phase-a and phase-b currents denoted timing '2' and timing '3' were obtained from the DSC manual and occur at 140 ns and 220 ns respectively, Table 3.3. From v_{timing} it is seen that the interrupt service routine starts at 1.05 μ s and that the calculated duty-ratio is loaded into the DPWM

registers at 3.52 μ s. The values of the principal timings in these figures are listed in Table 3.9. From the data collected it is seen that the total processing time, t_{d_proc} , is 3.5 μ s and is identical for both control algorithms as expected. Finally, the delay on the switching of the converter transistor, t_{dQ} , was measured to be approximately of 560 ns for the turn-on and 480 ns for the turn-off instants. From these measurements the mean value was calculated as 520 ns and it is used to model both the turn-on and the turn-off delays of the ideal switches employed in the SABER switched model of the converter, Chapter 2.

Parameter	Time
$t_{ADC} + t_{INT_SRVC}$	1.05 μs
t _{ADC} *	460 ns
t _{INT_SRVC}	590 ns
t _{comp}	2.4 μs
t _{d_proc}	3.5 μs
t _{dQ} (Valley-current)	560 ns
t _{dQ} (Peak-current)	480 ns

Table 3.9. Experimental timings of the valley-current and the peak-current control algorithms.

* As specified in the TMS320F28335 manual [2].

3.6.4. Range of operation

The range of operation of the valley-current and peak-current control algorithms was verified by gradually increasing the reference-current from 15 A to the maximum value allowed by the power supply. This test revealed an uncontrollable region or dead-zone around the point of operation D = 0.5. To investigate further this characteristic, a set of simulations was run to evaluate the capability of the system to track a gradually increasing ramp reference-current. Since the control platform was not designed to follow arbitrary reference-currents, the study was carried out employing the SABER switched simulation models.

Each graph shown in Figure 3.18 depicts a comparison of the reference current, i_{ref} , and the phase-a current, i_a , on the first plot; a comparison of the sampled phase-a current, i_a^* , and the sampled reference-current, i_{ref}^* on the second plot; and the phase-a error-current, defined as $i_{ea}^* = i_{ref}^* - i_a^*$, in the third plot. Figures 3.18(a), 3.18(b) and 3.18(c) were obtained employing the valley-current control strategy whilst the graphs shown in Figures 3.18(d), 3.18(e) and 3.18(f) were obtained using the peak-current control strategy.



Figure 3.18. Response of the valley and peak current control algorithms to a ramp reference-current obtained by simulation. (a), (b) and (c) Valley-current control algorithm for R_{load} of 10.3 Ω, 5.2 Ω and 2.6 Ω respectively. (d), (e) and (f) Peak-current control algorithm for R_{load} of 10.3 Ω, 5.2 Ω and 2.6 Ω respectively. V_{in} = 80 V.

As expected, all the plots show a constant error between the ramp-reference and the valley-current/peak-current, which is caused by the mismatch of the sampling instants caused by switching delay. However, it is also observed from i_{ea} * that the calculated error-current exhibits an increment for points or operation around 50% of the duty-ratio (where the phase-current ripple is minimum). These uncontrollable regions are caused by the non-linear behaviour of the current-ripple and it is seen that they are reduced when the load is increased, Figures 3.18(c) and 3.18(f).

3.6.5. Dynamic performance

The experimental results presented in this section were gathered to evaluate the capability of the control algorithms to regulate and balance the phase-currents during a step-change in the reference-current and a step-change in the load. The results are organized as follows. For the valley-current control algorithm, Figures 3.19 and Figure 3.20 show the transient response of the system to a step-change in the reference-current and to a step-change in the load respectively. Likewise, Figures 3.21 and 3.22 show the transient responses of the system employing the peak-current control algorithm. Figures 3.19 to 3.22 comprise three scope-screen captures denoted (a), (b) and (c) respectively. (a), shows an overview of the transient response, (b) shows a magnification of the increasing current transient and (c) shows a magnification decreasing current transient. The relevant waveforms presented are the output voltage, v_o ; the phase-a and phase-b currents shown one on top of the other, i_a and i_b respectively; and the differential current calculated as $i_{diff} = \frac{1}{2} (i_a - i_b)$. All the results were obtained with $V_{in} = 80 V$. In addition, Figures 3.19 and 3.21, include an extra waveform plotted in green that was employed to synchronize the scope with the step-changes in the reference-current.

During the step-reference tests of the valley-current control algorithm, Figure 3.19, R_{load} was fixed to 5.2 Ω whilst i_{ref} was toggled between 20 A and 40 A. Conversely, during the step-load tests, Figure 3.21, i_{ref} was fixed at 40 A whilst the load resistance was switched between 2.6 Ω and 5.2 Ω . For the step-reference tests of the peak-current control, Figure 3.21, R_{load} was fixed to 5.2 Ω whilst i_{ref} was toggled between 20 A and 70 A, whilst for
the step-load tests i_{ref} was programmed at 60 A and the load resistor was switched between 2.6 Ω and 5.2 Ω .

The results show in general a satisfactory performance. The differential and the phase-current waveforms show no evidence of saturation of the magnetic components and demonstrate the capability of the control strategies to maintain balanced currents in the converter phases even in transitory conditions. The transient times of the valley-current control were of approximately *30 ms* for the step-reference increment and *25 ms* for the step-reference decrement, whilst the transient times of the peak-current control were approximately of *25 ms* for the step-reference increment and *35 ms* for the step-reference decrement. The transient speeds are similar in both algorithms as the PI compensators used to regulate the current loops are equal for both control methodologies.

Finally, the step-load tests of both control algorithms, the phase current waveforms show a particular feature where the current seems to evolve in stages which is attributed to the non-instantaneous switching of the resistor bank.



Figure 3.19. Experimental results of the step-reference response of the valley-current current control.
 System conditions: I_{ref1} = 20 A, I_{ref2} = 40 A, V_{in} 80 V, R_{load} = 5.2 Ω. (a) Overview of the test. (b)
 Magnification of the step-up transient. (c) Magnification of the step-down transient.



Figure 3.20. Experimental results of the step-load response of the valley-current current control. System conditions: I_{ref} = 40 A, V_{in} 80 V, R_{load1} = 2.6 Ω, R_{load2} = 5.2 Ω. (a) Overview of the test. (b) Magnification of the step-up transient. (c) Magnification of the step-down transient.



Figure 3.21. Experimental results of the step-reference response of the peak-current current control. System conditions: $I_{ref1} = 20 \text{ A}$, $I_{ref2} = 60 \text{ A}$, $V_{in} 80 \text{ V}$, $R_{load} = 5.2 \Omega$. (a) Overview of the test. (b) Magnification of the step-up transient. (c) Magnification of the step-down transient.



Figure 3.22. Experimental results of the step-load response of the peak-current current control. System conditions: $I_{ref} = 60 \text{ A}$, $V_{in} 80 \text{ V}$, $R_{load1} = 2.6 \Omega$, $R_{load2} = 5.2 \Omega$. (a) Overview of the test. (b) Magnification of the step-up transient. (c) Magnification of the step-down transient.

3.7. Results of the digital average-current control strategies with normal and immediate update

In this section, the performance and operation of the average-current control with normal update and with immediate update are presented. The experimental results were obtained employing the converter prototype and the control platform previously introduced in Sections 3.2 and Section 3.3. The simulation results were obtained using the SABER switched model developed for triangular-carrier modulation and the parameters listed in Table 3.10.

Table 3.10. Simulation parameters average-current control.		
Power Converter		

Input inductance at 60 kHz, L _{in}	5.12 μΗ
Input inductor winding resistance at 60 kHz,	0.029 0
R _{in}	0.010
IPT windings self-inductance at 30 kHz, L _a , L _b	75.14 μH
Coupling coefficient of the IPT windings, k	0.997
Output capacitance, C _o	45 μH
Load resistor, R _{load}	5.2 Ω
Switching delay, <i>t_{dQ}*</i>	520 ns

Master clock frequency, <i>f</i> _{clk}	150 MHz	
DPWM counters configuration	Count up-down mode	
DPWM counter maximum value, v _{cmax}	2500	
Sampling delay phase-a	140 ns	
Sampling delay phase-b	220 ns	
Processing delay, t _{d_proc}	4.2 μs	
DPWM update delay	Half-cycle for normal update algorithm None for immediate update algorithm	

Digital Signal Controller

* As measured in Section 3.6.3. Includes the gate driver propagation delay, IGBT turn-on/turn-off delay, and the IGBT rise/fall time: $t_{dQ} = t_{p_gd} + t_{d(on)} + t_{r(on)}$.

The converter power stage was powered using a 15 kW dc power supply with maximum output voltage of 100 V and maximum output current of 150 A. A resistor bank of 48 kW with selectable values of 2.6 Ω , 3.8 Ω , 5.2 Ω and 10.4 Ω was employed as the load. The gains of the digital PI compensator employed during the experiments were $K_p = 30T$ and $K_i = 1$ for the normal update algorithm and $K_p = 30T$ and $K_i = 10$ for the immediate update algorithm. These gains were selected using the simplified non-interleaved model of the converter described in Chapter 4 in conjunction with SISO tool in MATLAB to guarantee the

Chapter 3

stability of the converter. The integral gain selected was the maximum possible before the system became unstable, however, these values were not properly designed and may affect how much better the immediate update method appears.

3.7.1. Steady-state operation

The results obtained with the average-current control algorithm with normal update and the variation employing immediate update showed virtually no difference in the steady-state results. For this reason, only the results obtained for the control algorithm with immediate update are presented in this section. Figures 3.23(a), 3.23(b) and 3.24 show a comparison of the measured experimental waveforms and the simulation results of the converter operating in continuous conduction mode and with $V_{in} = 80$ V and $R_{load} = 5.2 \Omega$. Figures 3.23(a) and 3.23(b) show the results obtained for the points of operation D = 0.43and D = 0.61, which were obtained by programming i_{ref} to 20 A and 60 A respectively. Figure 3.24 shows the results at D = 0.5. This particular point of operation was achieved by gradually increasing the reference-current until the system reached the desired duty-ratio. The waveforms plotted in these figures are the gate-driver signals, v_{1a} and v_{1b} , the phase currents, i_a and i_b , the input current, i_{Lin} , and finally the differential current, i_{diff} . In a similar way as for the valley and peak control strategies, Table 3.11 and 3.12 summarize the key aspects of the experimental and simulation waveforms.

Point of Operation	Reference peak-current	Experimental duty-ratio	Simulation duty- ratio	Experimental and simulation difference
D < 0.5	20 A	43.9 %	39.2 %	4.7 %
D = 0.5	≈ 30 A	50.1 %	49.8 %	0.2 %
D > 0.5	50 A	61.8 %	61.3 %	0.7 %

Table 3.11. Comparison of the duty-ratio of the experimental and simulation results of the average-current control algorithm with immediate update (Figures 3.23 and 3.24).

In general, the experimental and simulation results show a good correlation. The waveforms measured are in agreement with the theoretical expectation presented in Figure 2.2, Chapter 2. The practical values of the duty-ratio are slightly larger than those obtained in the simulation. From the measurements of Table 3.11 it is seen that the maximum difference between the experimental and the simulation duty-ratios is about *4.7* % and it







Figure 3.24. Experimental and simulation waveforms of the dual interleaved boost converter with IPT in steady state operation obtained employing the immediate update average-current control algorithm. $I_{ref} = 30 \text{ A}$. D = 0.5. System conditions: $V_{in} = 80 \text{ V}$: $R_{innet} = 5.2 \Omega$.

seems to diminish as the duty-ratio is increased (the value at D = 0.5 is the smallest as the reference-current was adjusted to set the converter in that point of operation). This difference is attributed to the slight mismatches in component parameters, and non-ideal elements of the converter as well as the turn-on and turn-off delays of the transistors.

The practical average level of the phase-currents is generally above the value obtained with the simulations, Table 3.12. The difference found was of about *5 A* between the experimental and simulation values. Once more this is attributed to the mismatch between the sampling instants and the actual point where the average-current is located which are a consequence of the delay in the control signals and the switching devices.

From Table 3.12, an imbalance in the average-value of the phase current waveforms is observed of about 1 A for D = 0.43 and 2 A for D = 0.61. This is reflected as a dc offset of 0.4 A and 1 A respectively in the differential current of the system. The cause of this imbalance is likely to be related to the mismatches in the parameters of the converter and control platform and stray parasitic elements.

Doint of	Deference	Experi	mental	Simu	lation
Operation	peak-current	Phase-a I _{a_p} / Err	Phase-b I _{b_p} / Err	Phase-a I _{a_p} / Err	Phase-b I _{b_p} / Err
D < 0.5	20 A	25.1 A / 25 %	24.3 A / 21 %	19.6 A / 2 %	19.5 A / 2 %
D = 0.5	≈ 30 A	30.5 A / 1.6%	30.6 A / 2 %	30.1 A / 0.3 %	30.1 A / 0.3 %
D > 0.5	50 A	54.1 A / 8 %	52.0 A / 4 %	49.9 A / 0.2 %	49.7 A / 0.3 %

Table 3.12. Phase-current measurements of the experimental and simulation results of theaverage-current control algorithm with immediate update. (Figures 3.23 and 3.24).

In conclusion, the results obtained confirm the correct operation of the control algorithms in steady-state and validate the correct operation of the SABER switched simulation model.

3.7.2. Timing of the control algorithms

Figures 3.25 and 3.26 illustrate the waveforms used to measure the processing time, t_{d_proc} , and estimate the location of the sampling instants for the average-current control algorithm with normal update and with immediate update. The waveforms shown in these figures correspond to the point of operation D = 63.1%. The system conditions were $V_{in} = 80 V$ and $R_{load} = 5.2 \Omega$. The waveforms displayed in these figures are the transistor driving signals v_{1a} and v_{1b} , the signal used to measure the processing time, v_{timing} , and the phase-a current i_a . The main events of the control process are denoted at the bottom of these figures. The sampling instants of the phase-a and phase-b currents denoted timing '2' and timing '3' were obtained from the DSC manual and occur at 140 ns and 220 ns respectively, Table 3.3.

Table 3.13 summarizes the key measurements from the waveforms. For the average-current control algorithm with normal update, the computation time of the control algorithm was found to be of 2.3 μ s and the total processing time was 3.35 μ s which is



Figure 3.25. Experimental measurement of the processing time of the average-current control with normal update. System conditions: V_{in} = 80 V; R_{load} = 5.2 Ω; D = 0.613, I_{Lin} = 101 A, V_o = 217 V. (a) Two complete cycles. (b) Expansion of the waveforms at the v_{timing} rising edge.

Table 3.13.	Experimental timings of the average-current control algorithm with normal and
	immediate update.

Parameter	Time
$t_{ADC} + t_{INT_SRVC}$	1.05 µs
t _{ADC} *	460 ns
t _{INT_SRVC}	590 ns
t _{comp} (normal algorithm)	2.3 μs
t _{d_proc} (normal algorithm)	3.35 μs
t _{comp} (immediate algorithm)	3.15 μs
t _{d_proc} (immediate algorithm)	4.2 μs

* As specified in the TMS320F28335 manual [2].

170 ns shorter than that of the valley-current and peak-current algorithms, Table 3.9. On the other hand, the computation time for the algorithm with immediate update was found to be 3.15 μ s, which is approximately 1 μ s longer than the normal update algorithm, as expected, due to the extra-tasks carried out for the anti-glitch scheme described in Section 3.4.4. Consequently, the total processing time of the immediate update algorithm was found to be 4.2 μ s.



Figure 3.26. Experimental measurement of the processing time of the average-current control with immediate update. System conditions: V_{in} = 80 V; R_{load} = 5.2 Ω; D = 0.613, I_{Lin} = 101 A, V_o = 217 V. (a) Two complete cycles. (b) Expansion of the waveforms at the v_{timing} rising edge.

3.7.3. Dynamic performance

This section presents the experimental results for the average-current control algorithms with normal and immediate update for both a step-change in the referencecurrent and a step-change in the load. The results are arranged in a similar way as those presented for the valley-current and the peak-current control algorithms in Section 3.6.4. The figures presented display an overview of the transient response, a magnification of the increasing current transient and a magnification of the decreasing current transient. The waveforms displayed are the output voltage, v_a ; the phase-a and phase-b currents shown one on top of the other, i_a and i_b respectively; and the differential current calculated as $i_{diff} = \frac{1}{2} (i_a - i_b)$. All the results were obtained with $V_{in} = 80 V$. In addition, Figures 3.26 and 3.28, include an extra waveform plotted in green that was employed to synchronize the scope with the step-changes in the reference-current.

Figures 3.27 and Figure 3.28 show the step-reference and step-load results with normal update, whilst Figures 3.29 and 3.30 depict the transient response of the system with immediate update. During the step-reference tests R_{load} was 5.2 Ω whilst i_{ref} was switched between 20 A and 50 A. Conversely, during the step-load tests i_{ref} was 50 A whilst the load resistance was switched between 2.6 Ω and 5.2 Ω .

Both control algorithms provided satisfactory regulation of the phase-currents. The average value of the phase-currents shows a steady-state error with respect to the reference-current of approximately 4 A for D > 0.5 and 2 A for D < 0.5 which is consistent with the steady-state tests of Section 3.7.1.

The two sets of results confirm the orderly operation of both control algorithms, with the phase currents remaining well-balanced throughout. Furthermore the results illustrate the much more rapid response that is obtained with the immediate update algorithm; the transient times of the phase currents are reduced by almost an order of magnitude, having values of around 2 *ms*. The improvement in speed is due to the use of a ten times larger integral gain in the immediate update system, which was made possible by the extended stability limits, which in turn were due to the reduced computation time.



Figure 3.27. Experimental results of the step-reference response of the average-current current control with normal update. System conditions: I_{ref1} = 20 A, I_{ref2} = 50 A, V_{in} 80 V, R_{load} = 5.2 Ω. (a) Overview of the test. (b) Magnification of the step-up transient. (c) Magnification of the step-down transient.



Figure 3.28. Experimental results of the step-load response of the average-current current control with normal update. System conditions: I_{ref} = 50 A, V_{in} 80 V, R_{load1} = 2.6 Ω, R_{load2} = 5.2 Ω. (a) Overview of the test. (b) Magnification of the step-up transient. (c) Magnification of the step-down transient.



Figure 3.29. Experimental results of the step-reference response of the average-current current control with immediate update. System conditions: I_{ref1} = 20 A, I_{ref2} = 50 A, V_{in} 80 V, R_{load} = 5.2 Ω. (a) Overview of the test. (b) Magnification of the step-up transient. (c) Magnification of the step-down transient.



Figure 3.30. Experimental results of the step-load response of the average-current current control with immediate update. System conditions: I_{ref} = 50 A, V_{in} 80 V, R_{load1} = 2.6 Ω, R_{load2} = 5.2 Ω. (a) Overview of the test. (b) Magnification of the step-up transient. (c) Magnification of the step-down transient.

Chapter 3

3.8. Summary

The design, component selection and construction of a digital control platform to interface a Texas Instruments TMS320F28335 digital signal controller with a *25 kW* of a dual interleaved converter prototype has been presented. Also the implementation of three digital control strategies for phase-current regulation have been described employing uniformly-sampled PWM with valley-current, peak-current and average-current control. In addition, a variant of the average-current control has been developed where the update delay caused by the DPWM operation is eliminated allowing the stability range of the converter to be extended and therefore enabling a faster response to be achieved. The correct operation of the control strategies was verified experimentally for a range of tests and compared against detailed SABER simulation results.

Although providing orderly operation of the converter and balanced phase-currents, both the valley-current and the peak-current algorithms were shown to be of limited value for this converter since uncontrollable regions were identified for duty-ratios in the region of *0.5*. Both forms of the average-current control algorithm allowed the converter to operate over the full range of duty-ratios whilst maintaining balanced phase currents. A maximum mismatch between the phase currents of *2 A* in *50 A* or *4 %* was observed, resulting in a *1 A* dc-offset in the IPT differential current.

Throughout detailed analysis of the processing delays, the immediate update version of the average-current control was developed, allowing the phase-current to be sampled and the duty-ratio to be updated within a switching period, but with the constraint of a minimum duty-ratio of 0.25. Also, special measures were devised to prevent the generation of erroneous switching signals by the digital modulator. The immediate update algorithm provided transient times of around 2 ms, however more detailed investigation of the control parameter settings and stability limits is considered in the following chapters.

3.9. References

- [1] Texas Instruments, "Delfino TMS320F28335 DSC Control Card," [on-line], Available at: http://www.ti.com/tool/tmdscncd28335, [Accessed on 19-03-2015].
- [2] Texas Instruments, "TMS320F28335 Digital Signal Controller," Data Manual [online], Available at: http://www.ti.com/product/tms320f28335, [Accessed on 21-05-2015], 2012.
- [3] LEM, "Current transducer LA 200-P," Product data sheet, LEM, 2008.
- [4] Texas Instruments, "OPA2343UA rail-to-rail amplifier," Product data sheet, Texas Instruments, 2014.
- [5] CONCEPT, "2SC0435T Dual-channel high-power low-cost IGBT driver," Description and application manual, CONCEPT, April 2014.
- [6] CONCEPT, "AN–1101, Application with Scale-2 gate driver cores," Application note, CONCEPT, September 2013.
- [7] Texas Instruments, "TMDSDOCK28335 Docking Station," [on-line], Available at: http://www.ti.com/tool/tmdsdock28335,[Accessed on 19-03-2015].
- [8] Texas Instruments, "TMS320F28335 CPU and Instruction Set Reference Guide," User Guide [on-line], Available at: http://www.ti.com/lit/ug/spru430f/spru430f.pdf, [Accessed on 21-05-2015], 2012.
- [9] MITSUBISHI IGBT modules, "CM300DU-12NFH," Product data sheet, MITSUBISHI, 2009.
- [10] K. J. Hartnett, M. S. Rylko, J. G. Hayes, and M. G. Egan, "A comparison of classical two phase (2L) and transformer - Coupled (XL) interleaved boost converters for fuel cell applications," in Applied Power Electronics Conference and Exposition (APEC), 2010 Twenty-Fifth Annual IEEE, 2010, pp. 787-793.

Chapter 4 Modelling phase interactions in dual interleaved converters

4.1. Introduction

In this Chapter, multirate discrete-data systems theory is used to extend the converter model developed in Chapter 2 to incorporate the interleaved operation of the phases, specifically when feedback is used to regulate the individual phase currents. The small-signal closed-loop transfer functions of the system in the z-domain are determined for two cases: when the interleaved operation of the converter is taken into account, termed the non-synchronous model, and when this interleaved interaction is disregarded, referred to as the synchronous model. The models are then compared with each other to examine their differences and against other switched and averaged models in order to provide validation. Finally, the predicted stability limits are compared with those from the practical system.

4.2. Representation of the converter in the control diagram form

Figure 4.1 shows a diagram of the average-current controlled dual interleaved boost converter with an inter-phase transformer and a digital signal controller. The DSC operates in an interleaved manner to regulate the average-current flowing through each phase of the converter using two individual current feedback loops, which share the same reference to ensure an equal distribution of the input current between the IPT windings.

In this diagram, the current-feedback signals $i_a(t)$ and $i_b(t)$ have transducer gains $H_a(s)$ and $H_b(s)$ respectively. These signals are sampled at a uniform rate by the analogue-to-digital converter, ADC. The sampling of $i_a(t)$, $i_b(t)$ and $i_{ref}(t)$ is represented by two



Figure 4.1. Digital feedback loop for regulation of the average phase-current of the dual-interleaved boost converter with inter-phase transformer.

samplers, S_a and S_b , with the sampled signals denoted $i_a[n]$, $i_b[n]$ and $i_{ref}[n]$ respectively and are employed to calculate the error signals $i_{ea}[n]$ and $i_{eb}[n]$. The samplers S_a and S_b share the same sampling period, T, but the sampling instants of the sampler S_b are delayed behind those of S_a by half a sampling period. The error signals, $i_{ea}[n]$ and $i_{eb}[n]$, in the two control loops are passed through a controller transfer function C(z) before being passed on to the DPWM modules which generate the driving signals for the converter transistors.

4.2.1. Small-signal block diagram modelling

Figure 4.2(a) shows the system of Figure 4.1 represented as a standard closed-loop, block diagram in the Laplace-domain. In this diagram and in convention with the digital

control theory notation, the Laplace-domain representation of the time-domain pulse signals produced by the sampling devices are denoted by an asterisk. This representation is a variation of the Laplace transformation, commonly known as the starred transformation, which is defined by Equation (4.1), [1,2]:

$$f^*(s) = \left[f(s)\right]^* = \sum_{k=-\infty}^{+\infty} f(kT) e^{-kTs}$$
(4.1)

where *k* is the sample index and *T* is the sampling period.

In the system block diagram of Figure 4.2(a), the small-signal dynamics of the dualinterleaved boost converter are represented by the four Laplace-domain transfer functions determined in Chapter 2, Section 2.3, that relate the small changes in the transistor dutyratios, $\tilde{d}_a(s)$ and $\tilde{d}_b(s)$, to the small changes in the converter phase-currents, $\tilde{i}_a(s)$ and $\tilde{i}_b(s)$, as shown in Equations (4.2) and (4.3):

$$\tilde{i}_{a}(s) = G_{daia}(s)\tilde{d}_{a}(s) + G_{dbib}(s)\tilde{d}_{b}(s)$$
(4.2)

$$\tilde{i}_{b}(s) = G_{daib}(s)\tilde{d}_{a}(s) + G_{dbib}(s)\tilde{d}_{b}(s)$$
(4.3)

where $G_{daia}(s) = G_{dbib}(s)$ and $G_{dbia}(s) = G_{daib}(s)$.

The current transducers of the control loops are represented by their respective transfer functions $H_a(s)$ and $H_b(s)$, whilst the analog-to-digital conversion process is modelled by the sampler devices S_a and S_b . For the sake of simplicity, the quantization non-linearity of the ADC and the delays associated with the measurements and the conditioning stages of the system are disregarded in this work.

The transfer function of the digital compensation network used in the control loops is represented by the blocks C(z), whilst its associated computation delay, t_{d_proc} , is included with the DPWM modules. The DPWM modules are represented by a ZOH extrapolator connected in cascade with the associated processing delay of the compensation network, t_{d_proc} . As described in Chapter 1, Section 1.5.1.2, this is the simplest way to represent a DPWM, but may be replaced by the more elaborated models, [3], to enhance the accuracy in future research.

Chapter 4

As shown in Chapter 3, Section 3.7, when average-current control is employed, the DPWMs can be updated immediately after the calculation of the new duty ratio or have a constant delay of T/2. In the former case, the processing time is $t_{d_proc} < T/2$ and it is referred to as the immediate update mode. In the latter case the processing time can be defined as $t_{d_proc} = T/2$ and is referred to as the normal update mode. Summarizing, the update delay of the DPWM modules can be represented as:

$$\Delta PWM(s) = \begin{cases} e^{-s(td_proc)} & \forall & t_{d_proc} < \frac{7}{2} \\ e^{-s(\frac{7}{2})} & \forall & t_{d_proc} = \frac{7}{2} \end{cases}$$
(4.4)





Figure 4.2. Functional control block diagram of the digital feedback loop for regulation of the average phase-currents of the DIBC with IPT. (a) Complete system; (b) Simplified system.

Finally, the model of Figure 4.2(a) can be simplified by moving S_a and S_b after the feedback summing junctions resulting in the diagram of Figure 4.2(b).

As described in Chapter 1, Section 1.6, previous work has neglected the interleaved sampling operation of the two phases. Therefore a key objective in this thesis is to devise a way of including the interleaved sampling and to examine its impact on controller design and converter stability.

4.3. Modelling of the interaction of the digital current control loops

To analyse the system stability and design an appropriate compensation network, analytical equations are required that relate the small changes in the reference-current input to small changes in the phase-currents. Therefore the closed-loop transfer functions $\tilde{i}_a(z)/\tilde{i}_{ref}(z)$ and $\tilde{i}_b(z)/\tilde{i}_{ref}(z)$, also denoted $G_{iairef}(z)$ and $G_{ibiref}(z)$, will be found employing multirate sampled-data systems theory.

4.3.1. The sampler-decomposition method

The sampler-decomposition [1,2] is used to model the effects of interleaved sampling. This methodology was chosen over other methods, such as the fictious-sampler



Figure 4.3. Decomposition of the sampler S with sampling rate T/n into *n* synchronized sub-samplers S_0 with sample rate T, [2].

method or the infinite-series representation, as it can be easily incorporated into the system model and it does not require a complex mathematical development. Moreover, owing to its versatility it is easily extendable to interleaved converters with two or more phases.

In this method, a sampler with sampling rate T/n, where n is an integer and T is the sampling period, can be decomposed into n parallel-connected, synchronized sub-samplers with sampling rate T using time advance and time delay units as illustrated in Figure 4.3. In this figure, the sampler S with sampling rate T/n is decomposed into n synchronized sub-samplers, S_0 , which allocate the information corresponding to each sampling instant encompassed in the period T. Using the time advance and time delay units on each branch, the sampled data at different time instants is determined and relocated in such a way that all the interactions occur simultaneously in time and are finally combined in a single output $y^*(t)_n$. From the diagram of Figure 4.3, $y^*(t)_n$ may be expressed as:

$$y^{*}(t)_{n} = y_{0}^{*}(t) + \sum_{\rho=1}^{n-1} y_{\rho}^{*}(t)$$
 (4.5)

4.3.2. Implementation of the sampler decomposition method in the digital averagecurrent control loop

Considering the closed loop system depicted in Figure 4.2(b) and making use of the sampler decomposition method, S_b can be represented with respect to S_a as a single branch with time advance and time delay units equal to $e^{+sT/2}$ and $e^{-sT/2}$ as shown in Figure 4.4(a). Furthermore, the starred transform representation of the discrete-time signals of Figure 4.2(a) may be substituted directly by their z-domain representations owing to the link between the starred transform and the z-transform definitions. Equation (4.7) shows the definition of the z-transform, which can be derived from the starred transform definition, Equation (4.3), by replacing the term e^{-sT} with the z operator.

$$f(z) = \sum_{k=0}^{\infty} f(kT) z^{-k}$$
 (4.7)

The resultant system block diagram is shown in Figure 4.4(b), where two fictious samplers were added at the outputs of the block diagram to allow the derivation of the

input-output relations of the system in the z-domain and where $G_M(s)$ is defined by Equation (4.6):

$$G_{M}(s) = G_{h0}(s) \Delta PWM(s)$$
(4.6)

This model is referred to as the non-synchronous small-signal model the DIBC with IPT.



Figure 4.4. Implementation of the sampler decomposition method to model the average-current feedback loop of the DIBC with IPT: (a) Equivalent representation of the samplers S_a and S_b by a sampler with sampling period T. (b) Block diagram of the system with synchronous samplers.

4.3.3. Derivation of the closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$

In the following paragraphs, the closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ will be derived from Figure 4.4(b) for the case where $t_{d_proc} < T/2$, in other words, for the immediate DPWM update mode.

By analysis of the block diagram of Figure 4.4(b) the following relations can be written:

$$\tilde{d}_{a}(z) = C(z)\mathcal{Z}\left[\tilde{i}_{ref}(s)\right] - C(z)\tilde{d}_{a}(z)\mathcal{Z}\left[G_{M}(s)G_{di}(s)H_{a}(s)\right] - C(z)\tilde{d}_{b}(z)\mathcal{Z}\left[e^{-s^{T}/2}G_{M}(s)G_{dxi}(s)H_{a}(s)\right]$$
(4.8a)

$$\tilde{d}_{b}(z) = C(z)\mathcal{Z}\left[e^{s^{T/2}}\tilde{i}_{ref}(s)\right] - C(z)\tilde{d}_{b}(z)\mathcal{Z}\left[G_{M}(s)G_{di}(s)H_{b}(s)\right] - C(z)\tilde{d}_{a}(z)\mathcal{Z}\left[e^{s^{T/2}}G_{M}(s)G_{dxi}(s)H_{b}(s)\right] \quad (4.8b)$$

$$\tilde{i}_{a}(z) = \tilde{d}_{a}(z)\mathcal{Z}\left[G_{M}(s)G_{di}(s)\right] + \tilde{d}_{b}(z)\mathcal{Z}\left[e^{-s\frac{1}{2}}G_{M}(s)G_{dxi}(s)\right]$$
(4.8c)

$$\tilde{i}_{b}(z) = \tilde{d}_{b}(z)\mathcal{Z}\left[G_{M}(s)G_{di}(s)\right] + \tilde{d}_{a}(z)\mathcal{Z}\left[e^{s\frac{T}{2}}G_{M}(s)G_{dxi}(s)\right]$$
(4.8d)

The z-transforms found in Equations (4.8a) to (4.8d) are listed in Table 4.1 and are redefined in a compact notation for this particular case-study where $t_{d_proc} < T/2$, in other words, $\Delta PWM(s) = e^{-s(td_proc)}$. Upon substitution of the compact notation of the constituent z-transforms into Equations (4.8a) to (4.8d), Equations (4.9a) to (4.9d) are found:

Z-transform identified	Modified z-transform	Compact notation
$\mathcal{Z}\!\left\{\!e^{s\!$	$z\tilde{i}_{ref}(z, \frac{1}{2})$	$\widetilde{i}_{ref heta}(z)$
$\mathcal{I}(\mathcal{L}_{\mathcal{L}}(\mathcal{L}))$	$\mathcal{Z}\left\{G_{h0}(s)\Delta PWM(s)G_{di}(s)H_{A}(s)\right\}=$	
$\mathcal{L}\left\{\Theta_{M}(s)\Theta_{di}(s)\Pi_{A}(s)\right\}$	$=G_{di}H_{A}(z,1-t_{d_{proc}})$	$\mathbf{G}_{di}\mathbf{n}_{A}(\mathbf{Z})$
$\mathcal{F}_{\mathcal{F}}^{\mathcal{F}}$ (c) $\mathcal{F}_{\mathcal{F}}^{\mathcal{F}}$ (c) $\mathcal{F}_{\mathcal{F}}^{\mathcal{F}}$	$\mathcal{Z}ig\{G_{h0}(s)\Delta PWM(s)G_{di}(s)H_{B}(s)ig\}$	GH(z)
$\mathcal{L}\left\{\mathbf{G}_{M}\left(\mathbf{S}\right)\mathbf{G}_{di}\left(\mathbf{S}\right)\mathbf{H}_{B}\left(\mathbf{S}\right)\right\}$	$= G_{di} H_B \left(z, 1 - t_{d_proc} \right)$	$\Theta_{di} \Pi_B(Z)$
$\mathcal{T} = \mathcal{T}_{\mathcal{C}} (s) \mathcal{C} (s) \mathcal{H} (s)$	$\mathcal{Z}\left\{e^{-s\frac{\tau}{2}}G_{h0}(s)\Delta PWM(s)G_{dxi}(s)H_{A}(s)\right\}=$	G H(z)
$\mathcal{Z}\left\{ e \cap G_{M}(s) G_{dxi}(s) H_{A}(s) \right\}$	$= G_{dxi}H_A(z, 1 - (\frac{1}{2} + t_{d_proc}))$	$O_{dxi\phi} n_A(2)$
$\mathcal{F} = \{ e^{S^{1/2}} G (s) G (s) H(s) \}$	$\mathcal{Z}\left\{e^{s\frac{\pi}{2}}G_{h0}(s)\Delta PWM(s)G_{dxi}(s)H_{B}(s)\right\}=$	G H(z)
$\mathcal{L}_{\mathcal{L}} = \mathcal{O}_{\mathcal{M}}(3)\mathcal{O}_{dxi}(3)\mathcal{O}_{\mathcal{B}}(3)$	$= zG_{dxi}H_B\left(z,\left(\frac{1}{2}-t_{d_proc}\right)\right)$	$O_{dxi\theta} \Gamma_B(Z)$
$\mathcal{T} = \{c, (c) \in (c)\}$	$\mathcal{Z}\left\{G_{h0}(s)\Delta PWM(s)G_{di}(s)\right\}=$	G(z)
$\mathcal{L}\left\{ \mathbf{G}_{M}\left(\mathbf{S}\right)\mathbf{G}_{di}\left(\mathbf{S}\right)\right\}$	$=G_{di}(z,1-t_{d_proc})$	$\mathbf{U}_{di}(\mathbf{Z})$
$\mathcal{T}_{e^{-sT_{2}}G}(s)G(s)$	$e^{-s\frac{T}{2}}\mathcal{Z}\left\{G_{h0}(s)\DeltaPWM(s)G_{dxi}(s)\right\}=$	$G_{z}(z)$
$\mathcal{L}\left(\mathcal{C} \mathbf{U}_{M}(\mathbf{S})\mathbf{U}_{dxi}(\mathbf{S})\right)$	$= G_{dxi}\left(z, 1 - \left(\frac{1}{2} + t_{d_proc}\right)\right)$	$\mathbf{C}_{dxi\phi}(\mathbf{Z})$
$\mathcal{T} = \mathcal{T}_{\mathcal{B}}^{s_{\mathcal{I}_{2}}} \mathcal{G} (s) \mathcal{G} (s)$	$\mathcal{Z}\left\{e^{s\frac{1}{2}}G_{h0}(s)\Delta PWM(s)G_{dxi}(s)\right\}=$	$G_{1}(z)$
$\mathcal{L}_{\mathcal{L}} = \mathcal{L}_{\mathcal{M}}(\mathcal{S})\mathcal{L}_{dxi}(\mathcal{S})$	$= zG_{dxi}\left(z,\left(\frac{1}{2}-t_{d_{proc}}\right)\right)$	$\mathbf{U}_{dxi\theta}(\mathbf{z})$

Table 4.1. Constituent transfer functions of the digital average-current controlled DIBCwith IPT in the z-domain.

$$\tilde{d}_{a}(z) = C(z)\tilde{i}_{ref}(z) - C(z)\tilde{d}_{a}(z)G_{di}H_{a}(z) - C(z)\tilde{d}_{b}(z)G_{dxi\phi}H_{a}(z)$$
(4.9a)

$$\tilde{d}_{b}(z) = C(z)\tilde{i}_{ref\theta}(z) - C(z)\tilde{d}_{b}(z)G_{di}H_{b}(z) - C(z)\tilde{d}_{a}(z)G_{dxi\theta}H_{b}(z)$$
(4.9b)

$$\tilde{i}_{a}(z) = \tilde{d}_{a}(z)G_{di}(z) + \tilde{d}_{b}(z)G_{dxi\phi}(z)$$
(4.9c)

$$\tilde{i}_{b}(z) = \tilde{d}_{b}(z)G_{di}(z) + \tilde{d}_{a}(z)G_{dxi\theta}(z)$$
(4.9d)

As $i_{ref}(z)$ can be considered to be constant over a sampling period, it is possible to assume $\tilde{i}_{ref}(z) = \tilde{i}_{ref\theta}(z)$. Under this assumption, $\tilde{d}_a(z)$ can be found by solving Equation (4.9a) for $\tilde{d}_b(z)$ and substituting it into Equation (4.9b):

$$\tilde{d}_{a}(z) = C(z)\tilde{i}_{ref}(z) \left\{ \frac{1 + C(z) \left[G_{di} H_{b}(z) - G_{dxi\phi} H_{a}(z) \right]}{\left[C(z) G_{di} H_{a}(z) + 1 \right] \left[C(z) G_{di} H_{b}(z) + 1 \right] - C^{2}(z) G_{dxi\phi} H_{a}(z) G_{dxi\theta} H_{b}(z)} \right\}$$
(4.10a)

In a similar way, solving Equation (4.9b) for $\tilde{d}_a(z)$ and substituting it into Equation (4.9a), $\tilde{d}_b(z)$ is obtained:

$$\tilde{d}_{b}(z) = C(z)\tilde{i}_{ref}(z) \left\{ \frac{1 + C(z) \left[G_{di} H_{a}(z) - G_{dxi\theta} H_{b}(z) \right]}{\left[C(z) G_{di} H_{a}(z) + 1 \right] \left[C(z) G_{di} H_{b}(z) + 1 \right] - C^{2}(z) G_{dxi\theta} H_{a}(z) G_{dxi\theta} H_{b}(z)} \right\}$$
(4.10b)

Finally, substituting Equations (4.10a) and (4.10b) into (4.9c), the closed-loop transfer function $G_{iairef}(z)$ can be derived as:

$$\frac{\tilde{i}_{a}(z)}{\tilde{i}_{ref}(z)} = C(z) \left\{ \frac{G_{di}(z) + G_{dxi\phi}(z) + C(z) \left\{ G_{di}(z) \left[G_{di}H_{b}(z) - G_{dxi\phi}H_{a}(z) \right] + G_{dxi\phi}(z) \left[G_{di}H_{a}(z) - z G_{dxi\theta}H_{b}(z) \right] \right\}}{\left[C(z)G_{di}H_{a}(z) + 1 \right] \left[C(z)G_{di}H_{b}(z) + 1 \right] - C^{2}(z)G_{dxi\phi}H_{a}(z)G_{dxi\theta}H_{b}(z)} \right] \right\} \dots (4.11a)$$

In a similar way, substituting Equations (4.10a) and (4.10b) in (4.9d), $G_{ibiref}(z)$ can be found:

$$\frac{\tilde{i}_{b}(z)}{\tilde{i}_{ref}(z)} = C(z) \left\{ \frac{G_{di}(z) + G_{dxi\theta}(z) + C(z) \left\{ G_{di\theta}(z) \left[G_{di}H_{a}(z) - z G_{dxi\theta}H_{b}(z) \right] + G_{dxi}(z) \left[G_{di}H_{b}(z) - G_{dxi\theta}H_{a}(z) \right] \right\}}{\left[C(z)G_{di}H_{a}(z) + 1 \right] \left[C(z)G_{di}H_{b}(z) + 1 \right] - C^{2}(z)G_{dxi\theta}H_{a}(z)G_{dxi\theta}H_{b}(z)} \right\} \dots$$

...(4.11b)

To further simplify Equations (4.11a) and (4.11b) it can be assumed that the current transducers have unity gain $H_a(s) = H_b(s) = 1$, resulting in:

$$G_{iairef}(z) = \frac{\tilde{i}_{a}(z)}{\tilde{i}_{ref}(z)} = \frac{C(z)G_{di}(z) + C(z)G_{dxi\phi}(z) + C^{2}(z)\left[G_{di}^{2}(z) - G_{dxi\phi}(z)G_{dxi\theta}(z)\right]}{1 + 2C(z)G_{di}(z) + C^{2}(z)\left[G_{di}^{2}(z) - G_{dxi\phi}(z)G_{dxi\theta}(z)\right]}$$
(4.12a)

$$G_{ibiref}(z) = \frac{\tilde{i}_{b}(z)}{\tilde{i}_{ref}(z)} = \frac{C(z)G_{di}(z) + C(z)G_{dxi\theta}(z) + C^{2}(z)\left[G_{di}^{2}(z) - G_{dxi\phi}(z)G_{dxi\theta}(z)\right]}{1 + 2C(z)G_{di}(z) + C^{2}(z)\left[G_{di}^{2}(z) - G_{dxi\phi}(z)G_{dxi\theta}(z)\right]}$$
(4.12b)

Equations (4.12a) and (4.12b) are the closed-loop transfer functions of the DIBC with IPT which account for the interleaved operation of the converter phases, specifically when digital average-current control is employed. The characteristic equation of these transfer functions can be identified as:

$$1+2C(z)G_{di}(z)+C^{2}(z)\left[G_{di}^{2}(z)-G_{dxi\phi}(z)G_{dxi\phi}(z)\right]=0$$
(4.13)

Also, a quick inspection of Equations (4.12a) and (4.12b) reveals that they do not have an obvious open-loop transfer function, making it very difficult to employ methods such as Bode-plots to evaluate the system stability and/or design a compensator.

Following a similar method as that above described, the closed-loop transfer functions may also be determined assuming the synchronous operation of S_a and S_b , resulting in:

$$\frac{\tilde{i}_{o}(z)}{\tilde{i}_{ref}(z)} = \frac{\tilde{i}_{b}(z)}{\tilde{i}_{ref}(z)} = \frac{C(z) [G_{di}(z) + G_{dxi}(z)]}{1 + C(z) [G_{di}(z) + G_{dxi}(z)]}$$
(4.14)

The resulting transfer-functions are greatly simplified and are identical for both phases. Furthermore, Equation (4.14) has the standard structure of a SISO unity-feedback, closed-loop system, where the characteristic equation may be identified as:

$$1 + C(z) [G_{di}(z) + G_{dxi}(z)] = 0$$
(4.16)

and the open-loop transfer function of the system is:

$$T(z) = C(z) [G_{di}(z) + G_{dxi}(z)]$$
(4.17)

This is a significant advantage over the interleaved small-signal model, Equations (4.12a) and (4.12b), as it enables the design of a compensation network and the analysis of the system by the use of standard methods. The derivation of Equation (4.14) can be found in Appendix A.2.

Finally, the closed-loop transfer functions for both the synchronous and the nonsynchronous models may also be derived assuming the normal update operation of the DPWM, or in other words, assuming $t_{d_proc} = T/2$. The transfer functions obtained are particular cases of the immediate update transfer functions and are summarized in Table 4.2, as Equations (4.18a), (4.18b) and (4.19).

Table 4.2. Closed-loop transfer functions of the digital average-current controlled DIBC with IPT inthe z-domain.

Non- Synchronous Samplers (Case I)*	$\frac{\tilde{i}_{a}(z)}{\tilde{i}_{ref}(z)} = \frac{C(z)G_{di}(z) + C(z)G_{dxi\phi}(z) + C^{2}(z)\left[G_{di}^{2}(z) - G_{dxi\phi}(z)G_{dxi\theta}(z)\right]}{1 + 2C(z)G_{di}(z) + C^{2}(z)\left[G_{di}^{2}(z) - G_{dxi\phi}(z)G_{dxi\theta}(z)\right]}$	(4.12a)
	$\frac{\tilde{i}_{b}(z)}{\tilde{i}_{ref}(z)} = \frac{C(z)G_{di}(z) + C(z)G_{dxi\theta}(z) + C^{2}(z)\left[G_{di}^{2}(z) - G_{dxi\theta}(z)G_{dxi\theta}(z)\right]}{1 + 2C(z)G_{di}(z) + C^{2}(z)\left[G_{di}^{2}(z) - G_{dxi\theta}(z)G_{dxi\theta}(z)\right]}$	(4.12b)
Synchronous Samplers (Case II)*	$\frac{\tilde{i}_{a}(z)}{\tilde{i}_{ref}(z)} = \frac{\tilde{i}_{b}(z)}{\tilde{i}_{ref}(z)} = \frac{C(z) [G_{di}(z) + G_{dxi}(z)]}{1 + C(z) [G_{di}(z) + G_{dxi}(z)]}$	(4.14)

Immediately updated DPWM ($t_{d_proc} < T/2$)

Normally updated DPWM ($t_{d_proc} = T/2$)

Non- Synchronous Samplers (Case III)** $\frac{\tilde{i}_{a}(z)}{\tilde{i}_{ref}(z)} = \frac{z^{-1}C(z)G_{di}(z) + z^{-1}C(z)G_{di}(z) + z^{-1}C(z)G_{di}(z) + C(z)G_{di}(z) $	$\frac{\tilde{i}_{a}(z)}{\tilde{i}_{ref}(z)} = \frac{z^{-1}C(z)G_{di}(z) + z^{-1}C(z)G_{dxi\phi}(z) + C^{2}(z)\left[z^{-2}G_{di}^{2}(z) - z^{-1}G_{dxi\phi}(z)G_{dxi\theta}(z)\right]}{1 + 2z^{-1}C(z)G_{di}(z) + C^{2}(z)\left[z^{-2}G_{di}^{2}(z) - z^{-1}G_{dxi\phi}(z)G_{dxi\theta}(z)\right]}$	(4.18a)
	$\frac{\tilde{i}_{b}(z)}{\tilde{i}_{ref}(z)} = \frac{z^{-1}C(z)G_{di}(z) + C(z)G_{dxi\theta}(z) + C^{2}(z)\left[z^{-2}G_{di}^{2}(z) - z^{-1}G_{dxi\theta}(z)G_{dxi\theta}(z)\right]}{1 + 2z^{-1}C(z)G_{di}(z) + C^{2}(z)\left[z^{-2}G_{di}^{2}(z) - z^{-1}G_{dxi\theta}(z)G_{dxi\theta}(z)\right]}$	(4.18b)
Synchronous Samplers (Case IV)**	$\frac{\tilde{i}_{a}(z)}{\tilde{i}_{ref}(z)} = \frac{\tilde{i}_{b}(z)}{\tilde{i}_{ref}(z)} = \frac{z^{-1}C(z)[G_{di}(z) + G_{dxi}(z)]}{1 + z^{-1}C(z)[G_{di}(z) + G_{dxi}(z)]}$	(4.19)
	* Constituent transfer functions as defined in Table 4.1	

* Constituent transfer functions as defined in Table 4.1

** Constituent transfer functions as defined in Table A.1, Appendix A

4.3.4. Implementation of the models in MATLAB

The closed-loop transfer functions of the non-synchronous model, Equations (4.12a) and (4.12b), and the synchronous model, Equation (4.14), were determined numerically employing a script developed in MATLAB. This script can be found in Appendix B.1.

In the first stage of this script, the parameters of the converter and initial conditions of the system are defined. In the second stage, the s-domain control-to-output transfer functions that describe the small-signal dynamics of the converter, Equations (4.1) and (4.2), are determined using the small-signal averaged model of the DIBC with IPT developed in Chapter 2, Section 2.4.1. The resulting transfer functions of this stage are represented as the ratio of two s-domain polynomials with numerical coefficients and are necessary to calculate the constituent transfer functions of the closed-loop model of the converter. In the third stage, the closed-loop transfer functions shown in Table 4.1 are calculated. Finally, in the last stage, the closed-loop transfer functions shown in Table 4.2 are calculated employing the constituent transfer functions previously determined in the third stage. The resulting transfer functions are represented as the ratio of two polynomials in the z-domain with numerical coefficients. The results presented in the following sections were obtained employing this script.

4.4. Validation of the closed-loop models

In this section the closed-loop models are validated and compared in three different ways: using their transient response to small step-reference changes, considering the response to small sinusoidal disturbances and finally by examining the stability range predicted by the models when a proportional-integral compensator is employed. To this end the small-signal models introduced in the previous section were implemented in MATLAB, and the results are compared against those from the simulation of both the switched and the averaged SABER models. The results presented were obtained assuming that the immediate-update control algorithm, described in Chapter 3, is used. The MATLAB scripts employed can be found in Appendix B. The SABER models employed are those previously introduced in Chapter 3, Section 3.7, which were validated using the converter prototype.

4.4.1. Transient response

Figure 4.5 shows the transient response of the models to a 5% step increase in the reference-current of the digital control-loops. The system conditions employed were $V_{in} = 80 V$ and $R_{load} = 5.2 \Omega$ and the PI compensator gains used are $K_p = 30T$ and $K_i = 10$ which are identical to those employed for the experimental results shown in Chapter 3 and were selected using the simplified non-interleaved model of the converter described by Equations (4.14) and in conjunction with SISO tool in MATLAB to guarantee the stability of the converter. An appropriate design methodology for this PI gains is later described in Chapter 5 of this Thesis. The graphs are divided into three groups. Figure 4.5(a) shows the results obtained for D = 0.25, Figure 4.5(b) for D = 0.5, and finally Figure 4.5(c) for D = 0.75. In each group, two sets of three graphs are shown. The top set provides an overview of the complete transient whilst the bottom set shows a time expansion at the start of the transient.

The parameters of the converter and the DSC unit employed in the SABER simulations are listed in Table 4.3. The transient responses of the mathematical models were obtained by applying the MATLAB *step()* command to the numerical solution of the transfer functions determined with the MATLAB script of Appendix B.1.

Input inductance at 60 kHz 5.12 µH			
IPT windings self-inductance at 30kHz	75.14 μΗ		
Coupling coefficient of the IPT windings	0.997		
Output capacitance	45 μF		
Load Resistance	2.8 Ω, 5.2 Ω		
Input Voltage	80 V		
Transistor turn-on/off delay*	520 ns		
Switching frequency	itching frequency 30 kHz		
Digital Signal Controller			
Master clock frequency*	150 Mhz		
ADC sampling delay*	140 ns phase-a, 220 ns phase-b		
Compensation network	Digital PI, Backward-Euler implementation		

Power Converter

* SABER switched model only.

Modelling phase interactions in dual interleaved converters



177

The waveform shown for the comparison of the models is the sampled phase-a current, $i_a*(t)$. This current is shown instead of $i_a(t)$ as the large ripple of the SABER switched model would obscure the details of the response. In the first graph of each set, the transient-response of the SABER switched model is compared to that obtained from the SABER averaged model. The second graph depicts the transient-response of the SABER small-signal synchronous model, Equation (4.14), which is compared against that of the SABER averaged model. Finally, in the last graph, the transient response of the non-synchronous model, Equation (4.12a), is superimposed on that of the SABER averaged model.

The results of the SABER switched model show bursts of random looking oscillations throughout the waveforms, however these oscillations have no more than *1 A* of amplitude. This behaviour was attributed to limit-cycle oscillations, LCOs, caused by quantization effects within the control loops, in particular the interaction produced by the resolutions of the ADC and the DPWM generators [4,5]. This was confirmed by increasing the resolution of the DPWM modules by a factor of four in the SABER switched model and the step response was re-simulated. The results from the SABER switched model employing a clock frequency of *150 MHz* and a clock frequency of *600 MHz* are superimposed in the graph of Figure 4.6, where the amplitude of the oscillations is reduced by about *75 %*.



Transient response

Figure 4.6. Comparison of transient responses of the SABER switched model using a 150 MHz DPWM clock or a resolution of 6.66 ns and using a 600 MHz DPWM clock or a resolution of 1.66 ns. V_{in} =80 V, R_{load} = 5.2 Ω , I_{ref} = 30.7 A, D = 0.5. Immediate update mode, $t_{d \ proc}$ = (0.21)T = 4.2 μ s.

Going back to the results of Figure 4.5, the SABER averaged model follows closely the transient oscillations of the SABER switched model. The difference in the frequency of the oscillations is thought to be caused by parameters not considered in the averaged modelling such as non-zero switching times of the transistors and quantization effects of the ADC and PWM module. A difference in the steady-state value of the SABER averaged model and the SABER switched model of *0.1 A* can be noticed in Figure 4.5(a), this might be caused by the presence of LCOs. For the other cases examined, the difference in the steady-state value is negligible compared to the level of current.

The results obtained for the synchronous and non-synchronous models depicted for all the cases, correspond closely to the results of the SABER averaged model. No significant difference was noticed in the results. However, a close-inspection of the time-expansions reveals that the non-synchronous model predicts with more accuracy the amplitude of the sampled current, but the difference is not very significant.

4.4.2. Frequency response to small sinusoidal perturbations

In this section, the frequency response of the small-signal synchronous model, Equation (4.14), and the small-signal non-synchronous model, Equation (4.12a) and (4.12b), are compared against that of the SABER averaged model. The closed-loop transfer functions were calculated using the MATLAB program in Appendix B.1. The system parameters used in the MATLAB script are shown in Table 4.3.

In order to undertake this analysis in SABER, the Time Domain System Analyser tool, TDSA, was employed. This tool emulates the operation of a network analyser allowing the user to measure the magnitude and phase response of a system to small-sinusoidal perturbations. The parameters used to obtain the results for the SABER averaged model are listed in Table 4.4. The signal generator of the TDSA was connected to the reference-current input of the control loops whilst the phase-a current was converted into a voltage signal and fed-back into the measuring port of the TDSA tool. The parameters of the converter and the DSC unit in SABER are as shown in Table 4.3.

Parameter Name	Value
fbegin	10
fend	20k
ampl	1
Offset	200
npoints	logswp
Mode	0.01
 Max_err	0.01
 Max_nper	100
Min_nper	5
Min_tspp	100
aO	1
q0	80
Filter	yes
ac	mag=1, phase=0

Table 4.4. Parameters of the TDSA tool employed in SABER to obtain the frequency response of
the model to small-sinusoidal perturbations.

Figures 4.7(a), 4.7(b) and 4.7(c) show the magnitude and phase response of the closed-loop transfer functions in the frequency domain for the points of operation D = 0.25, D = 0.5 and D = 0.75 from 10 Hz to 20 kHz. The system conditions employed were $V_{in} = 80$ V, $R_{load} = 5.2 \ \Omega$ and the DPWM modules were updated in normal mode, in other words, $t_{d_proc} = (0.5)T$. Furthermore the PI controller gains were $K_p = 30T$ and $K_i = 10$. The results shown are only valid up to half of the switching frequency of the converter, or 15 kHz, as stated by Nyquist.

In the first two graphs of Figures 4.7(a), 4.7(b) and 4.7(c), the magnitude and phase response of the SABER averaged model and the small-signal synchronous model, Equation (4.14), are compared. In the third and fourth graphs the magnitude and phase response of the phase-a current obtained with the SABER average model and the small-signal non-synchronous model, Equation (4.11a), are plotted. Finally, in the fifth and sixth graphs, the magnitude and phase response of the phase-b current obtained with the SABER averaged model and the small-signal non-synchronous model, Equation (4.11b), are shown.

The magnitude response of the small-signal synchronous model and the small-signal non-synchronous model show a close correspondence with the response of the SABER averaged model before the resonant peak. The maximum absolute difference between the


Chapter 4

Modelling phase interactions in dual interleaved converters

signals in this range was found to be no more of *3 dBA* approximately for the phase-a current in all the range of operation. However, the magnitude of the phase-b current exhibits a drop of approximately *6 dBA* between *250 Hz* to *1 kHz* which is thought to be due to the interleaved interaction of the converter phases.

The magnitude and location of the resonant-peak frequency was measured for all the models and is presented in Table 4.5. As seen in the measurements, both the magnitude and the location of the resonant-peak are in close agreement for all the models. Similarly, the phase response of all the models below the resonant-peak frequency shows a close agreement with a difference of no more than *10 degrees*.

Table 4.5. Comparison of the magnitude and frequency location of the resonant peak in the frequency-response of the SABER averaged model, the small-signal synchronous model and the small-signal non-synchronous models.

Doint of Operation	SABER averaged model	Small-signal non- synchronous model			
	Magnitude [dBA]				
<i>D = 0.25,</i> Figure 4.7(a)	2.09	3.43			
<i>D = 0.5,</i> Figure 4.7(b)	3.50 4.30		4.83		
<i>D = 0.75,</i> Figure 4.7(c)	11.42 9.25		10.10		
	Frequency [kHz]				
<i>D = 0.25,</i> Figure 4.7(a)	8.63	8.59	8.59		
<i>D = 0.5,</i> Figure 4.7(b)	6.86 6.76		6.76		
<i>D</i> = 0.75, Figure 4.7(c)	6.60 6.42		6.69		

Some of the differences observed in the synchronous and the non-synchronous models might be also produced by the discretization method employed to obtain the z-domain transfer functions of the model. The method employed was the ZOH method, which is known to discretize a system with high accuracy in the time-domain only for stair-case inputs [1,2]. Other methodologies exist, such as the Tustin approximation or the Zero-Pole Matching method, which can be employed to obtain a better approximation of the response of the system in the frequency domain.

For frequencies above the resonant frequency, the magnitude of the small-signal synchronous model drops with a steeper slope than the SABER averaged model in all the points of operation considered. Meanwhile, the magnitude of the non-synchronous model corresponds closely to the magnitude of the SABER averaged model. Conversely, the phase response of the synchronous model shows close agreement with the predictions of the SABER averaged model, whilst the non-synchronous model shows an increased phase-shift in the response of phase-a and a reduced phase shift in the response of phase-b. This is attributed to the time-delay and the time-advance units employed to model the interleaved sampling operation of the converter phases.

4.4.3. Stability range with a digital PI compensator

In this section the stability of the transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ obtained using the small-signal synchronous model, Equation (4.14), and the small-signal nonsynchronous model, Equations (4.11a) and (4.11b), is compared when the control-loops are compensated with a digital PI controller. The MATLAB script found in Appendix B.2 was used to determine the stability of $G_{iairef}(z)$ and $G_{ibiref}(z)$ at a specific point of operation by evaluating the location of the poles for several combinations of proportional gain, K_p , and integral gain, K_i . For each iteration the data obtained was stored in a file together with the corresponding values of K_p and K_i and was plotted using ORIGINLAB. The analysis was undertaken for several points of operation to ensure the consistency of the data and the predictions of the small-signal models.

The graphs depicted in Figures 4.8 and 4.9 summarize the data acquired for four of the different points of operation examined employing the normal-update mode of the DPWM modules, in other words, $t_{d_proc} = 0.5(T)$. Once more, the parameters of the converter are listed in Table 4.3. In the graphs, the results of the synchronous model are plotted in green whilst the ones corresponding to the non-synchronous model are plotted in orange. The stability range in these graphs is the shadowed region bound by the maximum values of K_p and K_i for which the system poles are stable and the minimum K_p and K_i values that were tested ($K_p = 0.005(T)$ and $K_i = 0.01$). To map out the stability region K_p and K_i were increased logarithmically up to 100(T) and 100 respectively.



Figure 4.8. Stability-range predicted by the synchronous model and the non-synchronous model of the converter employing a digital PI controller. Normal update mode, $t_{d_proc} = (0.5)T = 16.66 \ \mu s$. (a) $R_{load} = 5.2 \ \Omega$, D = 0.25, $I_{in} = 27.34 \ A$, $V_{in} = 80 \ V$; (b) $R_{load} = 5.2 \ \Omega$, D = 0.85, $I_{in} = 246 \ A$, $V_{in} = 80 \ V$.

Figures 4.8(a) and 4.8(b) depict the results obtained employing $V_{in} = 80$ V and $R_{load} = 5.2 \Omega$ for the points of operation D = 0.25 and D = 0.85. These results show the evolution of the stability range as the duty ratio is increased. Similarly, the graphs depicted in Figures 4.9(a) and 4.9(b) depict the results obtained using the conditions $V_{in} = 80$ V and $R_{load} = 2.8$



Figure 4.9. Stability-range predicted by the synchronous model and the non-synchronous model of the converter employing a digital PI controller. Normal update mode, $t_{d_proc} = (0.5)T = 16.66 \ \mu s$. (a) $R_{load} = 2.8 \ \Omega$, D = 0.25, $I_{in} = 50.6 \ A$, $V_{in} = 80 \ V$; (b) $R_{load} = 2.8 \ \Omega$, D = 0.85, $I_{in} = 457 \ A$, $V_{in} = 80 \ V$.

 Ω for D = 0.25 and D = 0.85. From the graphs, it is evident that there is a clear difference in the lower stability limit predicted by the models. The K_p and K_i stability region is approximately rectangular using the synchronous model, whereas the region is almost halved with the non-synchronous model. In particular with low values of K_p and higher

Chapter 4

values of K_i the synchronous model suggests the system will be stable whereas the nonsynchronous model predicts the opposite. The results also show that the $K_p \setminus K_i$ stability region reduces at higher duty-ratio, but is only slightly affected by a reduced load resistance.

Figure 4.10, shows a similar graph to those found in Figure 4.8 and 4.9. In this graph, the stability range of the non-synchronous model with immediate update mode, $t_{d_proc} = 0.1(T)$, is compared against the one with half-cycle delayed update, $t_{d_proc} = 0.5(T)$. The point of operation where this graph was obtained is for the conditions $V_{in} = 85 V$, $R_{load} = 5.2 \Omega$, D = 0.5480, $I_{in} = 80 A$, which was chosen as an example amongst several other points of operation that were evaluated as a similar pattern was observed. As seen from the graph, the stability range of the system is significantly increased when the PWM module is immediately updated after the duty ratio calculations.



Figure 4.10. Comparison of the stability-range predicted by the non-synchronous model with immediate update mode, $t_{d_proc} = (0.1)T = 3.33 \ \mu s$, and the small-signal non-synchronous model with normal update mode, $t_{d_proc} = (0.5)T = 16.66 \ \mu s$. Compensator: digital PI controller, Backward-Euler implementation. $V_{in} = 85 \ V$; $R_{load} = 5.2 \ \Omega$, D = 0.5480, $I_{in} = 80 \ A$.

To check the stability limit predicted by the small-signal non-synchronous model, a number of points around the parameter combinations, denoted points *a-e* and *a1-e1* in

Chapter 4

Figure 4.10, were examined using the SABER averaged model. To run these simulations, the system was first taken to the desired point of operation with $K_p = 30(T)$ and $K_i = 20$. Once the system reached the steady state, K_p and K_i were switched to the values required and the system was perturbed employing a 5% step-increment in the reference-current. For every point identified in the graphs of Figure 4.10, two points in the surrounding points were selected and simulated.

The points evaluated are summarized in Table 4.6 and the simulation results obtained for each of the points of operation can be found in Figures E.1 to E.4 of Appendix E. As seen in Table 4.6, the predictions of the non-synchronous model are in agreement with the stability range observed using the SABER averaged model. The small difference in the predictions was attributed to the size of the step in the proportional gain when the $K_p \setminus K_i$ stability region graphs were generated.

Non-synchronous model <i>(K_i, K_P(T))</i>	SABER averaged model (K _i , K _p (T))		
a(1,0.55(T))	(1,0.7(T)) – Stable	(1,0.4(T)) - Unstable	
b(10,5.04(T))	(10,6(T)) - Stable	(10,4(T)) - Unstable	
c(25.22,20(T))	(24,20(T)) - Stable	(26,20(T)) - Unstable	
d(4,57.9(T))	(4,60(T)) - Unstable	(4,56(T)) - Stable	
e(0.2,60.2(T))	(0.2,63(T)) - Unstable	(0.2,58(T)) - Stable	
a1(2,0.2(T))	(2,0.4(T)) - Stable	(2,0.2(T)) - Unstable	
b1(30,3.17(T))	(30,4(T)) - Stable	(30,2(T)) - Unstable	
c1(133.8,45(T))	(131,45(T)) - Stable	(135,45(T)) - Unstable	
d1(60,122.16(T))	(60,124(T)) - Unstable	(60,121(T)) - Stable	
e1(1,159.3(T))	(1,163(T)) - Unstable	(1,157(T)) - Stable	

Table 4.6. Comparison of the stability limits predicted by the non-synchronous model and theSABER averaged model. (Figure 4.10).

4.4.4. Experimental validation of the stability range

To evaluate experimentally the stability region of the converter, the system described in Section 3.7, Chapter 3, was started-up with a programmed reference phasecurrent of 40 A and with a set of initial values for K_p and K_i . The initial values for K_p and K_i were 10(T) and 1 for the normal update algorithm and 12.5(T) and 1 for the immediate update algorithm. Once the converter reached the steady-state operation, the initial values of K_p and K_i were replaced by the required values for the evaluation.

The DSC was programmed to evaluate six different discrete values of K_p per experiment, which were selected employing the potentiometer that in previous experiments was used to select the reference-current. To range K_i the DSC had to be reprogrammed. The additional code to generate this program created an extra delay in the processing time which will have an impact on the stability range of the immediate update algorithm. The t_{d_proc} measured was of 4.9 µs, or 0.147(T), compared with the 4.2 µs measured in Section 3.7.2, Chapter 3, when no extra-tasks other than controlling the converter phase-currents are undertaken by the DSC.

Figure 4.11(a) and 4.11(b) depict a comparison of the stability range predicted by the small-signal non-synchronous model and the stability limit observed using the converter prototype. Figure 4.11(a) corresponds to the normal update algorithm whilst Figure 4.11(b) corresponds to the immediate update algorithm. The point of operation evaluated corresponds to $V_{in} = 85 V$, $R_{load} = 5.2 \Omega$, D = 0.5480 and $I_{in} = 80 A$, identical to that shown in the graphs of Section 4.3.3. The green shadowed area corresponds to the stability range predicted by the small-signal non-synchronous model, whilst the blue data-points represent the combination of K_p and K_i for which the practical system was stable and the red data-points correspond to the ones where instability was observed. To provide evidence of the experimental results, the highlighted points in the graphs were captured and included in Appendix E, Figures E.5 to E.8. In the experimental results shown, for any value of K_p that was found to be stable and the first where instability was observed.

The discrepancies observed might be attributed to several practical factors that were disregarded in the modelling procedure, such as the quantization non-linearity and limited resolution of both the ADC and the DPWM modules, non-zero switching transitions and non-linear behaviour of the switching devices, variations in the parameters of the current transducers and the components used to measure and condition the feedback signals, and small discrepancies in the parameters of the components integrating the converter phases, amongst others.



Figure 4.11. Comparison of the stability-range predicted by the non-synchronous model against the experimental stability-range measured with the converter prototype. (a) Normal update mode $t_{d_proc} = (0.5)T = 16.66 \ \mu s$ (b) Immediate update mode, $t_{d_proc} = (0.147)T = 4.9 \ \mu$, D = 0.5480, $R_{load} = 5.2 \ \Omega$, $V_{in} = 85 \ V$, $I_{in} = 80 \ A$.

Chapter 4

4.5. Summary

The sampler decomposition technique was used to model the interleaved sampling operation of the current feedback loops in an average-current controlled, dual-interleaved converter. The small-signal dynamics of the converter were represented using the small-signal averaged model developed in Chapter 2. The closed-loop, transfer functions that relate small-changes in the reference-current to small-changes in the phase-currents were derived for two cases: when the interleaved sampling is disregarded, synchronous model, and when the interleaved sampling is modelled. It was found that the transfer-functions for each phase are identical and greatly simplified when the interleaved sampling is disregarded. Conversely, when the interleaved sampling was taken into account, different transfer functions were found for each phase of the converter.

A comparison between the closed-loop transfer functions from the synchronous and the non-synchronous model was undertaken using time domain step-responses and frequency domain responses. Furthermore, to validate the models developed, these results were compared with simulation results from a switched and a large-signal averaged model implemented in SABER simulator.

The transient response of the models to 5 % disturbances, showed an excellent agreement with the SABER simulations and the mathematical models showed no significant difference between their predictions. In the frequency domain, both models showed a good agreement with the simulation results presenting a deviation of no more than approximately 1 dBA in the magnitude response over most of the operating range, but increasing at low power, and no more than 10 degrees in the phase response for frequencies up to approximately 8 kHz. However, for frequencies above 8 kHz the transfer-functions of the non-synchronous model showed differences in the phase response, which are likely to be caused by the time-advance and time-delay units employed to model the interleaved interaction of the control-loops.

The stability-range comparison confirmed that the interleaved operation of the converter phases has a detrimental effect on the stability of the system and that the synchronous model fails to predict this behaviour. The stability range predicted by the non-

synchronous model was further validated by simulation and experimental results, showing an excellent agreement with the simulation results from the SABER averaged model and a good agreement with the pattern of stability observed in the experimental results.

4.6. References

- [1] B. C. Kuo, *Digital control systems*, 2nd ed. USA: Oxford University Press, Inc., 1992.
- [2] J. T. Tou, *Digital and sampled-data control systems*, USA: McGraw-Hill, 1959.
- [3] D. M. VandeSype, K. DeGusseme, F. M. L. L. DeBelie, A. P. VandenBossche, and J. A. Melkebeek, "Small-Signal z-Domain Analysis of Digitally Controlled Converters," *Power Electronics, IEEE Transactions on*, vol. 21, pp. 470-478, 2006.
- [4] S. Buso and P. Mattavavelli, *Digital Control in Power Electronics*. USA: Morgan & Claypool, 2006.
- [5] A. V. Peterchev and S. R. Sanders, "Quantization resolution and limit cycling in digitally controlled PWM converters," *Power Electronics, IEEE Transactions on*, vol. 18, pp. 301-308, 2003.

Chapter 5 Digital control loop: dynamics and design

5.1. Introduction

In the first section of this Chapter, the small-signal model of the DIBC developed in Chapter 4 is used to provide further insight into the system dynamics and stability limits by examining the loci of the system poles as the main parameters are changed. Several rootloci were generated with the aid of MATLAB to illustrate the trajectories of the poles when the quiescent duty-ratio is varied from 0.2 to 0.8 in steps of 0.05. The sensitivity of the trajectories to the main system parameters are then examined in turn.

In the second section, the parameters of the digital PI compensator employed to regulate the system are selected based on the models and theory developed in this thesis. The objective was to ensure stability and minimum settling time to step-responses over a defined range of operation. The design exercise is performed using the small-signal non-synchronous model with immediate DPWM update developed in Chapter 4.

5.2. Sensitivity analysis via Root-locus

In this section, the sensitivity of the pole trajectories from the small-signal nonsynchronous model developed in Chapter 4, Equations (4.12a) and (4.12b), to parameter variations is discussed. The root-loci generated are organized as shown in Table 5.1.

5.2.1. Varying load resistance with a proportional compensator

In Figure 5.1, the trajectories of the system poles are depicted when the duty-ratio and the load resistance are varied from 0.2 to 0.8 and from 1.4 Ω to 10.4 Ω respectively.

	Section	Figure	Duty-ratio D	Input voltage <i>V</i> in	Load resistance <i>R_{load}</i>	Proportional gain <i>K_p</i>	Integral gain <i>K</i> i	Processing time t _{d_proc}
rtional	5.2.1	5.1	0.2 -> 0.8	80 V	1.4Ω, 2.6Ω, 5.2Ω, 10.4Ω	1(T)	-	0.5(T)
Propo	5.2.2	5.2	0.2 -> 0.8	80 V	5.2 Ω	1(T), 5(T), 10(T), 20(T), 30(T), 40(T)	-	0.5(T)
egral	5.2.3	5.3	0.2 -> 0.8	80 V	5.2 Ω	1(T), 5(T), 10(T), 20(T), 30(T), 40(T)	1 10 20	0.5(T)
l plus Inte	5.2.4	5.4	0.2 -> 0.8	80 V	10.4 Ω 5.2 Ω 2.6 Ω	1(T), 20(T), 40(T)	10	0.5(T)
ortiona	5.2.5	5.5	0.2 -> 0.8	80 V	5.2 Ω	20(T)	10	0.1(T), 0.3(T), 0.5(T)
Propo	5.2.6	5.6	0.2 -> 0.8	50 V 120 V	5.2 Ω	1(T), 20(T),40(T)	10	0.5(T)

Table 5.1.	Root-locus plots generated to analyse the behaviour of the closed-loop poles of the
	small-signal non-synchronous model.



Figure 5.1. Root-locus of the poles of closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ of the smallsignal non-synchronous model. Compensator: digital proportional controller with $K_p = 1T$. Ranging D from 0.2 to 0.8 and R_{ioad} from 1.4 Ω to 10.4 Ω . $V_{in} = 80$ V.

A proportional compensator of $K_p = 1T$ was used throughout in both phases. The root-locus comprises a pair of conjugated complex poles and three real-axis poles: one close to the +1 point and two at the origin. As seen in the figure, the trajectory of the complex-conjugated poles moves around the unitary circle and towards the real positive axis as the duty ratio is increased, indicating a reducing natural frequency in the time-domain. If the load resistance is decreased, the trajectories are displaced concentrically towards the centre of the unit circle, increasing the damping of the poles. The three real-axis poles showed no significant movement over the range of the parameter variations.

5.2.2. Varying proportional gain of the proportional compensator



Figure 5.2. Root-locus of the poles of closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ of the smallsignal non-synchronous model. Compensator: digital proportional controller. D = 0.2 to 0.8; $K_p = 0.1(T)$, 1(T), 5(T), 10(T), 20(T), 30(T); $R_{load} = 5.2 \Omega$; $V_{in} = 80 V$.

Figure 5.2 depicts the root-locus of the system poles when the proportional gain is increased from 0.1(T) to 30(T). In this plot, the load resistance was fixed to a value 5.2Ω . As seen in the plot, the trajectory of the complex poles curves initially towards the centre of the unit circle and then turns back on itself as K_p is increased. In addition, the real-axis

poles at the origin move to the right whilst the pole at the +1 point moves to the left as K_p is increased.

5.2.3. Varying proportional gain and integral gain of the PI compensator

In the following sections the root-locus of the closed-loop system poles are examined when a digital proportional-integral compensator is employed to regulate the system. Similar to the previous sections, the group of plots in Figures 5.3 show the trajectories of the poles when the duty ratio is increased from 0.2 to 0.8 by 0.05 plus the evolution of these trajectories when K_p is varied from 1(T) to 40(T) and when K_i is equal to 1, 5 and 10. For each value of K_i examined, a general view of the unit circle and a magnified view of the poles located close to the +1 point of the real axis are shown. Figures 5.3(a) and (b) depict the root-loci when the integral gain is fixed to $K_i = 1$; Figures 5.3(c) and (d) correspond to $K_i = 5$; and finally, Figures 5.3(e) and (f) correspond to $K_i = 10$.

The introduction of the PI compensator in the system generates an extra-pair of complex poles close to the real-axis and in the vicinity of the +1 point. These poles will be referred to as the lower frequency complex conjugated poles from now on to distinguish them from the higher frequency pair of complex poles. It was found that the lower frequency complex poles determine the minimum K_p stability boundary that was observed in the stability-range graphs of Section 4.4.3. As the duty-ratio is increased the poles tend to move around the unit circle and away from the real-axis. Also in some cases, the trajectories of these poles lie outside the unit circle for small-values of K_p and move inwards as K_p is increased.

The movement of the higher frequency complex poles is similar to that observed in Figure 5.2, however as the duty-ratio increases the locus of some poles turns back on itself more sharply and moves outside the unit circle. This effect is stronger with larger values of K_i , Figures 5.3(c) and 5.3(e). The maximum K_p stability limit observed in the stability-range graphs of Section 4.4.3 was found to be directly related to these poles.

In summary, two main conclusions can be drawn, first, the maximum K_p stability limit which was identified in the stability-range plots of Section 4.4.3, is seen to be determined by the pair of high-frequency complex poles, and second, the minimum K_p stability limit is



Digital control loop: dynamics and design



Figure 5.3. Root-locus of the poles of closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ of the smallsignal non-synchronous model. Compensator: digital PI controller. D = 0.2 to 0.8; $K_p = 1(T)$, 5(T), 10(T), 20(T), 30(T), 40(T); $R_{load} = 5.2 \Omega$; $V_{in} = 80 V$. (a)&(b) $K_i = 1$; (c)&(d) $K_i = 5$; (e)&(f) $K_i = 10$.



Digital control loop: dynamics and design



Figure 5.4. Root-locus of the poles of closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ of the smallsignal non-synchronous model. Compensator: digital PI controller. D = 0.2 to 0.8; $K_p = 1(T)$, 20(T) and 40(T); $K_i = 10$; $V_{in} = 80$ V. (a)&(b) $R_{load} = 10.4 \Omega$; (c)&(d) $R_{load} = 5.2 \Omega$; (e)&(f) $R_{load} = 2.6 \Omega$.

mainly determined by the pair of lower frequency complex poles which seem to be introduced by the PI compensator.

5.2.4. Varying load resistance with a PI compensator

Figure 5.4 depicts the trajectories of the system poles when *D* is increased from 0.2 to 0.8 in steps of 0.05 for different values of load resistance. The figures are arranged in a similar way to those presented in the previous section and they show a general view of the unit circle plus a close-up in the region of the +1 point. For each value of resistance examined, the proportional gain was set to 1(T), 20(T) and 40(T). Figure 5.4(a) and (b) depict the trajectories for $R_{load} = 10.4 \Omega$; Figure 5.4(c) and 5.4(d) correspond to $R_{load} = 5.2 \Omega$; and Figure 5.4(e) and 5.4(f) correspond to $R_{load} = 2.6 \Omega$. The integral gain was kept constant at $K_i = 10$ throughout.

The results show that the low frequency complex poles move outside the unit circle for the lowest value of K_p whereas the high frequency complex poles turn towards the unit circle and tend to cross over the circle as *D* increases, with the effect becoming more pronounced with lower values of load resistance, Figure 5.4(e).

5.2.5. Varying PWM update delay with a PI compensator

Figure 5.5 illustrates the root-locus of the system poles when t_{d_proc} is reduced from 0.5(T), normal update mode of the DPWM, to 0.3(T) and 0.1(T), immediate update mode of the DPWM. The point of operation selected to illustrate the pole trajectories was $R_{load} = 5.2$ Ω , $V_{in} = 80 V$, $K_p = 20(T)$ and $K_i = 10$ with D ranging from 0.2 to 0.8 by 0.05.

The plot of Figure 5.5(a) shows that a reduction of t_{d_proc} has a favourable impact on the position of the complex poles. The trajectories of the high frequency complex poles as *D* is increased turn towards the centre of the unit circle. In addition, the trajectories of the lower frequency complex poles also move further into the unit circle, Figure 5.5(b), although the movement is not as great as seen for the higher frequency poles. Once more, these results further illustrate the increase in stability-range depicted in the graph of Figure 4.10 with immediate update mode, Chapter 4, Section 4.4.3.



Figure 5.5. Root-locus of the poles of closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ of the smallsignal non-synchronous model. Compensator: digital PI controller. D = 0.2 to 0.8; $t_{d_proc} = 0.1(T)$, 0.3(T), 0.5(T); $K_p = 20(T)$; $K_i = 10$; $R_{ioad} = 5.2 \Omega$; $V_{in} = 80 V$.

5.2.6. Varying input voltage with a PI compensator

Figure 5.6 illustrates the system pole trajectories for different values of input voltage. Figures 5.6(a) and 5.6(b) correspond to $V_{in} = 50 V$ whilst Figures 5.6(c) and 5.6(d) correspond to $V_{in} = 120 V$. Figures 5.4(c) and 5.4(d) can be used as a reference point when $V_{in} = 80 V$. The fixed conditions that were used to obtain these plots are $R_{load} = 5.2 \Omega$, $V_{in} = 80 V$, $K_i = 10$. The trajectories of the poles are illustrated for values of K_p equal to 1(T), 20(T) and 40(T). From these graphs, it can be concluded that the stability-range of the system is reduced as the input voltage is increased since the trajectories of the poles tend to move outside the unitary circle. However, consistent with what has been seen previously, the lower frequency complex poles remain outside the unit circle for the lowest value of K_p , $K_p = 1(T)$.



Figure 5.6. Root-locus of the poles of the closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ of the smallsignal non-synchronous model. Compensator: digital PI controller. D = 0.2 to 0.8; $K_p = 1(T)$, 20(T) and 40(T); $K_i = 10$; $R_{load} = 5.2 \Omega$. (a)&(b) $V_{in} = 50 V$; (c)&(d) $V_{in} = 120 V$.

5.3. Design of the PI compensator for digital average-current control with immediate update mode

The design method is based on the $K_p \setminus K_i$ stability-range plots introduced in Section 4.4.3, Chapter 4, as they offer an effective way to visualize the stability range of the system with a compensator. The transient performance of the system was evaluated using step responses generated from the non-synchronous model implemented in MATLAB.

The range of operation of the system, Table 5.2, was selected according to the maximum operating conditions of the converter prototype and the maximum ratings of the laboratory equipment that was readily available. In order to identify the necessary $K_p \setminus K_i$ plots to proceed with the design, the conditions of the system at the extremes of the operation range were identified and calculated. Table 5.3 summarizes these conditions.

 Table 5.2. Range of operation of the converter for the design of the digital PI controller.

	Input Voltage	Input Current	Input Power	Output Voltage
Minimum	50 V	60 A	3 kW	125 V
Maximum	100 V	120 A	12 kW	250 V

Table 5.3. Extremes of operation for the design of the digital PI controller.

Parameter Name	Minimum input voltage/ Minimum input Current	Maximum input voltage/ Minimum input current	
Input Voltage, V _{in}	50 V	100 V	
Input Current, I _{in}	60 A	60 A	
Load Resistance, R _{load}	5.2 Ω	5.2 Ω	
Output Power, Po	3000 W	6000 W	
Output Voltage, V _o	125 V	176 V	
Duty-ratio, D	0.6	0.43	

Parameter Name	Minimum input voltage/ Maximum input current	Maximum input voltage/ Maximum input current	
Input Voltage, V _{in}	50 V	100 V	
Input Current, I _{in}	120 A	120 A	
Load Resistance, R _{load}	5.2 Ω	5.2 Ω	
Output Power, Po	6000 W	12000 W	
Output Voltage, V _o	172 V	250 V	
Duty-ratio, D	0.71	0.6	

The $K_p \setminus K_i$ stability-range plots of the converter were generated and compared for the four points of operation listed in Table 5.3, using the converter parameters shown in Table 5.4 and with the system operated in immediate update mode assuming a processing time of $t_{d_proc} = 4.12 \ \mu s$. The value of t_{d_proc} corresponds to the value measured using the converter prototype in Section 3.7.2, Chapter 3. The comparison revealed that the stabilityrange of the system at maximum power, $12 \ kW$, was enclosed by the stability-range of the other points of operation evaluated, indicating that the combination of maximum input voltage and maximum input current (maximum output power) yield the worst case of stability. Figure 5.7 illustrates the $K_p \setminus K_i$ stability-range for this point of operation, where the highlighted points, A1 - D3, show the combinations of proportional and integral gains chosen to evaluate the step performance of the system.



Figure 5.7. Stability-range predicted by the non-synchronous model for the design of the PI compensator. Immediate update mode, $t_{d_{proc}} = (0.126)T = 4.12 \ \mu s$. D = 0.599, $R_{load} = 5.2 \ \Omega$, $V_{in} = 100 \ V$, $I_{in} = 120 \ A$. (Worst case scenario for stability of the design).

Table 5.4. Attributes of the converter and the DSC employed for the simulations.

Input inductance at 60 kHz	5.12 μΗ
IPT windings self-inductance at 30kHz	75.14 μΗ
Coupling coefficient of the IPT windings	0.997
Output capacitance	45 μF
Transistor turn-on/off delay*	520 ns
Switching frequency	30 kHz

Power Converter

Digital Signal Controller

Master clock frequency*	150 Mhz	
ADC sampling delay*	140 ns phase-a, 220 ns phase-b	
Compensation network	Digital PI, Backward-Euler implementation	

* SABER switched model only.

The results shown in Figures 5.8 and 5.9 show the transient response of the converter phase currents to a unit step-increment in the reference-current for the combinations *A1* to *D3* highlighted in Figure 5.7. In these figures an overview of the response is accompanied by a magnified view which details the initial oscillatory behaviour of the responses. The converter parameters used in MATLAB are shown in Table 5.4 and the point of operation employed for the evaluation corresponds to the maximum input voltage/maximum input current combination shown in Table 5.3.

Figures 5.8(a) and 5.8(b) depict the transient response when $K_i = 5$ and $K_p = 5(T)$, 30(T) and 70(T), points A1 to A3 in Figure 5.7. Figures 5.8(c) and 5.8(d) correspond to $K_i =$ 10 and $K_p = 10(T)$, 40(T) and 70(T), points B1 to B3 in Figure 5.7. Figures 5.8(e) and 5.8(f) correspond to $K_i = 10$ and $K_p = 20(T)$, 50(T) and 70(T), points D1 to D3 in Figure 5.7. Finally, Figures 5.9(a) and 5.9(b) correspond to $K_i = 10$ and $K_p = 30(T)$, 45(T) and 60(T), points E1 to E3 in Figure 5.7. In addition, Table 5.5 summarizes the maximum over-shoot and settling time measured for all the results depicted.

As expected from the root-locus analysis, two natural frequencies are evident in the responses, a lower frequency component which is related to the pair of complex z-domain poles close to the +1 point and a higher frequency due to the second pair of complex poles. The lower frequency oscillation becomes more pronounced with lower values of K_p whilst the higher frequency component becomes more obvious with larger values of K_p .



Figure 5.8. Transient-response of the closed-loop transfer function $G_{iairef}(z)$ to a 1 A step-increment in the reference-current. Small-signal non-synchronous model with immediate update mode. $t_{d_proc} = (0.126)T = 4.12 \ \mu s. D = 0.599, R_{load} = 5.2 \ \Omega, V_{in} = 100 \ V, I_{in} = 120 \ A.$ (a)&(b) $K_i = 5$, (c)&(d) $K_i = 10$, (e)&(f) $K_i = 20$.



Figure 5.9. Transient-response of the closed-loop transfer function $G_{iairef}(z)$ to a 1 A step-increment in the reference-current. Small-signal non-synchronous model with immediate update mode. $t_{d_{proc}} = (0.126)T = 4.2 \,\mu s.$ D = 0.599, $R_{load} = 5.2 \,\Omega$, $V_{in} = 100 \,V$, $I_{in} = 120 \,A$. (a)&(b) $K_i = 30$.

From the results listed in Table 5.5, it is evident that the settling-time of the transient-response is reducing for higher values of K_i . However, the overshoot of the response is also increased. In addition, the reduction of the settling time is not very significant when K_i is ranged from 10 to 30. From this analysis it is possible to conclude that the selection of the parameters of the PI compensator is a compromise between a fast settling-time and a low amplitude and frequency in the initial high-frequency oscillations. Therefore the final selection of K_p and K_i was 10(T) and 10 respectively, which is highlighted in Table 5.5 and its associated transient response is seen in Figure 4.19(c).

	(i igure 5.6 and 5.5)				
Figure	Point in Figure 5.7	Ki	Kp	Maximum over-shoot	Settling time
	A1		5	1.2 %	3.9 ms
5.8(d) 5.8(b)	A2	5	30	-	3.5 ms
5.8(U)	A3		70	2.8 %	4.1 ms
	B1		10	1.5 %	2.7 ms
5.8(c) 5.8(d)	B2	10	40	2.3 %	1.8 ms
	B3		70	7.5 %	2.1 ms
5.8(e) 5.8(f)	C1	20	20	6 %	1.9 ms
	C2		50	14.3 %	1.7 ms
	C3		70	21.3 %	1.0 ms
5.9(a) 5.9(b)	D1	30	30	32.2 %	1.0 ms
	D2		45	32.7 %	1.5 ms
	D3		60	27.5 %	1.7 ms

Table 5.5. Performance of the step-response of the closed-loop transfer function $G_{iairef}(z)$. (Figure 5.8 and 5.9)

To further validate the step responses obtained with the small-signal model, the SABER switched model introduced in Chapter 3, Section 3.5, was used to plot the transient response of the system to a 5 % step- increment in the reference current. Figures 5.10(a) to 5.10(d) show a comparison of the sampled phase-a current, $i_a*(t)$, obtained with the SABER switched model against that obtained with the small-signal non-synchronous model. Figure 5.10(a) shows the results obtained using the K_p and K_i combination corresponding to point A2 in Figure 5.10(b) corresponds to point B2, Figure 5.10(c) to point C2 and Figure 5.10(d) to point D3. The converter parameters used for these simulations are listed in Table 5.4 and the point of operation corresponds to the maximum output power conditions shown in Table 5.3. The schematic diagrams of the SABER switched model are given in Figures C.1 to C.3, Appendix C.

The results obtained showed no significant difference in the 5% settling-times measured, the maximum error of the small-signal model with respect to the switched model was approximately 3.8%, Figure 5.10(d). Also, the maximum over-shoot predicted by the switched model, stayed well within the ranges predicted by the small-signal model.

However, it is evident from Figures 5.10(c) and 5.10(d) that the small-signal model predicts a more lightly damped high frequency oscillation with higher values of $K_i \setminus K_p$ which are not predicted in the switched model results. This might be due to several factors that are included in the switched model but are not modelled in the small-signal one, such as non-linear switching characteristics of the devices in the converter, non-zero switching times of the transistors, and limited resolution of the ADC and DPWM modules which increase the overall damping effect of the system.



5.4. Performance evaluation of the designed PI compensator

5.4.1. Experimental evaluation to large step-increments

Figures 5.11(a), 5.11(b), 5.12(a) and 5.12(b) show the response of the converter phase current, $i_a(t)$, and the converter output voltage, $v_o(t)$, to a step-increment in the reference-current going from 30 A to 60 A. Each of these figures illustrate an overview of the transient response accompanied by a magnification of the event where several switching cycles are shown. The results shown were obtained using the SABER switched model and are compared against those obtained with the experimental prototype introduced in Chapter 3.

The reference-current values employed correspond to the extremes of operation of the design, Table 5.3, as they demand a minimum input current of 60 A and a maximum input current of 120 A respectively. The load resistance used was $R_{load} = 5.2 \Omega$. The step-increment in the simulation was undertaken at 20 ms, once the converter reached steady-state operation. In a similar way, the step-reference in the experimental tests was performed after the converter reached steady-state conditions. The converter parameters employed for the simulations are those listed in Table 5.4. Evidence of the experimental results shown is given in Appendix E, Figures E.9 and E.10.

The results of Figure 5.11(a) correspond to the case with the minimum input voltage, $V_{in} = 50 V$ whilst those of Figure 5.12(a) depict the results obtained with the maximum input voltage, $V_{in} = 100 V$. The values of the compensator gains used to obtain these results were $K_P = 10(T)$ and $K_i = 10$, as proposed in the previous section.

The results confirm the stability of the system over the operating region from 60 A to 120 A, Table 5.4. Moreover, the initial-transient oscillations are well controlled. The settling-time of the transient response is approximately 1.8 ms for the minimum input voltage and 2.1 ms for the maximum input voltage. Also a very good agreement between the simulation and the experimental results can be observed, further corroborating the validity of the SABER switched model. The noise shown in the experimental v_0 trace that is consistent in all the results, is attributed to the use of a differential voltage probe which is susceptible to electro-magnetic interference.









To further corroborate the stability predicted by the small-signal non-synchronous model, the proportional gain was increased to 70(T) where the converter is still expected to be stable. The results in Figure 5.11(b) and 5.12(b) with the new parameter of proportional gain are presented for $V_{in} = 50 V$ and $V_{in} = 100 V$ respectively and confirm the stability of the system. The settling time of the system at these points of operation was approximately 2.15 ms.

5.4.2. Evaluation to small step-increments

Figures 5.13(a), 5.13(b), 5.14(a) and 5.14(b) show the transient response of the converter phase-current to a 5% step-increment in the reference-current, which was obtained using the SABER switched model and the small-signal non-synchronous model. The switched model was used instead of the experimental prototype as the magnitude of the ripple in the prototype obscures the oscillations of the response, making it difficult to assess the validity of the small-signal model. The waveforms plotted in these figures are the phase-a current of the switched model, $i_a(t)$, and the sampled phase-a current of the switched model, $i_a^*(t)$. Once more, the parameters of the converter employed for the simulations are those listed in Table 5.4 and the load resistance used was 5.2 Ω .

The points of operation evaluated correspond to the four extremes of operation of the design example shown in Table 5.3. Figure 5.13(a) and 5.13(b) show the response of the system for the cases with the minimum input current, $l_{in} = 60 A$, and with minimum $V_{in} = 50 V$ and maximum $V_{in} = 100 V$ respectively. The results show a very good agreement between the switched model and the small-signal model. As expected some limit cycle oscillations are present along the switched model response. The settling-time measured is approximately 2.2 ms with the exception of the response of the switched model in Figure 5.13(a), which is about 2.4 ms which is caused by the LCOs. Finally, Figures 5.14(a) and 5.14(b) illustrate the responses for the case with maximum input current, $l_{in} = 120 A$, and minimum and maximum input voltages respectively. These results also show an excellent agreement between the switched and the small-signal model results where the settling-

time measured was approximately 2.1 ms in all the cases. The over-shoot measured in all the responses showed no significant difference.



Figure 5.13. Transient response of the SABER switched model and the small-signal non-synchronous model to a 5% step-increment in the reference-current. $t_{d_proc} = (1.2)T = 4.2 \ \mu s$, $K_p = 10(T)$, $K_i = 10$, $R_{load} = 5.2 \ \Omega$, $I_{ref1} = 30 \ A$, $I_{re2} = 31.5 \ A$. (a) $V_{in} = 50 \ V$; (b) $V_{in} = 100V$.



Figure 5.14. Transient response of the SABER switched model and the small-signal non-synchronous model to a 5% step-increment in the reference-current. $t_{d_proc} = (1.2)T = 4.2 \,\mu s$, $K_p = 10(T)$, $K_i = 10$, $R_{load} = 5.2 \,\Omega$, $I_{ref1} = 60 \,A$, $I_{re2} = 63 \,A$. (a) $V_{in} = 50 \,V$; (b) $V_{in} = 100V$.

5.5. Voltage control loop implementation and results

To verify the effectiveness of the digital average-current control loops in a voltage regulated system, an outer digital voltage control-loop was added to the SABER switched model. The voltage control-loop was included in the DSC unit and its block diagram can be found in Figure C.9, Appendix C.

The outer voltage control-loop compares the converter output voltage against a voltage-reference, the error between the two signals is passed through a PI compensator which calculates the current-reference for the inner average-current control loops. The parameters of the PI compensator employed for the inner current control loops used are those determined in the previous section: $K_p = 10(T)$ and $K_i = 10$. To select the parameters of the compensator for the voltage control-loop, a block diagram of the system was implemented in Simulink using the non-synchronous model developed in Chapter 4, Equation (4.14), which greatly simplifies the design. Using the Simulink control design tools, a digital PI backward-Euler compensator was automatically designed to regulate the voltage control-loop as to achieve the best transient performance and robustness of the system. For this example, the compensator was designed for the point of operation $V_{in} = 100 V$, $V_o = 175 V$, shown in Table 5.3.



Figure 5.15. Bode plots and root-locus of the open, digital voltage control-loop using a PI compensator. Operating poing: $V_{in} = 100 V$; $V_o = 175 V$; $R_{load} = 5.2 \Omega$; $t_{d_{proc}} = (1.2)T = 4.2 \mu s$.

Figure 5.15, shows the open-loop gain, phase and root-locus plots of the voltage control-loop that were used for the compensator design. It is possible to observe that the compensator has a real zero located at 0.902 in the z-plane, equivalent to a frequency of 493 Hz or $3.1 \times 10^3 \text{ rad/sec}$. The expected transient performance of the output voltage to a unit step-change in the voltage-reference is illustrated in Figure 5.16, which main features are an over-shoot of 11.2 % and a settling-time of 4.6 ms.



Figure 5.16. Transient response of the converter output voltage to a unit step-change in the voltage reference. Operating point: $V_{in} = 100 V$; $V_o = 175 V$; $R_{load} = 5.2 \Omega$; $t_{d \ proc} = (1.2)T = 4.2 \mu s$.

Figure 5.17 and 5.18 show the simulation results of the system with voltage regulation. The plots illustrated in these figures are as follows: in the first graph the converter output voltage, v_o , and the voltage reference, v_{ref} , in the second graph the sampled average value of the phase currents, i_a^* and i_b^* ; and in the last graph the instantaneous phase currents, i_a and i_b .

Figure 5.17(a) shows the response of the system to a 10 % step increase in the voltage reference at 20 ms, which goes from 175 V to 192.5 V. The conditions of the system employed were using $R_{load} = 5.2 \Omega$ and $V_{in} = 100 V$. The overshoot and settle-time measured in the voltage waveform are 13 % and 3.1 ms approximately which correspond closely to those predicted by the model in Figure 5.21. The slight difference is attributed to the fact that a step increase is larger in the SABER simulation. Similarly, Figure 5.17(b) shows the
transient response of the system to a large step change in the voltage reference, going from 175 V to 275 V. The results show that the phase-currents have a complementary slow-frequency oscillation which amplitude decays slowly and that is due to the interaction of the converter phases.



Figure 5.17. Transient response of the SABER switched model with voltage regulation to step-increments in the reference-voltage. $t_{d_proc} = (1.2)T = 4.2 \ \mu s$, $R_{load} = 5.2 \ \Omega$, $V_{in} = 100 \ V$. (a) $V_{ref1} = 175 \ V$, $V_{ref2} = 192.5 \ V$; (b) $V_{ref1} = 175 \ V$, $V_{ref2} = 275 \ V$.



Figure 5.18. Transient response of the SABER switched model with voltage regulation to step-changes in the output load. $t_{d_proc} = (1.2)T = 4.2 \ \mu s$, $V_{in} = 100 \ V$, $V_o = 275V$. (a) $R_{load} = 5.2 \ \Omega$, $R_{load2} = 10.4 \ \Omega$; (b) $R_{load} = 5.2 \ \Omega$, $R_{load2} = 2.6 \ \Omega$.

Finally, Figures 5.18(a) and 5.18(b) show the transient performance of the system to step changes in the load resistor at 20 ms, which go from 5.2 Ω to 10.4 Ω and from 5.2 Ω to 2.6 Ω respectively. The conditions used to obtain these results were $V_{in} = 100 V$ and $V_{ref} = 275 V$ which correspond to the maximum operating conditions for the design of the average-current control loops, see Table 5.3. The results confirm the capability of the

system to regulate the output voltage in several conditions. The system successfully regulates the step-load increase in approximately 2 ms whilst the step-load decrease in approximately 3 ms.

5.6. Summary

The sensitivity of the pole trajectories of the small-signal non-synchronous model to variations in the input voltage, output load, computational delay, proportional gain and integral gain was examined and discussed. The analysis revealed that the system is of order seven and contains a pair of high-frequency complex poles and a pair of low-frequency complex poles. The low-frequency complex poles are not present in the synchronous model of the converter and the study suggests that they are introduced by the PI compensator used to regulate the control loops. Moreover, the minimum K_p stability limit observed in the stability-range plots of Chapter 4, seems to be related to the location of the low-frequency complex poles.

Also, from this study, it was observed that variations in the output load and the input voltage seem to have no significant effect on the low-frequency complex poles, however the trajectories of the high-frequency poles are generally moved away from the centre of the unit circle as the load or the input voltage are increased. Finally, it was confirmed that the reduction of the computational delay has a favourable effect on the stability of the system as the trajectories of all the poles move towards the centre of the unit circle.

A design example to select the parameters of a PI compensator was carried out using the small-signal non-synchronous model developed in this thesis. The design aimed to guarantee the stable operation of the system over a defined range of operation and to ensure that the maximum response speed and the minimum overshoot to step-change transients were achieved. To this end, the stability-range plots of the system were used to select a number of proportional and integral gain combinations. The transient performance of the system was evaluated for each of the PI parameters selected, allowing a combination to be identified that meets the design requirements. The final design, $K_p = 10(T)$ and $K_i = 10$, was experimentally verified using 30 A to 60 A increments performed in the converter prototype. The results confirmed the stability of the system over the intended range of operation and showed a maximum speed response of approximately *2.3 ms* with no overshoot.

Finally, simulations results of the average-current controlled interleaved converter with voltage regulation were presented. To speed up the design process, the compensator of the outer voltage control-loop was designed using the non-interleaved model of the converter in combination with SIMULINK. The results demonstrated the capability of the system to regulate the converter output voltage in the presence of load disturbances and step-changes in the voltage reference. The maximum length of time that the system took to regulate the voltage was approximately *3 ms*.

Chapter 6 Conclusions and further research work

6.1. Introduction

This Chapter presents a summary of the achievements of the research work presented in this Thesis. The contributions to the field are highlighted and finally further research work opportunities are identified.

6.2. Summary of the thesis

6.2.1. Small-signal averaged modelling of the DIBC with IPT

A small-signal averaged model of the DIBC with IPT which accounts for the small ac perturbations of each control input independently was derived by replacing the switch networks of the converter with the averaged PWM switch model. By obtaining the transfer functions from the state-space representation of this model, transfer functions that describe the interaction between the control inputs and the converter phase-currents were identified.

Furthermore, the model was implemented in MATLAB and was thoroughly validated employing time-domain and frequency domain simulation results which were obtained from a switched and a large-signal averaged model developed in the SABER software. The results showed an excellent agreement at points of operation below D = 0.7 and for duty-ratio disturbances of up to 5 % where the error was found to be less than 4 %. However, for very high duty-ratios, D = 0.8 or more, which is a less common operating region, noticeable discrepancies occurred which were attributed to the assumptions of the

linearization analysis becoming inaccurate. Still, the error for duty-ratio disturbances up to 5 % remained less than approximately 10 % for the range 0.7 < D < 0.85.

Finally, the frequency response of the transfer functions of the model showed virtually no difference compared to the simulation results, thereby further conforming their validity.

6.2.2. Analysis of the UPWM control techniques

The uniformly sampled PWM control of the DIBC with IPT was analysed for the leading-edge, trailing-edge and triangular carrier modulations by considering the steadystate waveforms of the converter. It was concluded that, similar to the single-phase boost converter, the leading-edge modulation and the trailing-edge modulation allowed the valley phase current and the peak phase current to be controlled respectively. Also, it was demonstrated analytically that by the use of the triangular carrier modulation, it is possible to control the average value of the converter phase currents.

6.2.3. Implementation of the control strategies and experimental validation

The valley-current, peak-current and average-current control strategies were implemented on a Texas Instruments DSC TMS320F28335 and verified experimentally employing a *25 kW*, *30 kHz* prototype of the dual-interleaved converter with IPT.

Also, by examining the operation of the DPWM inbuilt in the DSC, an enhancement to the average-current control methodology was proposed which reduces the delay caused by the cycle-by-cycle update of the DPWM thereby allowing the bandwidth of the control loop to be extended. However, in order to avoid the generation of spurious waveforms using this strategy, the minimum duty-ratio has to be limited according to the computation time of the control algorithm, making the technique unsuitable for converters with high switching frequencies or with a high number of phases.

The steady-state and the dynamic operation of all the control techniques was evaluated by examining the transient response of the phase-currents to large stepreference changes and to step-load changes. The results revealed steady-state error in the phase-currents caused by the mismatched sampling instants which was attributed to the

Chapter 6

propagation delay of the gate-drivers and the turn-on and turn-off delay of the converter transistors. Moreover, it was also revealed that the valley-current control and the peak-current control with UPWM are not suitable for this type of converter as the non-linear behaviour of the phase ripple-current causes an uncontrollable region around D = 0.5. It was observed that the size of the uncontrollable region was strongly dependent on the load resistor, being less significant for resistive loads below $R_{load} = 5 \Omega$. In contrast, the average-current control methodologies showed no evidence of uncontrollable regions.

The system performance under large step-reference changes and to step-load changes further verified the correct regulation of the target control-current for all the control methodologies and the capability of the control loops to balance the phase-currents. It was found that even during transient events the phase currents were correctly balanced. The maximum current imbalance measured was found to be approximately 2 *A* which was observed as a dc-offset of 1 *A* in the differential current, the maximum peak phase-current being 70 *A*.

6.2.4. Modelling of phase interactions in dual-interleaved converters

Using the sampler decomposition methodology, the small-signal, closed-loop transfer functions of the dual interleaved boost converter with IPT were determined for two cases: when the interleaved interaction of the converter phases is disregarded, synchronous model, and when the interleaved interaction is taken into account, non-synchronous model.

A comparison between the closed-loop transfer functions from the two models and the simulation results from a switched and a large-signal averaged model implemented in the SABER simulator was undertaken using time domain step responses and frequency domain responses. Identical PI controllers were used to regulate the phase currents. It was found that the transient response of the models to 5 % disturbances, showed an excellent agreement with the SABER simulations and that the mathematical models showed no significant difference. In the frequency domain, both models showed a good agreement with the simulation results presenting a deviation of no more than approximately *1 dBA* in the magnitude response over most of the operating range, but increasing at low power.

Chapter 6

Also, no more than *10 degrees* in the phase response for frequencies up to approximately *8 kHz*. However, for frequencies above *8 kHz* the transfer-functions of the non-synchronous model showed differences in the phase response, which are likely to be caused by the time-advance and time-delay units employed to model the interleaved interaction of the control-loops.

The stability range of the synchronous and non-synchronous models was determined by examining the location of the poles of the closed-loop transfer functions for several combinations of proportional and integral gains. The comparison confirmed that the interleaved operation of the converter phases has a detrimental effect on the stability of the system and that the synchronous model fails to predict this behaviour. The stability range predicted by the non-synchronous model was further validated by simulation and experimental results.

6.2.5. Stability analysis of the DIBC with IPT for digital average-current control via rootlocus

The objective of this analysis was to study the impact that changes on the input voltage, load resistance, computational delay and the gains of a PI compensator have on the stability of the closed-loop transfer functions of the non-synchronous model.

It was found that with PI compensators two pairs of complex poles were present in the z-plane. A low frequency pair near to the +1 point and a higher frequency pair. The low frequency complex poles were not present in the synchronous model of the converter and their location seemed to be related to the instability caused by the interleaved operation of the converter phases.

Also the results showed that increasing either the input voltage or the load, moves the system poles away from the centre of the unit circle reducing system stability. Conversely, it was found that reducing the processing time of the control algorithms has a favourable effect on stability as the closed-loop poles move towards the centre of the unit circle.

6.2.6. Design of a PI compensator for the current control loops

Aided by the stability-range plots generated using the closed-loop transfer functions of the non-synchronous model, the parameters of a PI compensator were selected. The design was made to ensure maximum response speed and minimum overshoot to stepchange transients, and it was achieved by evaluating the step-response of the transferfunctions for a range of combinations of proportional and integral gains. The final design, $K_p = 10(T)$ and $K_i = 10$, was validated using simulation and experimental results of the system to large-step transients which confirmed the stability for the intended operation region and showed a maximum settling time of 2.2 ms with no over-shoot. The simulation and experimental results showed an excellent agreement.

6.3. Contributions of this research work

6.3.1. Small-signal averaged model of the dual interleaved boost converter with interphase transformer

By using the average PWM switch model, the modelling procedure of the converter accounting for small ac perturbations in the control inputs is fully described and the analytical expressions of the control-to-phase current, control-to-opposite phase current, control-to-output voltage, disturbance current-to-phase current and disturbance currentto-output voltage transfer functions are presented. Compared to similar models reported in the literature, [1-4], the model reported in this thesis has been developed for averagecurrent control of a boost converter, it accounts for the inter-phase transformer, and has been thoroughly validated for a wide range of operation.

6.3.2. Comprehensive study of the uniformly sampled, valley-current, peak-current and average-current control for the DIBC with IPT.

The use of digital uniformly-sampled control techniques has been widely reported for interleaved dc-dc converters in low power applications, such as voltage regulation modules for microprocessors and communication systems, and in particular, digital average-current control is commonly employed in power factor correction systems. However, the use of digital uniformly-sampled control techniques is relatively new to the field of high-power interleaved dc-dc converters and the study carried in this thesis has identified the most suitable form of this type of control for the dual interleaved boost converter.

6.3.3. Immediately updated DPWM operation for average-current control

By updating the DPWM registers immediately after the calculation of the new dutyratio, it is possible to reduce the computational delay of the digital control-loops and therefore extend their bandwidth, increasing the response speed of the converter. Due to its simplicity, this technique is a straightforward alternative to improve the digital controller bandwidth in converters with a small number of phases and/or with a modest switching frequency since the minimum duty-ratio has to be limited to avoid the generation of spurious PWM waveforms. Despite being relatively simple to implement, this technique has not been reported in the literature to the best of the author's knowledge.

6.3.4. Modelling phase interactions in dual interleaved converters.

By means of the sampler decomposition technique, a methodology to account for the interleaved operation and the interaction of the phases in interleaved converters has been developed and validated. This methodology was employed in conjunction with an averaged small-signal model of the converter to derive the small-signal, closed-loop, control-to-output transfer functions of the dual interleaved boost converter with IPT, and it has been demonstrated that the interleaved interaction produces instability issues which cannot be predicted with a model that assumes the synchronized operation of the converter phases. To the best of the author's knowledge, the use of this methodology has not been reported in the field of dc-dc converters, and the effects of interleaved sampling in dc-dc converters have not been previously been analysed or understood.

6.3.5. Stability study of the converter

The stability of the converter has been analysed via root-loci and conclusions have been drawn on how parametric changes in the input voltage, load resistance, computational delay and proportional and integral gains of the compensator affect the stability of the converter. An analysis of this nature has not been found in the literature and is an important contribution to the field as it provides a better understanding of the dynamics of interleaved converters and provides a sound basis for control loop design.

6.4. Further research opportunities

6.4.1. Simplification of the small-signal averaged state-space model of the converter

The state-space representation of the small-signal model of the converter developed in Chapter 2 contains a state-space matrix, A_{av} , defined by three state variables: \tilde{i}_a , \tilde{i}_b and \tilde{v}_o . However, the rank of A_{av} is two, which implies that at least two of the state variables are linearly dependent, therefore, it might be possible to eliminate one of these variables and further reduce the order of the system. Further research work could be carried out to verify this hypothesis which is of special interest as it might lead to a reduction of the order of the closed-loop transfer functions.

6.4.2. Generalization of the sampler decomposition technique for *n*-interleaved converters

The sampler decomposition method, introduced in Chapter 4, may be extended to model *n*-phase interleaved converters in order to obtain a generalized analytical closed-loop transfer function. This is of interest as it will provide further understanding of how the stability of the system is affected as the number of phases of the converter is increased. Also, this modelling methodology might be applied to other dc-dc converter topologies, such as the buck and the buck-boost converter.

6.4.3. Application of the sampler decomposition technique to model the dynamics of peak-current mode control

The sampler decomposition technique might be also employed in conjunction with the peak current control model proposed by Ridley to examine the instabilities caused by the interleaved operation of the converter phases. In contrast with data-sampled modelling methods, which have been already carried out for a DIBC, this methodology might lead to a simpler solution where analytical expressions can be determined. This is of special interest as peak-current control is widely employed in interleaved dc-dc converters due to its fast dynamic performance and inherent over-current protection.

6.4.4. Development of non-linear control methodologies

A further review of the literature identified a few publications on control systems with dual symmetrical control-loops such as those of the dual interleaved boost converter with digital average-current control considered in this thesis. These systems are termed multivariable symmetrical systems, [5-7], and formal non-linear control methodologies have been proposed to analyse their stability. However, the research work has been carried out using analogue controllers only and they do not account for the interleaved operation of the control loops. An opportunity for interdisciplinary research therefore arises, where non-linear control can be used in conjunction with the new closed-loop non-synchronous model of the converter proposed in this thesis to further analyse the system.

6.4.5. Digital implementation of bidirectional control for the dual interleaved converter

Operating the top transistors of the h-bridge that integrates the dual interleaved converter prototype allows the operation of the system in step-down mode and allows the current to flow in the opposite direction as when the bottom transistors are operated only. Developing a digital control strategy that allows the converter to be operated in both stepup and step-down modes is an interesting possibility with a wide range of applications.

6.4.6. Modelling and control of the dual interleaved converter for bidirectional operation

A similar modelling procedure as the one shown in this thesis might be employed to obtain a closed-loop model of the converter when it operates in step-down mode. It would then be possible to have a complete bidirectional model of the converter which can be employed to design a digital controller for bidirectional operation.

6.4.7. Symmetrical (double) sampling on immediate update control algorithm

When the triangular carrier modulation is used, it may be possible to acquire two samples of the average value of the phase-current per switching cycle by triggering a sampling sequence at both the minimum and the maximum values of the DPWM counters. This is of interest as it will allow to improve the control-loop bandwidth in a simple manner.

6.4.8. Implementation of digital control for converter with wide band gap devices

Further work could consider applying the digital control techniques developed in a converter that employs Silicon Carbide devices or other type of wide band gap devices. This is of interest as this devices are capable of higher switching speeds than common Silicon devices, which in consequence will allow to increase the switching frequency of the converter.

6.4.9. Implementation of other current-regulation methodologies

Other current-regulation methods such as predictive control techniques or deadbeat control may be implemented and designed using the model of the converter developed in this Thesis. This opens an opportunity of research as the use of predictive techniques has not been fully investigated for interleaved converters.

6.4.10. Implementation of alternative methods for balancing the IPT current

As mentioned in the literature review, the use of digital control opens the possibility to implement other type of current-regulation methodologies to ensure an adequate balancing of the phase-current in the IPT without the need of measuring the phase-currents. Some of these methods, such as the digital reconstruction of the phase currents employing a single dc-link current transducer, are of interest when the number of phases of the converter is large.

6.5. Conclusion

In this Thesis, the examination of digital uniformly-sampled current control techniques for a dual-interleaved boost converter has been undertaken using the converter steady-state waveforms and experimental results from a 25 kW, 30 kHz prototype. The results revealed that average-current control is the most suitable control technique for this type of converter. On this basis, a small-signal averaged model of the converter has been developed in conjunction with the sampler decomposition technique to account for the interleaved interaction and operation of the converter phases. The resulting closed-loop model demonstrated that the interaction of the converter phases has a detrimental effect

on the stability of the converter which cannot be predicted when the interleaved interaction of the phases is neglected. Further work is necessary to determine analytical expressions of the closed-loop, control-to-phase current transfer-functions for the converter, and to extend the proposed modelling methodology for *n*-interleaved dc-dc converters.

6.6. References

- [1] X. Ming, J. Zhou, K. Yao, and F. C. Lee, "Small Signal Modeling of a High Bandwidth Voltage Regulator Using Coupled Inductors," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 399-406, 2007.
- [2] L. Yean-Kuo, S. Yi-Ping, H. Yu-Ping, L. Yu-Huei, C. Ke-Horng, and H. Wei-Chou, "Time-Multiplexing Current Balance Interleaved Current-Mode Boost DC-DC Converter for Alleviating the Effects of Right-half-plane Zero," *Power Electronics, IEEE Transactions on*, vol. 27, pp. 4098-4112, 2012.
- [3] K. Kono, Y. Nakamura, and M. Yamamoto, "Optimal design of digital control system for interleaved boost chopper circuit with the coupled inductor," in *Power Electronics and Drive Systems (PEDS), 2011 IEEE Ninth International Conference on*, 2011, pp. 995-998.
- [4] D. De, C. Klumpner, C. Patel, K. Ponggorn, M. Rashed, and G. Asher, "Modelling and control of a multi-stage interleaved DC-DC converter with coupled inductors for super-capacitor energy storage system," *Power Electronics, IET*, vol. 6, pp. 1360-1375, 2013.
- [5] A. G. Lindgren and R. F. Pinkos, "Stability of symmetric nonlinear multivariable systems," *Journal of the Franklin Institute,* vol. 282, pp. 92-101, 8// 1966.
- [6] D. Newman, "The analysis of cross-coupling effects on the stability of twodimensional, orthogonal, feedback control systems," *Automatic Control, IRE Transactions on*, vol. 5, pp. 314-320, 1960.
- [7] J. Rootenberg, "On cross coupling and stability in nonlinear control systems," *Automatic Control, IEEE Transactions on,* vol. 16, pp. 73-75, 1971.

Appendix A Derivation of mathematical expressions

A.1. Derivation of equations for steady-state ripples

The following sections show the mathematical procedure used to obtain the equations found in Table 2.1, Chapter 2.

A.1.1. Input inductor current ripple for D < 0.5

The input inductor current ripple, ΔI_{Lin} , may be obtained for D < 0.5 from Figure 2.2(a) by integrating the positive volt-seconds applied to the input inductor:

$$\Delta I_{in} = \frac{1}{L_{in}} \int_{0}^{DT} v_{Lin}(t) dt = \frac{1}{L_{in}} \left(V_{in} - \frac{V_o}{2} \right) DT$$

The equation above can be expressed in terms of V_{in} employing the converter conversion ratio shown in Equation (2.1), $V_o = \frac{V_{in}}{(1-D)}$:

$$\Delta I_{in} = \frac{V_{in}DT}{2L_{in}} \left(\frac{1-2D}{1-D}\right) \tag{A.1}$$

A.1.2. Input inductor current ripple for D < 0.5

The input inductor current ripple may be obtained for D > 0.5 from Figure 2.2(c) by integrating the negative volt-seconds applied to the input inductor:

$$\Delta I_{in} = \frac{1}{L_{in}} \int_{0}^{(1-D)^{T}} V_{Lin}(t) dt = \frac{1}{L_{in}} \left(\frac{V_{o}}{2} - V_{in} \right) (1-D)T$$

The equation above can be expressed in terms of V_{in} employing the converter conversion ratio shown in Equation (2.1), $V_o = \frac{V_{in}}{(1-D)}$:

$$\Delta I_{in} = \frac{V_{in}T}{2L_{in}} (2D - 1)$$
(A.2)

A.1.3. Differential current ripple for D < 0.5

The differential current ripple, ΔI_{diff} , may be obtained for D < 0.5 from Figure 2.2(a) by integrating the positive volt-seconds between the terminals of the IPT, in other words:

$$\Delta I_{diff} = \frac{1}{L_{diff}} \int_{0}^{DT} v_{diff}(t) dt = \frac{1}{L_{diff}} (V_o) DT$$

The equation above can be expressed in terms of V_{in} employing the converter conversion ratio shown in Equation (2.1), $V_o = \frac{V_{in}}{(1-D)}$:

$$\Delta I_{diff} = \frac{V_{in}DT}{L_{diff}} \left(\frac{1}{1-D}\right)$$
(A.3)

A.1.4. Differential current ripple for D > 0.5

The differential current ripple may be obtained for D > 0.5 from Figure 2.2(c) by integrating the positive volt-seconds between the terminals of the IPT, in other words:

$$\Delta I_{diff} = \frac{1}{L_{diff}} \int_{0}^{(1-D)^{T}} V_{diff}(t) dt = \frac{1}{L_{diff}} (V_{o})(1-D)T$$

The equation above can be expressed in terms of V_{in} employing the converter conversion ratio shown in Equation (2.1), $V_o = \frac{V_{in}}{(1-D)}$:

$$\Delta I_{diff} = \frac{V_{in}T}{L_{diff}} \tag{A.4}$$

A.1.5. Large and small phase-current ripples

The large phase-current ripple can be expressed as the sum of two ripple magnitudes: the input inductor current ripple and the differential current ripple. Under the assumption that the input inductor current divides equally between the converter phases and that the IPT windings inductance is identical, the input inductor current ripple will be also equally divided between the phases, hence, the large phase-current ripple may be expressed as:

$$\Delta I_{lrg} = \frac{1}{2} \Delta I_{in} + \Delta I_{diff} \tag{A.5}$$

The small phase-current ripple may also be deduced similarly, and can be expressed as the difference between the input inductor current ripple and the differential current ripple:

$$\Delta I_{smll} = \frac{1}{2} \Delta I_{in} - \Delta I_{diff}$$
 (A.6)

A.1.6. Large and small phase-current ripple for D < 0.5

Substituting Equations (A.1) and (A.3) in Equations (A.5), the large phase-current ripple for D < 0.5 can be obtained:

$$\Delta I_{lrg} = \frac{1}{2} \frac{V_{in} DT}{2L_{in}} \left(\frac{1-2D}{1-D} \right) + \frac{V_{in} DT}{L_{diff}} \left(\frac{1}{1-D} \right) = \frac{V_{in} DT}{(1-D)} \left(\frac{1-2D}{4L_{in}} + \frac{1}{L_{diff}} \right)$$
(A.7)

In a similar way, the small phase current ripple may be obtained by substituting Equations (A.1) and (A.3) in Equation (A.6):

$$\Delta I_{lrg} = \frac{1}{2} \frac{V_{in} DT}{2L_{in}} \left(\frac{1-2D}{1-D}\right) - \frac{V_{in} DT}{L_{diff}} \left(\frac{1}{1-D}\right) = \frac{V_{in} DT}{(1-D)} \left(\frac{1-2D}{4L_{in}} - \frac{1}{L_{diff}}\right)$$
(A.8)

A.1.7. Large and small phase-current ripple for D > 0.5

Substituting Equations (A.2) and (A.4) in Equations (A.5), the large phase-current ripple for D > 0.5 can be obtained:

$$\Delta I_{lrg} = \frac{1}{2} \frac{V_{in}T}{2L_{in}} (2D-1) + \frac{V_{in}T}{L_{diff}} = V_{in}T \left(\frac{2D-1}{4L_{in}} + \frac{1}{L_{diff}}\right)$$
(A.9)

In a similar way, the small phase current ripple may be obtained by substituting Equations (A.2) and (A.4) in Equation (A.6):

$$\Delta I_{lrg} = \frac{1}{2} \frac{V_{in}T}{2L_{in}} (2D-1) - \frac{V_{in}T}{L_{diff}} = V_{in}T \left(\frac{2D-1}{4L_{in}} - \frac{1}{L_{diff}}\right)$$
(A.10)

A.2. Derivation of the closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ with no interleaving

In the following paragraphs, the closed-loop transfer functions $G_{iairef}(z)$ and $G_{ibiref}(z)$ will be derived from Figure 4.4(b), Chapter 4, for the case where the samplers S_a and S_b operate synchronously and the DPWM module is updated immediately after the processing of the sampled current. From inspection of the block diagram of Figure 4.4(b), Chapter 4, the following relations can be written by ignoring the time advance and time delay units of the sampler S_b :

$$\tilde{d}_{a}(z) = C(z)\mathcal{Z}\left[\tilde{i}_{ref}(s)\right] - C(z)\tilde{d}_{a}(z)\mathcal{Z}\left[G_{M}(s)G_{di}(s)H_{a}(s)\right] - C(z)\tilde{d}_{b}(z)\mathcal{Z}\left[G_{M}(s)G_{dvi}(s)H_{a}(s)\right]$$
(A.11a)

$$\tilde{d}_{b}(z) = C(z)\mathcal{Z}\left[\tilde{i}_{ref}(z)\right] - C(z)\tilde{d}_{b}(z)\mathcal{Z}\left[G_{M}(s)G_{di}(s)H_{b}(s)\right] - C(z)\tilde{d}_{a}(z)\mathcal{Z}\left[G_{M}(s)G_{dui}(s)H_{b}(s)\right]$$
(A.11b)

$$\tilde{i}_{a}(z) = \tilde{d}_{a}(z)\mathcal{Z}\left[G_{M}(s)G_{di}(s)\right] + \tilde{d}_{b}(z)\mathcal{Z}\left[G_{M}(s)G_{dxi}(s)\right]$$
(A.11c)

$$\tilde{i}_{b}(z) = \tilde{d}_{b}(z)\mathcal{Z}\left[G_{M}(s)G_{di}(s)\right] + \tilde{d}_{a}(z)\mathcal{Z}\left[G_{M}(s)G_{dxi}(s)\right]$$
(A.11d)

The constituent z-transforms found in Equations (A.11a) to (A.11d) are listed in Table A.1 and are redefined in a compact notation for this particular case-study where t_{d_proc} <

Table A.1. Constituent transfer functions of the digital average-current controlled DIBC with IPT in the z-domain assuming the synchronous operation of the samplers S_a and S_b .

Modified z-transform	Compact notation
$\mathcal{Z}\left\{G_{h0}(s)\Delta PWM(s)G_{di}(s)H_{A}(s)\right\} = \\ = G_{di}H_{A}(z,1-t_{d_proc})$	$G_{di}H_{A}(z)$
$\mathcal{Z}\left\{G_{h0}(s)\Delta PWM(s)G_{di}(s)H_{B}(s)\right\}$ $=G_{di}H_{B}(z,1-t_{d_proc})$	$G_{di}H_{B}(z)$
$\mathcal{Z}\left\{G_{h0}(s)\Delta PWM(s)G_{dxi}(s)H_{A}(s)\right\} = \\ = G_{dxi}H_{A}(z,1-t_{d_proc})$	$G_{dxi}H_{A}(z)$
$\mathcal{Z}\left\{G_{h0}(s)\Delta PWM(s)G_{dxi}(s)H_{B}(s)\right\} = \\ = zG_{dxi}H_{B}(z,1-t_{d_{proc}})$	$G_{dxi}H_{B}(z)$
$\mathcal{Z}\left\{G_{h0}(s)\Delta PWM(s)G_{di}(s)\right\} = G_{di}\left(z,1-t_{d_proc}\right)$	$G_{di}(z)$
$\mathcal{Z}\left\{G_{h0}(s)\Delta PWM(s)G_{dxi}(s)\right\} = G_{dxi}(z,1-t_{d_proc})$	$G_{dxi}(z)$
	$ \begin{array}{l} \mbox{Modified z-transform} \\ \hline \mathcal{Z}\left\{G_{h0}\left(s\right)\Delta PWM(s)G_{di}\left(s\right)H_{A}(s)\right\} = \\ = G_{di}H_{A}\left(z,1-t_{d_proc}\right) \\ \hline \mathcal{Z}\left\{G_{h0}\left(s\right)\Delta PWM(s)G_{di}\left(s\right)H_{B}(s)\right\} \\ = G_{di}H_{B}\left(z,1-t_{d_proc}\right) \\ \hline \mathcal{Z}\left\{G_{h0}\left(s\right)\Delta PWM(s)G_{dxi}\left(s\right)H_{A}(s)\right\} = \\ = G_{dxi}H_{A}\left(z,1-t_{d_proc}\right) \\ \hline \mathcal{Z}\left\{G_{h0}\left(s\right)\Delta PWM\left(s\right)G_{dxi}\left(s\right)H_{B}\left(s\right)\right\} = \\ = zG_{dxi}H_{B}\left(z,1-t_{d_proc}\right) \\ \hline \mathcal{Z}\left\{G_{h0}\left(s\right)\Delta PWM(s)G_{dxi}\left(s\right)H_{B}\left(s\right)\right\} = \\ = G_{di}\left(z,1-t_{d_proc}\right) \\ \hline \mathcal{Z}\left\{G_{h0}\left(s\right)\Delta PWM(s)G_{dxi}\left(s\right)\right\} = \\ = G_{di}\left(z,1-t_{d_proc}\right) \\ \hline \mathcal{Z}\left\{G_{h0}\left(s\right)\Delta PWM(s)G_{dxi}\left(s\right)\right\} = \\ = G_{dxi}\left(z,1-t_{d_proc}\right) \\ \hline \end{array} $

T/2, in other words, $\Delta PWM(s) = e^{-s(td_proc)}$. Upon substitution of the compact notation of the constituent z-transforms into Equations (A.11a) to (A.11d), Equations (A.12a) to (A.12d) are found:

$$\tilde{d}_{a}(z) = C(z)\tilde{i}_{ref}(z) - C(z)\tilde{d}_{a}(z)G_{di}H_{a}(z) - C(z)\tilde{d}_{b}(z)G_{dxi}H_{a}(z)$$
(A.12a)

$$\tilde{d}_{b}(z) = C(z)\tilde{i}_{ref}(z) - C(z)\tilde{d}_{b}(z)G_{di}H_{b}(z) - C(z)\tilde{d}_{a}(z)G_{dxi}H_{b}(z)$$
(A.12b)

$$\tilde{i}_{a}(z) = \tilde{d}_{a}(z)G_{di}(z) + \tilde{d}_{b}(z)G_{dxi\phi}(z)$$
(A.12c)

$$\tilde{i}_b(z) = \tilde{d}_b(z) \mathbf{G}_{di}(z) + \tilde{d}_a(z) \mathbf{G}_{dxi}(z)$$
(A.12d)

Solving Equation (A.12a) for $\tilde{d}_b(z)$ and substituting it into Equation (A.12b) yields:

$$\tilde{d}_{a}(z) = C(z)\tilde{i}_{ref}(z) \left\{ \frac{1 + C(z) \left[G_{di} H_{b}(z) - G_{dxi} H_{a}(z) \right]}{\left[C(z) G_{di} H_{a}(z) + 1 \right] \left[C(z) G_{di} H_{b}(z) + 1 \right] - C^{2}(z) G_{dxi} H_{a}(z) G_{dxi} H_{b}(z)} \right\}$$
(A.13a)

In a similar way, by solving Equation (A.12b) for $\tilde{d}_a(z)$ and substituting it into Equation (A.12a), $\tilde{d}_b(z)$ is obtained:

$$\tilde{d}_{b}(z) = C(z)\tilde{i}_{ref}(z) \left\{ \frac{1 + C(z) \left[G_{di} H_{a}(z) - G_{dxi} H_{b}(z) \right]}{\left[C(z) G_{di} H_{a}(z) + 1 \right] \left[C(z) G_{di} H_{b}(z) + 1 \right] - C^{2}(z) G_{dxi} H_{a}(z) G_{dxi} H_{b}(z)} \right\}$$
(A.13b)

Finally, substituting Equations (A.13a) and (A.13b) into (A.12c), the closed-loop transfer function $G_{iairef}(z)$ can be derived as:

$$\frac{\tilde{i}_{a}(z)}{\tilde{i}_{ref}(z)} = C(z) \left\{ \frac{G_{di}(z) + G_{dxi}(z) + C(z) \{G_{di}(z) [G_{di}H_{b}(z) - G_{dxi}H_{a}(z)] + G_{dxi}(z) [G_{di}H_{a}(z) - zG_{dxi}H_{b}(z)] \}}{[C(z)G_{di}H_{a}(z) + 1][C(z)G_{di}H_{b}(z) + 1] - C^{2}(z)G_{dxi}H_{a}(z)G_{dxi}H_{b}(z)} \right\} \dots (A.14a)$$

In a similar way, substituting Equations (A.13a) and (A.13b) into (A.12d), *G*_{ibiref}(z) can be found:

$$\frac{\tilde{i}_{b}(z)}{\tilde{i}_{ref}(z)} = C(z) \left\{ \frac{G_{di}(z) + G_{dxi}(z) + C(z) \left\{ G_{di}(z) \left[G_{di}H_{a}(z) - z G_{dxi}H_{b}(z) \right] + G_{dxi}(z) \left[G_{di}H_{b}(z) - G_{dxi}H_{a}(z) \right] \right\}}{\left[C(z) G_{di}H_{a}(z) + 1 \right] \left[C(z) G_{di}H_{b}(z) + 1 \right] - C^{2}(z) G_{dxi}H_{a}(z) G_{dxi}H_{b}(z)} \right\} \dots$$

...(A.14b)

To further simplify Equations (A.14a) and (A.14b) it can be assumed that the current transducers have unity gain $H_a(s) = H_b(s) = 1$, resulting in:

$$G_{iairef}(z) = \frac{\tilde{i}_{a}(z)}{\tilde{i}_{ref}(z)} = \frac{C(z) \left[G_{di}(z) + G_{dxi}(z) \right] \left[1 + C(z) G_{di}(z) - C(z) G_{dxi}(z) \right]}{\left[1 + C(z) G_{di}(z) \right]^{2} + C^{2}(z) G_{dxi}^{2}(z)}$$
(A.15a)

$$G_{ibiref}(z) = \frac{\tilde{i}_{b}(z)}{\tilde{i}_{ref}(z)} = \frac{C(z) \left[G_{di}(z) + G_{dxi}(z) \right] \left[1 + C(z) G_{di}(z) - C(z) G_{dxi}(z) \right]}{\left[1 + C(z) G_{di}(z) \right]^{2} + C^{2}(z) G_{dxi}^{2}(z)}$$
(A.15b)

Appendix B MATLAB Scripts

This appendix comprises the MATLAB scripts that were employed in this work to plot the results of the analytical closed-loop models of the DIBC with IPT presented in Chapter 4.

B.1. Calculation of the closed-loop transfer functions

The script shown at the end of this section was employed to calculate the numerical solution of the analytical closed-loop transfer functions of the DIBC with IPT shown in Chapter 4, Table 4.2.

MATLAB Script

```
%TITLE: MODEL OF THE DIGITAL AVERAGE-CURRENT CONTROL LOOP FOR THE
            DUAL INTERLEAVED BOOST CONVERTER WITH INTERPHASE TRASNFORMER
8
%FILE NAME: DIBC DACC NoDelay.m
%AUTHOR: Alejandro Villarruel-Parra, alexvip_22@hotmail.com
%VERSION: 6.0 - DEC 2014, FINAL THESIS VERSION
clc;
clear;
Ts = 1/30000;
Tspha = [Ts];
Tsphb = [Ts (Ts/2)];
Kp = 10 * Ts;
Ki = 20;
Delay proc = (0.21*Ts);
L c = 75.14e-6;
L = 5.12e-6;
M = L c^{*}(0.997);
LT = 2*L*(L c + M) + ((L c^2) - (M^2));
C = 45e-6;
%V o = 220;
```

```
V i = 100;
R^{-1} = 5.2;
R_{in} = 0.029;
%P = 25000;
\%Iin = P/V i
Iin = 120;
D = 1 - sqrt((V i/(R l*Iin)))
&D = 0.63;
%V i = (1-D)*V o
%Iin = (V_i) / (R_l*((1-D)^2))
%V_o = V_i / (1-D);
Step Duty = D*0.05;
Step Iref = 1;
Aav = [((-R in*(L c+M))/LT) ((-R in*(L c+M))/LT) (-((1-K))/LT) (-((1-K))/LT) (-((1-K))/LT))]
D)*(L c+M))/(LT));((-R in*(L c+M))/LT) ((-R in*(L c+M))/LT) (-((1-
D) * (L^{c+M}) / (LT); ((1-D) / C) ((1-D) / C) ((-1) / (R 1 + C));
V i)*(L-M))/((1-D)*LT)) ((V i*(L+L c))/((1-D)*LT)) 0;((-
V i)/(2*R l*C*((1-D)^2))) ((-V i)/(2*R l*C*((1-D)^2))) (1/C)];
Cav = [1 \ \overline{0} \ 0; 0 \ 1 \ 0; 0 \ 0 \ 1];
Dav = [0 \ 0 \ 0; 0 \ 0 \ 0; 0 \ 0];
X0 = [0;0;0];
X02 = [(V i/(2*R l*((1-D)^2)));(V i/(2*R l*((1-D)^2)));(V i/(1-D))];
SSav = ss(Aav, Bav, Cav, 0); %State Space of the Average Small-signal model.
TFav = tf(SSav); %Individual transfer functions of all the state
space.
%TFav Distribution
8 {
               +----+
               | |1.da |2.db |3.iz |
               |----|----|
               |1.Ia |Gdaia|Gdbia|Gizia|
               |----|----|
               |2.Ib |Gdaib|Gdbib|Gizib|
               |----|----|
               |3.vo |Gdavo|Gdbvo|Gizvo|
               +----+
8}
%Tranfer Functions
Gdaia = TFav(1,1);
Gdaib = TFav(2,1);
Gdbia = TFav(1,2);
Gdbib = TFav(2,2);
Gdavo = TFav(3,1);
Gdbvo = TFav(3,2);
Gizia = TFav(1,3);
Gizib = TFav(2,3);
Gizvo = TFav(3,3);
[numGdaia,denGdaia] = tfdata(Gdaia,'v');
[numGdaib,denGdaib] = tfdata(Gdaib,'v');
[numGdbia,denGdbia] = tfdata(Gdbia,'v');
[numGdbib,denGdbib] = tfdata(Gdbib,'v');
```

Appendix B

```
Gdi =
minreal(absorbDelay(c2d(tf(numGdaia,denGdaia,'iodelay',(Delay proc)),Ts,'
zoh')));
Gdxi =
minreal(absorbDelay(c2d(tf(numGdaib,denGdaib,'iodelay',(Delay proc)),Ts,'
zoh')));
if (Delay proc < (0.5*Ts)) || (Delay proc > (0.51*Ts))
    Gdxi phi =
zpk(minreal(absorbDelay(c2d(tf(numGdaib,denGdaib,'iodelay',((Ts/2)+Delay
proc)),Ts,'zoh'))));
    Gdxi teta =
zpk(minreal(tf([10],1,Ts)*absorbDelay(c2d(tf(numGdaib,denGdaib,'iodelay',
(Ts-(Ts/2)+Delay proc)),Ts,'zoh'))));
else
    Gdxi phi =
zpk(minreal(tf(1,[10],Ts)*(c2d(tf(numGdaib,denGdaib),Ts,'zoh'))));
    Gdxi teta = zpk(minreal(c2d(tf(numGdaib,denGdaib),Ts,'zoh')));
end
Comp = pid(Kp,Ki,0,0,Ts,'IFormula','BackwardEuler');
%Comp = zpk([],[1],Kp,Ts);
Num IaIref = zpk(minreal(Comp*(Gdi + Gdxi)));
Den IaIref = zpk(minreal(1 + Comp*(Gdi + Gdxi)));
IaIref = zpk(minreal((Num IaIref/Den IaIref)));
[numIaIref denIaIref] = tfdata(IaIref, 'v');
CE_IaIref = zpk(minreal(Comp*(Gdi + Gdxi)));
A = minreal(Comp*((Gdi^2)-(Gdxi phi*Gdxi teta)));
B = minreal(2*Comp*Gdi);
Num IaIref3 = minreal(zpk(Comp*(Gdi + Gdxi phi + A)));
Den IaIref3 = zpk(minreal(1 + B + minreal(Comp*A)));
IaIref3 = zpk(minreal((Num IaIref3/Den IaIref3)));
[numIaIref3 denIaIref3] = tfdata(IaIref3, 'v');
CE IaIref3 = zpk(minreal(B + minreal(Comp*A)));
Num IbIref3 = zpk(minreal(Comp*(Gdi + Gdxi teta + A)));
IbIref3 = zpk(minreal((Num IbIref3/Den IaIref3)));
[numIbIref3 denIbIref3] = tfdata(IbIref3, 'v');
figure(1);
step(IaIref3);
hold on;
```

B.2. Stability range script

This script was employed to obtain the stability region plots shown in Chapter 4, employing the transfer functions listed in Table 4.2, Chapter 2. It operates by evaluating in an iterative manner the stability of the transfer-functions for different values of proportional and integral gains. The resulting data is stored in a *.txt* file which was plotted

employing ORIGINLAB. This program is comprised by two scripts. The outer-loop script defines the parameters and stores the data of each iteration whilst the Inner-loop script calculates the closed-loop transfer functions of the converter for each iteration.

```
_MATLAB Script
```

```
%TITLE:
            STABILITY RANGE PLOTS FOR THE AVERAGE-CURRENT CONTROL
            OF THE DIBC WITH IPT (OUTER LOOP)
8
%FILE NAME: Stability Analysis2
%AUTHOR: Alejandro Villarruel-Parra, alexvip 22@hotmail.com
%VERSION: 2.0 - JAN 2015, FINAL THESIS VERSION
clc;
clear;
Ts = 1/30000;
Delay proc = (0.5*Ts);
L c = 75.14e-6;
L = 5.12e-6;
M = L c^{*}(0.997);
LT = 2*L*(L c + M) + ((L c^2) - (M^2));
C = 45e-6;
V \circ = 220;
V \bar{i} = 100;
R^{-1} = 5.2;
R in = 0.029;
\$P = 25000;
\%Iin = P/V i
Iin = 120;
D = 1 - sqrt((V i/(R l*Iin)))
%D = 0.8;
%V i = (1-D)*V o
%Iin = (V i)/(R l*((1-D)^2))
V \circ = V i/(1-D)
file name = '5R2 Iin120 Vin100 Sync 0Td5.txt';
fid = fopen(file_name,'w');
fprintf(fid, 'R load = %f\t', R l);
fprintf(fid, 'D = %f\t',D);
fprintf(fid, '\n');
fprintf(fid,['Ki\t','Kp_min\t','Kp_max\t']);
fprintf(fid, '\n');
file name = '5R2 Iin120 Vin100 NonSync 0Td5.txt';
fid2 = fopen(file_name,'w');
fprintf(fid2,'R load = %f\t',R l);
fprintf(fid2, 'D = %f\t', D);
fprintf(fid2, '\n');
fprintf(fid2,['Ki\t','Kp min\t','Kp max\t']);
fprintf(fid2,'\n');
handle wait = waitbar(0,['Currently Evaluating Ki=',num2str(j)]);
```

```
%Res Ki = [logspace(-2,log10(0.09),10),logspace(-
1,log10(0.9),10),logspace(0,log10(9),20),logspace(1,log10(60),30)];
%Res Kp = [logspace(log10(0.005),log10(0.09),15),logspace(-
1,log10(0.9),10),logspace(0,log10(9),20),logspace(1,log10(100),60)];
%Res Ki = [logspace(-2,log10(0.09),10),logspace(-
1,loq10(0.9),10),logspace(0,loq10(9),20),logspace(1,loq10(90),20),logspac
e(2,log10(200),20)];
%Res Kp =
[logspace(log10(0.0005),log10(0.009),7),logspace(log10(0.001),log10(0.09)
,10),logspace(-
1,log10(0.9),10),logspace(0,log10(9),20),logspace(1,log10(200),80)];
Res Ki = logspace(0, log10(100), 50);
Res Kp =
[logspace(log10(0.1),log10(0.9),15),logspace(log10(1),log10(9),15),10:2.5
:200];
y = 1;
for j = Res Ki
    waitbar((y/length(Res Ki)), handle wait, ['Currently Evaluating
Ki=',num2str(j)]);
    Ki = j;
    x = 1;
    flag = 0;
    Stability_IaIref = nan(length(Res_Ki),2);
    Stability_IaIref3 = nan(length(Res Ki),2);
    Stability IbIref3 = nan(length(Res Ki),2);
    for k = \text{Res } Kp
        Kp = k*Ts;
        run Stability Analysis;
        Stability_IaIref(x,1) = Ki;
        Stability_IaIref3(x,1) = Ki;
Stability_IbIref3(x,1) = Ki;
        if isstable(IaIref)
            Stability IaIref(x, 2) = Kp/Ts;
        else
            Stability IaIref(x,2) = NaN;
        end
        if isstable(IaIref3)
            Stability IaIref3(x,2) = Kp/Ts;
        else
            Stability_IaIref3(x,2) = NaN;
        end
        if isstable(IbIref3)
            Stability_IbIref3(x,2) = Kp/Ts;
        else
            Stability IbIref3(x,2) = NaN;
        end
        x = x+1;
    end
    [Val Index] = min(Stability_IaIref);
    StabilityRange(1,1:2) = Stability IaIref(Index(1,2),:);
    [Val Index] = min(Stability_IaIref3);
    StabilityRange(1,3:4) = Stability_IaIref3(Index(1,2),:);
    [Val Index] = max(Stability_IaIref);
    StabilityRange(2,1:2) = Stability_IaIref(Index(1,2),:);
    [Val Index] = max(Stability IaIref3);
```

```
StabilityRange(2,3:4) = Stability_IaIref3(Index(1,2),:);
fprintf(fid,'%f\t',Ki);
fprintf(fid,'%f\t',StabilityRange(1,2));
fprintf(fid,'%f\t',StabilityRange(2,2));
fprintf(fid2,'%f\t',StabilityRange(2,2));
fprintf(fid2,'%f\t',StabilityRange(1,4));
fprintf(fid2,'%f\t',StabilityRange(2,4));
fprintf(fid2,'%f\t',StabilityRange(2,4));
fprintf(fid2,'\n');
y = y +1;
end
close(handle_wait);
fclose('all');
```

MATLAB Script___

%TITLE:	STABILITY RANGE PLOTS FOR THE AVERAGE-CURRENT CONTROL
90	OF THE DIBC WITH IPT (INNER LOOP)
%FILE NAME:	Stability_Analysis
%AUTHOR:	Alejandro Villarruel-Parra, alexvip 22@hotmail.com
%VERSION:	2.0 - JAN 2015, FINAL THESIS VERSION

```
Aav = [((-R in^{(L c+M)})/LT) ((-R in^{(L c+M)})/LT) (-((1-R))/LT)]
D)*(L_c+M))/(LT));((-R_in*(L_c+M))/LT) ((-R_in*(L_c+M))/LT) (-((1-
D) * (L C+M)) / (LT)); ((1-D) /C) ((1-D) /C) ((-1) / (R 1*C))];
Bav = [((V_i*(L+L_c))/((1-D)*LT)) (((-V_i)*(L-M))/((1-D)*LT)) 0;(((-D)*LT))  0;(((-D)*LT))  0;(((-D)*LT)) 0;(((-D)*LT)) 0;(((-D)*LT))) 0;(((-D)*LT)) 0;(((-D)*LT)) 0;(((-D)*LT))) 0;(((-D)*LT)) 0;(((-D)*LT))) 0;(((-D)*LT)) 0;(((-D)*LT))) 0;(((-D)*LT)) 0;(((-D)*LT))) 0;(((-D)*LT)) 0;(((-D)*LT))) 0;(((-D)*LT))) 0;(((-D)*LT)) 0;(((-D)*LT))) 0;(((-D)*LT))) 0;(((-D)*LT))) 0;(((-D)*LT))) 0;((((-D)*LT)))) 0;((((-D)*LT)))) 0;(
V i)*(L-M))/((1-D)*LT)) ((V i*(L+L c))/((1-D)*LT)) 0;((-
V_i)/(2*R_l*C*((1-D)^2))) ((-V_i)/(2*R_l*C*((1-D)^2))) (1/C)];
Cav = [1 \ \overline{0} \ 0; 0 \ 1 \ 0; 0 \ 0 \ 1];
Dav = [0 \ 0 \ 0; 0 \ 0 \ 0; 0 \ 0];
X0 = [0;0;0];
X02 = [(V i/(2*R l*((1-D)^2)));(V i/(2*R l*((1-D)^2)));(V i/(1-D))];
SSav = ss(Aav,Bav,Cav,Dav); %State Space of the Average Small-signal
model.
TFav = tf(SSav);
                                                           %Individual transfer functions of all the state
space.
%Tranfer Functions
Gdaia = TFav(1,1);
Gdaib = TFav(2,1);
Gdbia = TFav(1,2);
Gdbib = TFav(2,2);
Gdavo = TFav(3, 1);
Gdbvo = TFav(3,2);
 [numGdaia denGdaia] = tfdata(Gdaia,'v');
 [numGdaib denGdaib] = tfdata(Gdaib, 'v');
 [numGdbia denGdbia] = tfdata(Gdbia, 'v');
 [numGdbib denGdbib] = tfdata(Gdbib, 'v');
```

Appendix B

```
Gdi =
minreal(absorbDelay(c2d(tf(numGdaia,denGdaia,'iodelay',(Delay proc)),Ts,'
zoh')));
Gdxi =
minreal(absorbDelay(c2d(tf(numGdaib,denGdaib,'iodelay',(Delay proc)),Ts,'
zoh')));
if (Delay proc < (0.5*Ts)) || (Delay proc > (0.51*Ts))
    Gdxi phi =
zpk(minreal(absorbDelay(c2d(tf(numGdaib,denGdaib,'iodelay',((Ts/2)+Delay
proc)),Ts,'zoh'))));
    Gdxi teta = zpk(minreal(tf([1
0],1,Ts) *absorbDelay(c2d(tf(numGdaib,denGdaib,'iodelay',(Ts-
(Ts/2)+Delay proc)),Ts,'zoh'))));
else
    Gdxi phi = zpk(minreal(tf(1,[1
0],Ts)*(c2d(tf(numGdaib,denGdaib),Ts,'zoh'))));
    Gdxi teta = zpk(minreal(c2d(tf(numGdaib,denGdaib),Ts,'zoh')));
end
Comp = pid(Kp,Ki,0,0,Ts,'IFormula','BackwardEuler');
Num IaIref = zpk(minreal(Comp*(Gdi + Gdxi)));
Den IaIref = zpk(minreal(1 + Comp*(Gdi + Gdxi)));
IaIref = zpk(minreal((Num IaIref/Den IaIref)));
[numIaIref denIaIref] = tfdata(IaIref, 'v');
CE IaIref = zpk(minreal(Comp*(Gdi + Gdxi)));
A = minreal(Comp*((Gdi^2)-(Gdxi_phi*Gdxi_teta)));
B = minreal(2*Comp*Gdi);
Num IaIref3 = minreal(zpk(Comp*(Gdi + Gdxi phi + A)));
Den IaIref3 = zpk(minreal(1 + B + minreal(Comp*A)));
IaIref3 = zpk(minreal((Num IaIref3/Den IaIref3)));
[numIaIref3 denIaIref3] = tfdata(IaIref3, 'v');
CE IaIref3 = zpk(minreal(B + minreal(Comp*A)));
Num IbIref3 = zpk(minreal(Comp*(Gdi + Gdxi teta + A)));
IbIref3 = zpk(minreal((Num IbIref3/Den IaIref3)));
[numIbIref3 denIbIref3] = tfdata(IbIref3, 'v');
```

B.3. Root-loci plots

This script was employed to produce the root locus plots presented in Chapter 4, Section 4.5. This program is comprised by two scripts. The outer-loop script defines the evaluation parameters and plots the poles of the closed-loop transfer functions for each iteration. The Inner loop script calculates the closed-loop transfer functions of the converter for the parameters defined in the outer loop.

MATLAB Script

```
%TITLE:
            ROOT LOCI OF THE DIGITAL AVERAGE-CURRENT CONTROLLED
8
            DIBC WITH IPT (OUTER LOOP)
%FILE NAME: PZ_MAPS_2.m
%AUTHOR:
            Alejandro Villarruel-Parra, alexvip 22@hotmail.com
%VERSION:
            2.0 - FEB 2015, FINAL THESIS VERSION
clc;
clear;
Ts = 1/30000;
Tspha = [Ts];
Tsphb = [Ts (Ts/2)];
Kp = 40 * Ts;
\bar{Ki} = 10;
colorcito = color(6);
Comp = pid(Kp,Ki,0,0,Ts,'IFormula','BackwardEuler');
%Comp = Kp*Ts;
%Comp = zpk([],[1],Kp,Ts);
Delay proc = (0.5*Ts);
L c = 75.14e-6;
L = 5.12e-6;
M = L c^{*}(0.997);
LT = 2*L*(L_c + M) + ((L_c^2) - (M^2));
C = 45e-6;
V_i = 50;
R_1 = 5.2;
R in = 0.029;
\$P = 4500;
%Iin = P/V_i
\$ Iin = (30.7*2);
%D = 1 - sqrt((V_i/(R_l*Iin)))
D = 0.63;
%Iin = (V_i)/(R_l*((1-D)^2))
%V_o = V_i/(1-D)
%P = V_i*Iin
clpoles = [];
clpoles3 = [];
param = [];
x = 1;
for D = [0.2:0.05:0.8]
    PZ MAPS;
    %Num CE3 = (pole(IaIref))';
    Num CE3 = (pole(IaIref3))';
    real_CE3 = real(Num_CE3);
    [CE3sortedreal,index] = sort(real_CE3);
    Num CE3 = Num CE3(index);
    imag_CE3 = imag(Num_CE3);
    [CE3sortedimag,index] = sort(imag_CE3);
    Num_CE3 = Num_CE3(index);
    param = [param;D];
    clpoles3 = [clpoles3;Num CE3];
```

Appendix B

end

```
%General Plots To Establish the Context
hf1 = figure(1);
units = get(hf1, 'Units');
set(hf1,'Units','inches');
set(hf1, 'Position', [15,0.2,7,6.5]); set(hf1, 'Units', units);
for x = 1:length(clpoles3(1,:))
    p h = plot(real(clpoles3(:,x)),imag(clpoles3(:,x)),'x-');
    set(p h, 'Color', colorcito);
    hold on;
end
set(gca,'xtick',[-3,-2,-1,0,1]);
set(gca,'ytick',[-1,0,1]);
xlabel('Real Axis');
ylabel('Imaginary Axis');
title(sprintf('Root Locus - Poles of G_i_a_i_r_e_f(s) and
G_ib_ir_ef(s) \ R_lo_ad = \$5.1f; \overline{V}in = \$5.1f', R l, V i);
zgrid;
grid;
hf2 = figure(2);
for x = 1:length(clpoles3(1,:))
    p h = plot(real(clpoles3(:,x)),imag(clpoles3(:,x)),'x-');
    set(p h, 'Color', colorcito);
    hold on;
end
axis([0.75 1 -0.25 0.25]);
xlabel('Real Axis');
ylabel('Imaginary Axis');
title(sprintf('Root Locus - Poles of G_i_a_i_r_e_f(s) and
G_ib_ir_ef(s) \ R_lo_ad = \$5.1f; V_in = \$5.1f', R_l, V_i);
zgrid;
grid;
```

MATLAB Script

```
%TITLE:
            ROOT LOCI OF THE DIGITAL AVERAGE-CURRENT CONTROLLED
            DIBC WITH IPT (INNER LOOP)
2
%FILE NAME: PZ MAPS.m
           Alejandro Villarruel-Parra, alexvip 22@hotmail.com
%AUTHOR:
%VERSION:
            2.0 - FEB 2015, FINAL THESIS VERSION
Aav = [((-R in*(L c+M))/LT) ((-R in*(L c+M))/LT) (-((1-
D)*(L c+M))/(LT));((-R in*(L c+M))/LT) ((-R in*(L c+M))/LT) (-((1-
D)*(L c+M))/(LT));((1-D)/C) ((1-D)/C) ((-1)/(R l*C))];
Bav = [((V_i*(L+L_c))/((1-D)*LT)) ((((-V_i)*(L-M))/((1-D)*LT)) 0;(((-D)*LT)))]
V i) * (L-M) ) / ((1-D) *LT) ) ((V_i * (L+L_c)) / ((1-D) *LT) ) 0; ((-
V i)/(2*R l*C*((1-D)^2))) ((-V i)/(2*R l*C*((1-D)^2))) (1/C)];
Cav = [1 \ 0 \ 0; 0 \ 1 \ 0; 0 \ 0 \ 1];
Dav = [0 \ 0 \ 0; 0 \ 0 \ 0; 0 \ 0];
X0 = [0;0;0];
X02 = [(V i/(2*R l*((1-D)^{2}))); (V i/(2*R l*((1-D)^{2}))); (V i/(1-D))];
SSav = ss(Aav,Bav,Cav,0); %State Space of the Average Small-signal model.
```

```
%Individual transfer functions of all the state
TFav = tf(SSav);
space.
%Tranfer Functions
Gdaia = TFav(1,1);
Gdaib = TFav(2,1);
Gdbia = TFav(1,2);
Gdbib = TFav(2, 2);
Gdavo = TFav(3, 1);
Gdbvo = TFav(3,2);
Gizia = TFav(1,3);
Gizib = TFav(2,3);
Gizvo = TFav(3,3);
[numGdaia, denGdaia] = tfdata(Gdaia, 'v');
[numGdaib, denGdaib] = tfdata(Gdaib, 'v');
[numGdbia, denGdbia] = tfdata(Gdbia, 'v');
[numGdbib,denGdbib] = tfdata(Gdbib,'v');
Gdi =
minreal(absorbDelay(c2d(tf(numGdaia,denGdaia,'iodelay',(Delay proc)),Ts,'
zoh')));
Gdxi =
minreal(absorbDelay(c2d(tf(numGdaib,denGdaib,'iodelay',(Delay proc)),Ts,'
zoh')));
if (Delay proc < (0.5*Ts)) || (Delay proc > (0.51*Ts))
    Gdxi phi =
zpk(minreal(absorbDelay(c2d(tf(numGdaib,denGdaib,'iodelay',((Ts/2)+Delay
proc)),Ts,'zoh'))));
    Gdxi teta = zpk(minreal(tf([1
0],1,Ts) *absorbDelay(c2d(tf(numGdaib,denGdaib,'iodelay',(Ts-
(Ts/2)+Delay proc)),Ts,'zoh'))));
else
    Gdxi phi = zpk(minreal(tf(1,[1
0],Ts)*(c2d(tf(numGdaib,denGdaib),Ts,'zoh'))));
    Gdxi_teta = zpk(minreal(c2d(tf(numGdaib,denGdaib),Ts,'zoh')));
end
Num IaIref = zpk(minreal(Comp*(Gdi + Gdxi)));
Den IaIref = zpk(minreal(1 + Comp*(Gdi + Gdxi)));
IaIref = zpk(minreal((Num_IaIref/Den_IaIref)));
[numIaIref denIaIref] = tfdata(IaIref, 'v');
CE IaIref = zpk(minreal(Comp*(Gdi + Gdxi)));
A = minreal(Comp*((Gdi^2)-(Gdxi_phi*Gdxi_teta)));
B = minreal(2*Comp*Gdi);
Num IaIref3 = minreal(zpk(Comp*(Gdi + Gdxi phi + A)));
Den IaIref3 = zpk(minreal(1 + B + minreal(Comp*A)));
IaIref3 = zpk(minreal((Num IaIref3/Den IaIref3)));
[numIaIref3 denIaIref3] = tfdata(IaIref3,'v');
CE_IaIref3 = zpk(minreal(B + minreal(Comp*A)));
Num IbIref3 = zpk(minreal(Comp*(Gdi + Gdxi_teta + A)));
IbIref3 = zpk(minreal((Num_IbIref3/Den_IaIref3)));
[numIbIref3 denIbIref3] = tfdata(IbIref3, 'v');
```

B.4. Export files for SABER Cosmo Scope

For the purpose of result comparisons, this script allows to save time-domain and frequency-domain data generated in MATLAB into *.txt* files which are compatible with SABER Cosmo Scope.

MATLAB Script

```
%TITLE:
           Export to SABER
%FILE NAME: SABER Export.m
%AUTHOR:
            Alejandro Villarruel-Parra, alexvip 22@hotmail.com
ia = extract portion(I a,0,I a(length(I a)-1,1),(Ts),X02(1,1));
ib = extract portion(I b, 0, I b(length(I b)-1, 1), (Ts/2), X02(2, 1));
ia3 = extract portion (I a3, 0, I a3 (length (I a3) -1, 1), (Ts), X02 (1, 1));
ib3 = extract portion (I b3, 0, I b3 (length (I b3), 1), (Ts/2), X02(2, 1));
%vo = extract portion(V o, 0, V o(length(V o), 1), 0, X02(3, 1));
%iref = extract portion(I ref,0,I ref(length(I ref)-1,1),0,Iin/2);
%SimulinkExport = concatenate simulink arrays(ia,ib);
export dwf saber('SimExport Iaz.txt', {'t`s', 'Ia`analog`-`real'},ia);
export dwf saber('SimExport Ibz.txt', {'t`s', 'Ib`analog`-`real'},ia);
export dwf saber('SimExport Ia3z.txt', {'t`s', 'Ia3`analog`-`real'},ia3);
export dwf saber('SimExport Ib3z.txt', {'t`s', 'Ib3`analog`-`real'},ib3);
%ia = extract portion(I a,0,I a(length(I a),1),0,X02(1,1));
%ib = extract portion(I b,0,I b(length(I b),1),0,X02(2,1));
%vo = extract_portion(V_o,0,V_o(length(V_o),1),0,X02(3,1));
%Dab = extract portion(D ab,0,D ab(length(D ab),1),0,D);
%SimulinkExport = concatenate simulink arrays(ia,ib,vo,Dab);
%export_dwf_saber('SimExport.txt',{'t`s','Ia`analog`A`real','Ib`analog`A`
```

_____MATLAB Script_

real', 'Vo`analog`V`real', 'D`analog`V`real'}, SimulinkExport);

```
function export dwf saber(file name, header, data)
%Author: Alejandro Villarruel Parra
8
            alexvip 22@hotmail.com
%Version:
           1.0
           August 2014
%Date:
%This function is used to create a file that contains the data of anarray
0 sof the form f(t, y(t)). The format of the file created is suitable
%for SABER Cosmoscope.
    *file name is a string of the type filename.extension.
9
90
90
    *header is a cell of string indicating the header for the saber file:
8
        h{1} = 'tname''tunit'
8
        h{2} = 'y1name''y1type''y1unit''y1datatype'
00
       h{3} = 'y2name''y2type''y2unit''y2datatype'
8
        . . .
8
       h{n} = 'ynname''yntype''ynunit''yndatatype'
```

```
Appendix B
```

```
00
00
    *data is an array of the form f(t, y1(t), y2(t), \ldots yn(t))
00
8
    Example:
9
        export dwf saber('input current.txt',head,Ia);
00
        where:
        head = {'t''s','Ia''analog''A''t'};
00
        Ia = [t, t^{2}];
00
[hx,hy] = size(header);
[dx, dy] = size(data);
fid = fopen(file name, 'w');
for x = 1 : hy
    fprintf(fid,'%s\t',header{x});
end
fprintf(fid, '\n');
for x = 1 : dx
    for y = 1 : dy
        fprintf(fid,'%f\t',data(x,y));
    end
    fprintf(fid, '\n');
end
fclose(fid);
beep on;
beep;
beep;
beep;
msgbox('File ready','Export Data Waveform to SABER txt file','help');
end
```

Appendix C SABER simulation models

This appendix includes the schematic diagrams of the SABER models developed to simulate the digital current control strategies for the dual-interleaved boost converter. Three simulation models were developed and are described briefly in the following sections.

C.1. Switched model of the valley-current and peak-current control

The digital valley-current and digital peak-current control strategies were modelled employing the switched model of the DIBC with IPT introduced in Chapter 2, Section 2.4.2 and a special module referred to as DSC unit. This module reproduces the functional behaviour of the control algorithms implemented in the digital signal controller that was used to control the prototype of the DIBC introduced in Chapter 3.

Figure C.1 shows the schematic diagram of the switched model of the DIBC with IPT and the DSC unit employed to model the valley-current and the peak-current control strategies. Figure C.2 shows the complete schematic diagram of the DSC unit. The module operates using z-domain sampling-signals to speed-up the simulation time in SABER. Some of the elements found in this diagram were specifically designed for this application. These elements are shown with a uniform grey fill and the code developed can be found in the last section of this appendix.

As seen in Figure C.2, the DSC unit is organized in three main stages: the Master Clock Generator, the DPWM Counters and the Control Loops A and B. Figure C.3(a) shows a magnification of the DPWM A counter whilst Figure C.3(b) shows a magnification of the phase-a control loop.

C.2. Switched model of the average-current control with normal and immediate update

Figure C.4 shows the schematic diagram of the SABER switched model of the DIBC with IPT and the DSC unit employed to model the average-current control strategies with normal and immediate update of the DPWM. Figure C.5 shows the complete schematic diagram of the DSC unit developed for this control strategies. Figure C.6(a) shows a magnification of the DPWM A counter and Figure C.6(b) shows a magnification of the phase-a control loop.

C.3. Averaged model of the average-current control with normal and immediate update

In addition to the switched model the average-current control strategies were also simulated employing the average model of the DIBC with IPT introduced in Chapter 2, Section 2.4.3. Figure C.7 shows the schematic diagram of the SABER average model of the DIBC with IPT and the DSC unit employed to model the average-current control strategies with normal and immediate update of the DPWM units. In contrast to the DSC unit employed for the switched models introduced in the previous sections, the DSC unit employed in this simulation model does not contain DPWM modules and does not account for the quantization effects of the ADC conversion, thereby reducing considerably the simulation time. Figure C.8 shows the complete schematic diagram of the DSC unit employed.

C.4. SABER Scripts

C.4.1. Z-domain selector

```
#TITLE: Selector, z-domain SABER
#AUTHOR: Alejandro Villarruel-Parra
#VERSION: 1.0 - DEC 2012
#SUMMARY: This template models a controllable two-input one-output
selector.
template Selector zin1 zin2 select zout = td
state nu zin1, zin2, zout, select
number td = 0
{
    when (event_on(zin1) | event_on(zin2)) {
```
}

```
if(select < 1) {
    schedule_event(time, zout, zin1)
}
else {
    schedule_event(time, zout, zin2)
}
</pre>
```

C.4.2. Z-domain saturation

```
#TITLE:
            Saturation, z-domain SABER
            Alejandro Villarruel-Parra
#AUTHOR:
            1.0 - DEC 2012
#VERSION:
            This template models a saturation in the z-domain. The
#SUMMARY:
maximum and minimum limits are user selectable.
template SaturationZ zin1 zout1 = lvlmax, lvlmin
state nu zin1, zout1
number lvlmax, lvlmin
{
      when (event on(zin1)) {
            if(zin1 >= lvlmax) {
                  schedule event(time, zout1, lvlmax)
            }
            else if (zin1 <= lvlmin) {</pre>
                  schedule event(time, zout1, lvlmin)
            }
            else {
                  schedule event(time, zout1, zin1)
            }
      }
}
```

C.4.3. Z-domain event generator

```
#TITLE:
            Event generator, z-domain SABER
#AUTHOR:
           Alejandro Villarruel-Parra
            1.0 - DEC 2012
#VERSION:
#SUMMARY:
            This template generates a pulse when the difference between
the input signals are smaller than epsilon.
template EventGenerator zin1 zin2 dout = epsilon
state nu zin1, zin2
state logic 4 dout
number epsilon
state logic_4 out_state
      when (event on(zin2)) {
            if( abs(INT(zin1)-INT(zin2)) < epsilon) {</pre>
                  out_state = 14 1
            }
            else {
                  out_state = 14_0
            }
            schedule event(time, dout, out state)
      }
```



Dual Interleaved Boost Converter Switched Model

Figure C.1. SABER schematic diagram of the switched model of the DIBC with IPT and the DSC unit employed to simulate the valley-current and the peak-current control strategies.

Appendix C

SABER simulation models



255





Dual Interleaved Boost Converter Switched Model

Figure C.4. SABER schematic diagram of the switched model of the DIBC with IPT and the DSC unit employed to simulate the average-current control strategies with normal and immediate update.

Appendix C

SABER simulation models







Figure C.7. SABER schematic diagram of the average model of the DIBC with IPT and the DSC unit employed to simulate the average-current control strategies with normal and immediate update.







Appendix D Control implementation diagrams

In this appendix, the Simulink block diagrams employed to program the TMS320F28335 digital signal controller are shown and briefly explained.

D.1. Uniformly-sampled valley-current, peak-current and average-current control implementations

Figures D.1 and D.2 illustrate the hierarchical Simulink block diagrams that comprise the implementation of the control algorithms in the DSC. Figure D.1 groups the tasks that the DSC executes to initialize the system (System Start block, System Initialize block and Target preferences block), to enable/disable the DPWM modules operation (bottom left corner), and to handle the ADC interrupts (bottom right corner). Also, in this diagram it is possible to select the mode of operation of the gate drivers that interface the DSC with the interleaved converter prototype (upper right corner).



Figure D.1. Simulink diagram of the top-level tasks of the control algorithms.



Figure D.2. Simulink diagram of the ADC interrupt service routine used for the normal DPWM update control algorithms.

Every time that the DSC is notified of an interrupt request from the ADC, the block diagram illustrated in Figure D.2(a) is executed. This section of the diagram performs the ADC interrupt service routine which reads the data acquired by the ADC (leftmost block), determines which control-loop has to be compensated (group of blocks at the rightmost side), and calculates new duty-ratio of the corresponding phase. The compensation of the control loops is executed by the blocks denoted Control loop A and Control loop B, which are respectively illustrated in Figure D.3. These sections of the algorithm are employed to scale the data read from the ADC, calculate the new duty-ratio of the corresponding DPWM modules. The block employed to scale the ADC data is shown in detail in Figure D.4.

Finally, the StepRef block which is connected to one of the output terminals of the ADC block, is employed to generate a step-increment or decrement in the reference-current of the control algorithms and is shown in detail in Figure D.5. The output of this block is connected to the control loop blocks.

Appendix D



Figure D.3. Simulink diagrams of the current control-loops used for the normal DPWM update control algorithms.

ADC Scaling



Figure D.4. Simulink diagram of the ADC scaling process.



Figure D.5. Simulink diagram of the step-reference generator.

D.2. Uniformly-sampled average-current control with immediate DPWM update

Figures D.1 and D.6 illustrate the hierarchical Simulink block diagrams that comprise the implementation of the digital average-current control with immediate DPWM update in the DSC. Figure D.1 contains the tasks that the DSC executes to initialize the system, to enable/disable the DPWM modules operation, and to handle the ADC interrupts. This section of the block diagram is identical to the one employed for the control strategies with normal DPWM update shown in Appendix D.1.

Figure D.6 illustrates the section of the diagram that performs the ADC interrupt service routine. This diagram performs the ADC interrupt service routine which reads the data acquired by the ADC (leftmost block), determines which control-loop has to be compensated (group of blocks at the rightmost side), and calculates the new duty-ratio of the corresponding phase. The compensation of the control loops is executed by the blocks denoted Control loop A and Control loop B, which are respectively illustrated in Figure D.7. Compared to the control loop algorithms illustrated in Figure D.3, these control loops



ADC INTERRUPT SERVICE ROUTINE, ADC_ISR

Figure D.6. Simulink diagram of the ADC interrupt service routine used for the immediate DPWM update control algorithm.

incorporate a building block called Anti-glitch, Figure D.8, which is used to avoid the issue described in Section 3.4.4, Chapter 3. The minimum duty-ratio constraint required to avoid the generation of spurious PWM waveforms, is implemented using the saturation blocks shown in the diagrams.



Figure D.7. Simulink diagrams of the current control-loops used for the immediate DPWM update control algorithm.



Figure D.8. Simulink diagram of the anti-glitch algorithm.

Appendix E Simulation and experimental results

This appendix shows the simulation and experimental evidence of the stability-range plots presented in Figures 4.10 and 4.11 from Chapter 4, and the experimental results used to evaluate the converter performance PI compensator designed in Chapter 5 Section 5.4.1 (Figures 5.11 and 5.12).

Figures E.1 and E.2 present the simulation results obtained with the SABER averaged model when the system is operated with normal DPWM update mode, which supports the results shown in Figure 4.10(a). Similarly, Figures E.3 and E.4 present the simulation results obtained with the SABER averaged model when the system is operated with immediate DPWM update mode, which support the plot shown in Figure 4.10(b). The first plot of each figure shows the value of the proportional gain, K_p , and the integral gain, K_i , whilst the second plot shows the reference current, i_{ref} , and the sampled phase-a current, $i_a^*(t)$.

The simulation was undertaken in three stages. First, the simulation was run until the system achieved stable steady-state operation. In the second stage, the proportional and integral gain values where changed for the values to be evaluated. Finally, *500 ns* after the second stage, a *5 %* step-increment was applied to the reference-current input. The system conditions can be found in the figure titles.

Figures E.5 and E.6 present the experimental results obtained when the system operates with normal DPWM update mode, which support the results of Figure 4.11(a). Similarly, Figures E.7 and E.8 show the experimental results obtained when the system operates with immediate DPWM update mode, which support the results of Figure 4.11(b).

In Figures E.5 to E.8 the first plot shows the transistor driving signal v_{1a} . However due to the size of the time scale and the amount of switching cycles, it is not possible to appreciate the waveforms on detail. The second plot, shows a PWM signal which was modulated using the proportional gain of the PI compensator. The purpose of this signal in these experiments was identify the value of the proportional gain that was being used during the experiments and to synchronize the scope to capture the evolution of the phase-a current when the proportional gain value was changed. Similar to the v_{1a} plot, due to the size of the time scale it is not possible to appreciate the details of this waveform. The third plot shows the phase-a current, i_a . Finally, the last plot illustrates the converter output voltage, v_o .

The procedure followed to obtain these results is similar to the one employed with the SABER simulator. First the converter was allowed to reach steady-state conditions using a combination of proportional and integral gains that allowed a stable operation of the system. Once the system reached steady-state operation, the proportional and integral gains were modified and the response of the phase-a current was captured. In some cases, despite that the oscillations of i_a were almost unnoticeable, the oscillation produced a constant audible noise.

Finally, Figure E.9 an E.10 show the experimental results obtained with the PI controller designed in Chapter 5, Section 5.4.1.

Simulation and experimental results



Figure E.1. Simulation results employed to validate the stability range predicted by the small-signal nonsynchronous model illustrated in Figure 4.10, Chapter 4. Initial test conditions $K_p = 30 K_i = 20$. $V_{in} = 85$, $R_{load} = 5.2 \Omega$, D = 0.5480.

Simulation and experimental results



Figure E.2. Simulation results employed to validate the stability range predicted by the small-signal nonsynchronous model illustrated in Figure 4.10, Chapter 4. Initial test conditions $K_p = 30 K_i = 20$. $V_{in} = 85$, $R_{load} = 5.2 \Omega$, D = 0.5480.

Simulation and experimental results



Figure E.3. Simulation results employed to validate the stability range predicted by the small-signal synchronous model illustrated in Figure 4.10, Chapter 4. Initial test conditions $K_p = 30 K_i = 20$. $V_{in} = 85$, $R_{load} = 5.2 \Omega$, D = 0.5480.

Simulation and experimental results



Figure E.4. Simulation results employed to validate the stability range predicted by the small-signal synchronous model illustrated in Figure 4.10, Chapter 4. Initial test conditions $K_p = 30 K_i = 20$. $V_{in} = 85$, $R_{load} = 5.2 \Omega$, D = 0.5480.

Simulation and experimental results



Figure E.5. Experimental results employed to validate the stability range predicted by the small-signal nonsynchronous model with normal DPWM update mode found in Figure 4.11(a), Chapter 4. Initial test conditions $K_p = 10 K_i = 1$. $V_{in} = 85$, $R_{load} = 5.2 \Omega$, D = 0.5480.

Simulation and experimental results



Figure E.6. Experimental results employed to validate the stability range predicted by the small-signal nonsynchronous model with normal DPWM update mode found in Figure 4.11(a), Chapter 4. Initial test conditions $K_p = 10 K_i = 1$. $V_{in} = 85$, $R_{load} = 5.2 \Omega$, D = 0.5480.

Simulation and experimental results



Figure E.7. Experimental results employed to validate the stability range predicted by the small-signal nonsynchronous model with immediate DPWM update mode found in Figure 4.11(b), Chapter 4. Initial test conditions $K_p = 10 K_i = 1. t_{d_proc} = (0.5)T$, $V_{in} = 85$, $R_{load} = 5.2 \Omega$, D = 0.5480.



Figure E.8. Experimental results employed to validate the stability range predicted by the small-signal nonsynchronous model with immediate DPWM update mode found in Figure 4.11(b), Chapter 4. Initial test conditions $K_p = 10 K_i = 1. t_{d_proc} = (0.5)T$, $V_{in} = 85$, $R_{load} = 5.2 \Omega$, D = 0.5480.



Figure E.9. Experimental results employed to validate PI design given in Chapter 5, Section 5.4. $K_p = 10T K_i = 10. t_{d_proc} = (0.12)T$, $R_{load} = 5.2 \Omega$, $I_{ref1} = 30 A$, $I_{ref2} = 60 A$. (a) $V_{in} = 50 V$. (b) $V_{in} = 100 V$.



Figure E.10. Experimental results employed to validate PI design given in Chapter 5, Section 5.4. $K_p = 70T K_i$ = 10. $t_{d_proc} = (0.12)T$, $R_{load} = 5.2 \Omega$, $I_{ref1} = 30 A$, $I_{ref2} = 60 A$. (a) $V_{in} = 50 V$. (b) $V_{in} = 100 V$.

Appendix F Control platform schematics





